Sustainability and Cost Assessment of Lithium Battery Health Management Strategies for Electric Unmanned Aerial Vehicles

Master of Science Thesis

P.J. Vossen



Sustainability and Cost Assessment of Lithium Battery Health Management Strategies for Electric Unmanned Aerial Vehicles

Master of Science Thesis

by

P.J. Vossen

to obtain the degree of Master of Science at the Delft University of Technology, to be defended publicly on 21st March 2022.

Student number:4454979Project duration:February 2021 – March 2022Thesis committee:Dr. ir. B.F. (Bruno) SantosDr. M. (Márcia) BaptistaDr. E.J.J. (Ewoud) SmeurIr. P.C. (Paul) Roling

An electronic version of this thesis is available at http://repository.tudelft.nl/.



Acknowledgements

This thesis report presents research that has been carried out with the aim to obtain a master's degree in Aerospace Engineering at Delft University of Technology. When I first started the Literature Study of this research last year, I did foresee the journey that lay ahead of me. Although each thesis has its ups and downs, I look back at a very fruitful year and am proud of the research I have completed.

Throughout my research, I was inspired by my supervisor Márcia Baptista who helped scope my project. Márcia motivated me to think outside the box and supported my enthusiasm to venture off the beaten path and dive into the unknown world of battery health management strategies. Despite most of our meetings being online given the COVID pandemic, I felt a close connection to Márcia and am very grateful for her guidance throughout my thesis.

Additionally, my gratitude goes towards Bruno Santos who provided critical feedback as an experienced thesis supervisor at the faculty of Aerospace Engineering. I also thank Wiljan Vermeer and Marco Stecca from the faculty of Electrical Engineering, Mathematics & Computer Science who helped me along the way by answering my questions related to battery practices. Moreover, I would like to acknowledge the PyBaMM community that I became a part of and provided me support in using the battery modelling simulation package in Python.

Finally, I cannot begin to express my thanks to my family for their unconditional support throughout my studies. A special thank you to my boyfriend Matys who always knew how to bring a smile to my face no matter what time of day it was. Last my not least, I am grateful for my friends and housemates in Delft and Amsterdam who enriched my life as a student in countless ways.

Pippa Vossen Amsterdam, March 2022

Contents

List of Tables List of Abbreviations Introduction I Scientific Paper II Literature Study previously graded under AE4020 1 Introduction IIA 2 Aircraft Maintenance 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.2 Maintenance Types. 2.2.1 Scheule Maintenance 2.2.2 Preventive Maintenance 2.2.2.1 Schulten Based Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Schulten Based Maintenance 2.2.3 Predictive Maintenance	xi xiii xvii 1 29 31 33 35
List of Abbreviations Introduction I Scientific Paper II Literature Study previously graded under AE4020 1 Introduction IIA 2 Aircraft Maintenance 2.1 General Background 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.2 Maintenance Types 2.2.1 Corrective Maintenance 2.2.2.1 Scheduled Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach 2.3.1 Prognostics 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles	xiii xvii 1 29 31 33 35
Introduction I Scientific Paper II Literature Study previously graded under AE4020 1 Introduction IIA IIA 2 Aircraft Maintenance 2.1 General Background 2.1.2 Organisations and Regulations. 2.2.2 Preventive Maintenance 2.2.2 Preventive Maintenance 2.2.2.1 Scheduled Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach 2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles	xvii 1 29 31 33 35
 I Scientific Paper II Literature Study previously graded under AE4020 Introduction IIA Aircraft Maintenance General Background	1 29 31 33 35
 II Literature Study previously graded under AE4020 1 Introduction IIA 2 Aircraft Maintenance 2.1 General Background 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.2 Maintenance Types 2.2.1 Corrective Maintenance 2.2.2 Preventive Maintenance 2.2.2.1 Scheduled Maintenance 2.2.2.2 Condition Based Maintenance 2.2.3 Predictive Maintenance 2.2.3 Integrated Approach. 2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles. 	29 31 33 35
Interactive Study previously graded under AE4020 1 Introduction IIA 2 Aircraft Maintenance 2.1 General Background 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.2.1 Corrective Maintenance 2.2.2 Preventive Maintenance 2.2.2.1 Scheduled Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach. 2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles.	29 31 33 35
 Introduction IIA Aircraft Maintenance General Background Safety in Aviation. Safety in Aviation. Organisations and Regulations. Organisations and Regulations. Organisations and Regulations. Corrective Maintenance . Scheduled Approach . <li< td=""><td>31 33 35</td></li<>	31 33 35
IIA 2 Aircraft Maintenance 2.1 General Background 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.1.2 Organisations and Regulations. 2.1.1 Corrective Maintenance 2.2.2 Preventive Maintenance 2.2.2.1 Scheduled Maintenance 2.2.2.2 Condition Based Maintenance 2.2.2.3 Predictive Maintenance 2.2.2.3 Integrated Approach 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles.	33 35
 2 Aircraft Maintenance 2.1 General Background	35
2.11 General Background	00
 2.1.1 Safety in Aviation. 2.1.2 Organisations and Regulations. 2.2 Maintenance Types. 2.2.1 Corrective Maintenance. 2.2.2 Preventive Maintenance. 2.2.2.1 Scheduled Maintenance 2.2.2.2 Condition Based Maintenance 2.2.2.3 Predictive Maintenance. 2.2.3 Integrated Approach. 2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles. 	35
 2.1.2 Organisations and Regulations. 2.2 Maintenance Types. 2.2.1 Corrective Maintenance. 2.2.2 Preventive Maintenance. 2.2.2.1 Scheduled Maintenance 2.2.2.2 Condition Based Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach. 2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles. 	35
 2.2 Maintenance Types	36
 2.2.1 Corrective Maintenance	36
 2.2.2 Preventive Maintenance	37
 2.2.2.1 Scheduled Maintenance 2.2.2.2 Condition Based Maintenance 2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach 2.3 Research gaps 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles 	
2.2.2.3 Predictive Maintenance 2.2.3 Integrated Approach 2.3 Research gaps 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles	38
 2.2.3 Integrated Approach	39
2.3 Research gaps. 2.3.1 Prognostics 2.3.2 Electric Aerial Vehicles.	40
2.3.1 Prognostics	40
2.3.2 Electric Aerial Vehicles	40
	40
3 Battery Health Management	43
3.1 Relevance	43
3.2 Battery Type	44
3.3 Battery Characteristics	44
3.3.1 Performance and Reliability	44
3.3.1.2 End-of-Discharge	40
3.3.2 State of Health	47
3.3.2.1 End-of-Life	50
3.4 Battery Health Management	50
3.4.1 Battery Management System	51
3.4.2 Scheduled Battery Health Management	51
3.4.3 Predictive Battery Health Management	51
3.4.4 Unalleliges	
3.6 Research Gaps	54
4 Derfermen - Frederitien	
4 Performance Evaluation 4 Accuracy and Precision	57
4.2 False Positives and False Negatives	

	4.3	Performance Metrics
		4.3.1 Average Bias
		4.3.2 Sample Standard Deviation
		4.3.3 Mean Squared Error
		4.3.4 Mean Absolute Percentage Error
		4.3.5 Prognostic Horizon
		4.3.6 $\alpha - \lambda$ Performance
		4.3.7 Relative Accuracy
		438 Cumulative Relative Accuracy
		439 Convergence
	11	Conclusion
	7.7	
5	\cos	t-Benefit Analysis 6
	5.1	Parameters
		5.1.1 Financial Costs
		5111 Maintenance Costs
		5 1 1 2 Evaluation Methods
		5.1.2 Effectiveness
		5.1.2 Effectiveness
		5.1.3 Scienulic Performance
		5.1.4 Social Impact
		5.1.5 Environmental Impact
	5.2	Experimental Setup
		5.2.1 Data
		5.2.2 Analysis
		5.2.2.1 Scenario Analysis 6
		5.2.2.2 Discrete Event Simulation
		5.2.2.2 District Event Simulation
		5.2.3 Cost-Benefit Model
		5.2.4 Assumptions
	5.3	Conclusion
	IID	7
	пр	
6	Pro	blem Statement 7
	6.1	Research Gaps
	62	Research Outline 7
	63	Research Questions
	0.5	
7	Met	hodology and Project Timeline 7
	7.1	Methodology
		7.1.1 Model Preparation
		712 Model Development 7
		7.1.2 Results Analysis Validation and Conclusion
	7.0	
	(.2	
		7.2.1 Phase I
		7.2.2 Phase 2 and 3
		7.2.3 Phase 4, 5 and 6
III	Appe	ndices 8
1	с.	
1	Stat	Asucai Anaiysis 8
	1.1	Mixed range
		1.1.1 Mixed range - Monte Carlo
		1.1.2 Mixed range - Distribution of Sustainability and Cost Assessment results 8
		1.1.3 Mixed range - Statistical T-tests
	1.2	Short range
		121 Short range - Monte Carlo
		122 Short range _ Distribution of Sustainability and Cost Assessment results
		1.2.2 Onor range - Distribution of Sustamability and Cost Assessment results 0

1	1.2.3 Short range - Statistical T-tests 88
1.3 N	Medium range
1	1.3.1 Medium range - Monte Carlo 90
1	1.3.2 Medium range - Distribution of Sustainability and Cost Assessment results 91
1	1.3.3 Medium range - Statistical T-tests 92
1.4 I	Long range
1	1.4.1Long range - Monte Carlo92
1	L.4.2 Long range - Distribution of Sustainability and Cost Assessment results
1	1.4.3 Long range - Statistical T-tests 93
1.5 I	Distances - Statistical T-tests
1.6 V	/alidation results
2 Resea	arch Limitations and Recommendations 103
2.1 N	Mission Profiles
2.2 E	Battery Model
2.3 E	Battery Health Management Strategies
2.4 (Cost-Benefit Analysis
Bibliogra	105 apply

List of Figures

2.1 2.2	Breakdown Tree of Maintenance Types [53]	37 39
3.1	Hexagonal spider graphics of performance characteristics for Lithium-ion battery types LCO, LMO, NMC, LFP, NCA and NMC, showing specific energy, specific power, safety, performance,	
3.2	life span and cost [23]Typical Open Circuit Voltage hysteretic behaviour of a Lithium Iron Phosphate battery subject	45
33	to a charge and discharge process at 20Ah [11]	46 47
3.4	Discharge cycles at different battery life stages for constant load profiles from one single Lithium battery [112]	48
3.5	Lithium-ion battery capacity retention behaviour related to different Depth-of-Discharge ranges during battery Dynamic Stress Testing (DST) [132]	50
3.6	Lithium-ion battery capacity loss due to different Depth-of-Discharge ranges, showing (a) max- imum available capacity degradation, (b) nominal capacity degradation [47]	50
4.1 4.2	Schematic figure of the $\alpha - \lambda$ prognostic algorithm evaluation metric with $\alpha = 0.2$ [110] Schematic figure of the convergence prognostic algorithm evaluation metric showing three cases	60
	that converge at different rates [110]	61
5.1	Analysis of value for different maintenance strategy levels [53]	66
7.1	Project timeline in the form of a Gantt Chart	81
1.1	Stabilisation of mean distance results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range	86
1.2	Quantile-Quantile (Q-Q) plots for outputs of 1000 batteries tested through Monte Carlo simula- tion for mixed range	87
1.3	Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range	88
1.4	Quantile-Quantile (Q-Q) plots for outputs of 300 batteries tested through Monte Carlo simula- tion for short range	89
1.5	Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium range	90
1.6	Quantile-Quantile (Q-Q) plots for outputs of 300 batteries tested through Monte Carlo simula-	01
1.7	Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation	51
1.0	for SOC 100%, SOC 80% and mission-based strategy for long range	93
1.0	tion for long range	94
1.9	Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for	00
1.10	Distribution of Sustainability and Cost Assessment (Battery Usage) results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy	96
1.11	for mixed range	96
	tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range	97

1.12	Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short	
	range	97
1.13	Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for	
	medium range	98
1.14	Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested	
	through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium	
	range	98
1.15	Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 300 batteries	
	tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for	
	long range	99
1.16	Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested	
	through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long	
	range	99
1.17	Sensitivity Analysis (Part 1) - Stabilisation of mean distance flown per battery through Monte	
	Carlo runs per sensitivity parameter varied for minimum and maximum values	100
1.18	Sensitivity Analysis (Part 2) - Stabilisation of mean distance flown per battery through Monte	
	Carlo runs per sensitivity parameter varied for minimum and maximum values	101

List of Tables

3.1	Overview of common Lithium-ion battery types listing the name, abbreviation and formula [23]	45
4.1	Overview terms and notations for prognostic data-driven algorithm evaluation metrics [110]	58
1.1	Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and $Efficiency$ outputs of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range	87
1.2	Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and $Efficiency$ outputs of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed	07
1.3	range	87
	range	89
1.4	Unpaired one-sided T-test results for <i>B</i> , P_{total} and <i>Efficiency</i> outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range	90
1.5	Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and $Efficiency$ outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium	50
1.6	range	91
1.7	range	92
	range	92
1.8	Unpaired one-sided T-test results for <i>B</i> , <i>P</i> _{total} and <i>Efficiency</i> outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long	
1.9	range	93
	medium and long range	95

List of Abbreviations

Abbreviations

ANAC	Agência Nacional de Aviação Civil
BASA	Bilateral Aviation Safety Agreements
BM	Breakdown Maintenance
BMS	Battery Management System
BOL	Beginning-of-Life
CALCE	Center for Advanced Life Cycle Engineering
CBA	Cost-Benefit Analysis
CBM	Condition Based Maintenance
CC/CV	Constant Current and Constant Voltage
СМ	Corrective Maintenance
СМТ	Certification Management Team
CRA	Cumulative Relative Accuracy
DBM	Detection-Based Maintenance
DES	Discrete Event Simulation
DOC	Direct Operating Costs
DOD	Depth-of-Discharge
DOM	Design-Out Maintenance
EASA	European Aviation Safety Agency
eAV	Electrical Aerial Vehicle
EOD	End-of-Discharge
EOL	End-of-Life
EOP	End-of-Prediction
eUAV	Electrical Unmanned Aerial Vehicle
FAA	Federal Aviation Administration
FBM	Failure-Based Maintenance
FN	False Negative
FP	False Positive
FT	Failure Threshold
GHG	Greenhouse Gas

GWP	Global Warming Potential
ICAO	International Civil Aviation Organisation
IPA	Implementation Procedures for Airworthiness
JAA	Joint Aviation Authority
LCO	Lithium Cobalt Oxide
LFP	Lithium Iron Phosphate
Li-ion	Lithium-ion
Li-Po	Lithium-Polymer
LMO	Lithium Manganese Oxide
LR	Linear Regression
LTO	Lithium Titanate Oxide
MAD	Mean Absolute Deviation
MAPE	Mean Absolute Percentage Error
МС	Monte Carlo
MCC	Multistage Constant Current
MdAD	Median Absolute Deviation
MEA	More Electric Aircraft
MLR	Multiple Linear Regression
MSE	Mean Squared Error
MTBF	Mean Time Between Failure
MTBUR	Mean Time Between Unit Replacements
NASA	National Aeronautics and Space Administration
NCA	Nickel Cobalt Aluminum Oxide
NMC	Nickel Manganese Cobalt Oxide
NPV	Net Present Value
OAT	One-at-a-Time
OCV	Open Circuit Voltage
OTS	Off-the-Shelf
PDF	Probability Density Function
PdM	Predictive Maintenance
PH	Prognostic Horizon
PHM	Prognostics and Health Management
PM	Preventive Maintenance
РуВаММ	Python Battery Mathematical Modelling

Q-Q	Quantile-Quantile
RA	Relative Accuracy
RMSE	Root Mean Squared Error
ROI	Return On Investment
RTF	Run-To-Failure
RUL	Remaining Useful Life
SA	Sensitivity Analysis
SEI	Solid Electrolyte Interphase
SHM	Structural Health Monitoring
SNL	Sandia National Laboratories
SOC	State of Charge
SOH	State of Health
SPM	Single Particle Model
SSD	Sample Standard Deviation
TBM	Time-Based Maintenance
TCCA	Transport Canada Civil Aviation
UAV	Unmanned Aerial Vehicle
UBM	Usage-Based Maintenance
Symbols	
\widetilde{Cost}_{BOL}	BOL costs [\in] / flying time [s]
$\widetilde{Cost}_{charge}$	Charging costs [\in] / flying time [s]
$\widetilde{CO}_{2, BOL}$	BOL emissions $[kg] / flying time [s]$
$\widetilde{CO}_{2,\ charge}$	Charging emissions $[kg]$ / flying time $[s]$
\tilde{P}_{total}	Charging power $[kWh]$ / flying time $[s]$
CO _{2, BOL}	BOL emissions [kg]
CO _{2, charge}	Charging emissions [kg]
$Cost_{BOL}$	BOL costs [€]
$Cost_{charge}$	Charging costs [€]
d_{max}	Maximum target distance [m]
d_{min}	Minimum target distance [m]
d _{target}	Target distance [<i>m</i>]
Ι	Current [A]
Ic	Current at cruise [A]
I_l	Current at landing [A]

$I_{t/o}$	Current at take-off [A]
P _{total}	Charging power [kWh]
Q	Capacity [<i>Ah</i>] or [<i>C</i>]
Q_0	Initial capacity at beginning of life [Ah] or [C]
Q_{max}	Maximum capacity at a cycle $[Ah]$ or $[C]$
R	Resistance [Ω]
t _c	Time duration of cruise [<i>s</i>]
t_l	Time duration of landing [s]
$t_{t/o}$	Time duration of take-off [s]
V	Voltage [V]
V _{exp}	Expected DOD voltage $[V]$
V _{max}	Maximum voltage [V]
V _{min}	Minimum voltage [V]
Vreq	Required voltage $[V]$
V _{SOC 30%}	Battery voltage when SOC = 30% [V]

Introduction

Given the growing use of Lithium batteries in electric Unmanned Aerial Vehicles (eUAV), it is chosen to conduct research regarding battery health management strategies. One of the biggest challenges that the aerospace industry currently faces is the rapid rate of battery degradation which limits a battery's capacity and lifetime. In this report, the application of an eUAV battery health management strategy that reduces the severeness of battery degradation is explored.

A potential method to decrease the battery degradation rate is by minimising the average State of Charge (SOC) and Depth-of-Discharge (DOD) levels. Translating this to eUAV practices, a 'mission-based' battery health management strategy is proposed that charges the battery to the estimated required level of SOC to complete a flight. Moreover, the impact of varying DOD ranges is reviewed by regulating the flying range.

To evaluate the performance of the mission-based strategy, it is compared to two other battery health management approaches that charge the eUAV battery to 100% and 80% SOC. Although a more advanced model is recommended before applying this model to real-life applications, this research provides insights and tools to support operators in exploring and comparing the benefits of battery health management strategies.

The research is carried out as part of a master thesis of the Air Transport Operations track at the Faculty of Aerospace Engineering at Delft University of Technology. This thesis report has the following structure. Part I presents the scientific paper of the research. Thereafter, Part II contains a report of the Literature Study, providing background information on the topic of maintenance in the aerospace industry as well as battery health management and sustainability and cost assessment practices. Lastly, Part III provides supporting work for the research presented in the scientific paper. An elaborate statistical analysis of the model is first provided in Appendix 1, followed by an overview of the model limitations and recommendations for future studies in Appendix 2.

I

Scientific Paper

Sustainability and Cost Assessment of Lithium Battery Health Management Strategies for Electric Unmanned Aerial Vehicles

P.J. Vossen*

Air Transport Operations, Faculty of Aerospace Engineering, Delft University of Technology, Delft, The Netherlands March 2022

Abstract

In this research, a sustainability and cost assessment of battery health management strategies applied to Lithium batteries of an electric Unmanned Aerial Vehicle (eUAV) is performed. A mission-based strategy is proposed with the aim to elongate battery lifetime. With this strategy, the battery is charged to the estimated State of Charge (SOC) level required to complete the next flight. The mission-based strategy is compared to two other strategies: the SOC 100% strategy that always fully charges the battery before flight, and, the SOC 80% strategy that charges that battery to 80% before flying. The three strategies are tested for a variety of flight distances. The battery model is simulated using Python Battery Mathematical Modelling (Py-BaMM). A Monte Carlo (MC) simulation is run to review the response to uncertainties in initial battery compositions and operating conditions. Ultimately, the strategies are evaluated on environmental impact, financial costs and flying efficiency. The results show that the mission-based strategy outperforms the SOC 100%, yielding lower emissions and costs and higher flying efficiency performance. However, depending on the range flown, the SOC 80%shows environmental, cost and flying efficiency benefits that challenge the relevance of implementing a mission-based battery health management strategy.

Index Terms — electric Unmanned Aerial Vehicle (eUAV), Monte Carlo (MC), Lithium battery, health management, sustainability

1 Introduction

In line with the sustainability trends, an increase in studies reviewing the use of alternative propulsion systems such as rechargeable batteries is observed [18, 19, 60, 79, 83]. Within the aviation industry, electrically powered aerial vehicles do not only bring along environmental benefits, but also operational advantages such as noise reduction and reactive thrust [17, 83], making them are a popular subject for research.

The prevailing used battery type in aviation is Lithium [56, 60, 68, 78, 79]. The scope of this research is set to Lithium batteries that are widely used in Unmanned Aerial Vehicles (UAVs) [8, 18, 83]. When an UAV is fully electric, it is also referred to as an electric UAV (eUAV). In aviation, UAVs and eUAVs are favoured for a multitude of reasons such as their low operating costs and ability to fly in dangerous or extreme conditions [3, 33, 54, 75]. Applications include military, search and rescue and agricultural operations.

One important field of study for batteries revolves around their performance and reliability, characterised by the battery's capacity and internal resistance [49]. The capacity defines how much energy can be stored in the battery, while the internal resistance sets a limit on the maximum level of power that the battery is able to deliver. Of these two parameters, capacity is commonly used as the main component to evaluate a battery's performance and reliability [89]. These characteristics form the guiding principles for safe flight.

To improve the performance and reliability of eUAV batteries, several different battery health management approaches can be identified. Firstly, most batteries onboard an eUAV are linked to a Battery Management System (BMS) that monitors several parameters including the battery voltage, current, temperature and State of Charge (SOC) [12, 47, 93]. The BMS provides a battery protection by ensuring it does not exceed operating limits. Depending on the level of sophistication, a BMS could also be able to determine the State of Health (SOH) and/or make End-of-Discharge (EOD) and End-of-Life (EOL) estimations. Secondly, in recent years, the benefit of implementing Prognostics and Health Management (PHM) to enhance battery reliability and capabilities is being reviewed [64]. For batteries, PHM focuses on predictions of parameters such as SOC, EOD, SOH and EOL [17, 89].

During one discharge cycle, a battery's available capacity decreases with respect to the maximum capacity at the start of the cycle [9, 89]. This is reflected by the SOC, with a battery's SOC being 100% when it is fully charged. Additionally, the maximum achievable capacity degrades over the course of multiple cycles compared to the battery's initial capacity at Beginning-of-

^{*}MSc Student, Air Transport and Operations, Faculty of Aerospace Engineering, Delft University of Technology

Life (BOL). The status of maximum achievable capacity is known as SOH, with a battery having 100% SOH at BOL. The reduction of capacity due to cycling, also known as battery ageing or degradation, does not only affect the system's performance and reliability, but also has environmental and financial consequences as this phenomenon is directly linked to the battery's lifetime [19, 83].

While battery ageing is unavoidable, there are several factors that accelerate the degradation process. For example, for cycle ageing, the severeness of SOH degradation is heavily driven by the operating temperature, rate of current, average SOC and Depth-of-Discharge (DOD) [30, 41, 53, 80, 91]. Here, the average SOC can be regulated by charging to a SOC level below 100%. Furthermore, the DOD relates to the amount of energy extracted from the battery. Regulating these parameters throughout eUAV operations could potentially lead to battery lifetime elongation.

To determine which health management strategy is most beneficial, an assessment of costs is a commonly applied technique to evaluate and compare different strategies [11, 90]. However, other metrics including effectiveness [62], scientific performance [70] as well as sustainability and/or social impact [28, 31] can also be reviewed. The selection and weights of the relevant parameters depend on the topic of concern and stakeholders that are involved.

Following the aforementioned trends of sustainability within aviation, it is chosen to conduct a research on the implementation of health management approaches for eUAV batteries. The research objective is to determine the environmental and costbenefit of a mission-based battery health management strategy for Lithium batteries for electric Unmanned Aerial Vehicles. Specifically, this research aims to achieve battery lifetime elongation by modelling a mission-based battery health management strategy that minimises the battery's average SOC by charging to a lower SOC before flight. The missionbased strategy is compared to two alternative health management strategies where the battery is always charged to 100% SOC, and a strategy in which the battery is charged to 80% SOC. The strategies are assessed for a set of different DOD ranges to investigate how the distance flown relates to the battery degradation rate. The battery models are tested by running a Monte Carlo (MC) simulation to review the response to small uncertainties. The MC incorporates stochastic inputs to represent battery manufacturing impurities in new batteries [6, 22, 61, 89] and in-flight variations in battery power usage and flying times [65]. Ultimately, an assessment of battery health management strategies' environmental emissions, monetary costs and efficiency with respect to downtime is carried out.

The academic contribution of this study consists of three parts. Firstly, although lower SOC and DOD levels are known to decrease battery degradation rate [30, 91], no literature is found in which this battery health management strategy is applied to eUAV applications. The second contribution lies in the proposed framework to evaluate and compare eUAV battery health management strategies which can be used in future eUAV battery analyses, covering three pillars including sustainability, financial costs and efficiency. Lastly, this research explores the use Python Battery Mathematical Modelling (PyBaMM) [77] to review new eUAV battery lifetime elongation methods. PyBaMM is a simulation package launched in 2021 that is able to efficiently solve battery models in Python. This tool has not been used to model Lithium batteries for eUAV applications before.

This paper has the following structure. First, related work on battery health management research is elaborated on in section 2. Subsequently, the methodology of this study is presented in several separate sections. A description of the mission profile set-up is given in section 3, followed by an elaboration of the framework of the eUAV battery simulation and health management strategies in section 4. Then, section 5 presents the assessment metrics used to review the battery model outputs. To conclude the methodology of this study, a brief overview of the experimental set-up is given in section 6. Section 7 presents the results of the experiment, after which the model is validated by assessing its robustness through the means of a sensitivity analysis in section 8. Finally, the research conclusion and future recommendations are summarised in section 9.

2 Related Work

Model choices for this research are substantiated by findings from literature. The following sections present examples of related work featuring eUAV battery health management and performance evaluation studies. Relevant research regarding battery health and simulation practices are first elaborated on in section 2.1. Secondly, examples of assessment frameworks are given in section 2.2.

2.1 Battery Health

Given the global shift towards more sustainable fuel alternatives, an increase in battery studies is observed [17, 18, 79, 83]. This section first reviews literature concerning battery health management practices related to Lithium batteries in section 2.1.1. Health management applications to enhance a battery system's performance and reliability is first elaborated on, after which sustainability-oriented health management opportunities are highlighted. Then, section 2.1.2 discusses several battery modelling approaches providing substantiation as to why PyBaMM is chosen to model the eUAV batteries in this study.

2.1.1 Battery Health Management

By applying health management to battery systems, operators aim to improve the system's reliability and performance by closely monitoring the battery's capacity and health [14, 23, 49, 54, 89]. For Lithium batteries, it is especially important to monitor both the SOC

and SOH, as voltage levels significantly drop when operated beyond their safety limits. For eUAV applications, the SOC threshold is equal to 30% to avoid the precipitous decrease in voltage that occurs beyond the knee point while also ensuring the batteries are able to provide the eUAV sufficient power to perform at least two additional landing attempts [37, 72, 84]. The battery's EOL is defined to be as soon as the SOH reaches 80% as flying with the battery beyond 70 - 80% SOH is considered to be highly unsafe [49, 85, 87, 89]. Alternatively, some researchers such as Viwanathan and Knapp [83] state that internal increase of resistance could define the EOL of an eUAV Lithium battery. For Lithium batteries, this is defined to be when the battery's internal resistance is twice the value it had at BOL [25, 49, 89]. Both capacity and resistance are therefore advised to be taken into account when determining an eUAV's battery's EOL.

Similar to determining the Remaining Useful Life (RUL) of a component for maintenance, most battery health management studies focus on predicting the moment at which the battery reaches EOD and EOL [7, 37, 68, 69, 75]. By incorporating prognostics, experts aspire to move away from relying on fixed flying time and lifetime prescriptions as these are highly conservative [17, 75].

Instead of exclusively applying battery health management methods to improve a battery's reliability, authors such as Iung and Levrat [40], advocate the integration of other factors such as minimal energy consumption or effectiveness. Especially in the view of sustainability, this offers great perspectives. For example, Xu et al. [91] and Gao et al. [30] explored the influence of different SOC and DOD levels on the rate of battery degradation. From their studies, the researchers found that battery ageing is less severe when the battery is cycled at a lower average SOC with minimum DOD ranges. These batteries have, however, been cycled under constant charging and discharging conditions. Subsequently, in this research paper, it is chosen to review how the lifetime of batteries operating eUAV flights would respond to lower SOC and DOD levels. Translating this to a battery health management approach, two aspects are regulated. Firstly, a battery is charged to a lower SOC level before flight (instead of 100%). In order to ensure that the eUAV is safely able to complete it's flight, this initial SOC level should be carefully estimated. In this research, this strategy is referred to as being 'mission-based'. Secondly, the DOD ranges are regulated by flying different distances and seeing how the battery degradation rate responds.

Rather than applying a potentially costly prognostic battery health management strategy, researchers such as Nair and Garimella [58] reviewed the option of setting the BMS limits to a minimum of 20% and maximum of 80% SOC. By doing so, a lower average SOC and DOD is also yielded. This strategy could be inputted as an alternative to always charging to SOC 100% before flight, and is addressed in this research as the SOC 80% strategy. By decreasing the average SOC and DOD range that an eUAV battery cycles through, the available capacity for flight is also reduced. Nevertheless, this battery health management strategy could still yield significant benefits when combining this charging and flying methodology with other pivotal solutions such as eUAV battery swapping or wireless charging at power lines [29, 51]. Additionally, the effects of average SOC and DOD on battery degradation could provide insightful understandings for optimisation problems concerning eUAV hub location [2] and last-mile delivery models with eUAVs and trucks [32, 57].

2.1.2 Battery Models

Battery health management models can be established with the use of electrochemical or data-driven methods [4, 14, 23, 49, 54, 89]. Commonly, data-driven approaches are favoured since electrochemical models are very complex. Moreover, data-driven approaches are beneficial as they allow the user to include a broad range of parameters such as voltage, current and temperature. A wide variety of data-driven techniques exist including regression, filtering, and machine learning applications.

To promote data-driven battery research, Dos Reis et al. [20] recently published a paper providing an overview of all available open-source Lithium battery data sets. Well-known data sets are for example from the Prognostics Center of Excellence department of National Aeronautics and Space Administration (NASA) [1], Center for Advanced Life Cycle Engineering (CALCE) [27] and Sandia National Laboratories (SNL) [45]. Drawbacks of using these data sets, however, are that the cycling conditions are casespecific and not directly applicable to eUAV operations.

Alternatively, software can be used to model batteries. An advantage of using a simulation programme is its flexibility to tailor battery specifications and cycling throughput values. On the downside, many software tools are difficult to synthesize with other data processing operating systems such as Matlab and Python or require costly licences. Mitigating these shortcomings, a package called 'PyBaMM' was recently launched by a community of researchers [77]. The package allows batteries to be modelled and simulated in a rapid and versatile manner. Users can either choose to exploit readily available Lithium or Lead-Acid battery parameter sets and ageing models, or implement their own models.

To model an eUAV flight, battery data is required. Although some research publications that provide documentation about the power usage for certain eUAV manoeuvres are available [5, 16, 94], these papers lack detailed battery information such as the voltage and current levels the battery outputs during operation. The open-source data set by Rodrigues et al. [65] does specify in-flight energy use and battery cycling information of a DJI Matric 100 eUAV quadcopter. Together with PyBaMM, this battery data is used to generate the flight profiles for the battery model in this study.

2.2 Sustainability and Cost Assessment

An assessment of battery health management strategies can be done to review the benefits associated with each approach. For the aerospace field specifically, however, authors such as Saxena et al. [71] state that the industry lacks a set of standardised performance metrics to evaluate prognostic strategies. Similarly, Liu et al. [50] conclude that battery health management studies can have a variety of main objectives. Hence, this research will combine a set of sustainability and cost assessment parameters that are relevant to evaluate and compare eUAV battery health management approaches.

For this study, the following evaluation metrics are included: environmental impact, financial costs and effectiveness. Each of these parameters is briefly introduced in the following section 2.2.1 to 2.2.3, along with a battery health management example from literature.

2.2.1 Environmental Impact

In recent years, authors such as Franciosci et al. [28] and Ghazi et al. [31] are promoting the use of more sustainability related assessment parameters, instead of solely focusing on financial and technical aspects. Factors they list to assess the environmental impact of a system are for example material resources, energy consumption, noise emissions, pollutant emissions and waste. Ghazi et al. [31] recognise the fact that the establishment of a method to compare the impact of different environmental parameters is a complex task. Often, the opinion of the stakeholders plays an important role in this evaluation process in order to conduct a (weighted) trade-off.

In a study by Koiwanit [44] assessing the environmental impacts of using eUAVs to deliver online shopping, it is estimated that 80% of an eUAV's battery Global Warming Potential (GWP) originates from carbon emissions. Other environmental impact parameters such as abiotic depletion and ecotoxicity are also evaluated, but suffer from incomplete data. The eUAV delivery system is evaluated with the use of a lifecycle analysis that includes the BOL, operation and maintenance and EOL activities. With the use of the CML2001 model [34], the author concludes that the production of parts at BOL sum up to be 99% of the model's total environmental footprint. Similarly, reviewing the life-cycle cost-benefit of a battery storage system, Li et al. [48] concluded that BOL production emissions equal 70-75% of the battery's total emissions, followed by operation emissions and EOL emissions equal to 20-25% and 5%, respectively.

To determine the emissions for production, estimations based on the battery specifications and energy in kWh are commonly made [35, 52, 66]. For operation, charging emissions are often country-based and must therefore be analysed locally [24, 48]. Note that emissions vary vastly depending on each battery type, application and location. Therefore, if possible, it is highly recommended to validate results before drawing conclusions.

On a larger scale, Goodchild and Toy [32] and Stolaroff et al. [76] compared Greenhouse Gas (GHG) emissions yielded from the use of eUAVs versus trucks for last-mile package delivery. Here, when charging an eUAV, they state the importance of considering the efficiency of transferring electricity from the power plant to its motors. On average, the studies suggest that this efficiency factor is equal to 0.78 [32, 76].

2.2.2 Financial Costs

When assessing economic costs, the most obvious main objective is to minimise the total costs. However, van den Bergh et al. [81] point out that other aims such as minimum downtime and minimal delays can also be distinguished. To quantify the costs, Kent and Murphy [42] describe three methodologies: (1) an analogy approach where costs are based on historical or actual data, (2) a parametric method using formulas to cost relations and finally, (3) an engineering estimation concept through the use of an extensive cost breakdown. To review the financial costs over the lifetime of a component, metrics such as Return on Investment (ROI), Payback Period and Net Present Value (NPV) are commonly applied [26, 63, 90].

In the light of reviewing the costs of Lithium battery health management strategies, Liu et al. [50] assessed the economic performance of several battery charging approaches. In their model, the authors included the initial cost of the battery, the charging costs of the battery as well as the resale opportunities at the battery's EOL. In another study, Li et al. [48] analysed the costs of a battery storage system and concluded that the system's life-cycle costs mainly consisted of the initial costs at BOL and charging costs.

2.2.3 Efficiency

In a study reviewing maintenance approaches, Pecht and Rafanelli [62] analysed the effectiveness of a system to take its availability, dependability and capability into account. Here, availability refers to the system being usable at the beginning of a period of desired usage. Moreover, dependability indicates the probability of the system failing during operation. Lastly, the systems capability shows at which level of performance the system is able to carry out its function.

3 Mission Profiles

In the previous sections, an introduction and substantiation of the research topic has been given. The following sections 3, 4, 5 and 6 describe the methodology of this study regarding eUAV flights, battery simulation, performance assessment and MC set-up.

In order to test the battery health management strategies defined for this research, this section presents

the framework of the missions that the eUAV executes. To simulate a cycle that a battery undergoes during a flight, eUAV flight data tracked for a DJI Matrice 100 quadcopter by Rodrigues et al. [65] is used. In section 3.1 an overview of the flight layout is presented. Secondly, the flight distance flown is discussed in section 3.2, after which the input data for the flight profiles is elaborated on in section 3.3.

3.1 Flight Outline

To model discharge cycles that an eUAV battery undergoes, flight missions are defined. As the focus of this paper lies on determining the benefit of applying a mission-based approach for eUAV battery health management, a flying plan is proposed during which an eUAV performs 'simple' manoeuvres to reach a randomly assigned 'target'. The eUAV flies to the target by carrying out actions including take-off, cruise and landing. For each flight, the following steps are executed:

- 1. The eUAV is initially positioned at a hub where it is able to charge its battery.
- 2. Then, a mission target is randomly generated within the maximum defined distance that the eUAV is able to fly, given the maximum capacity of the battery.
- 3. Once the target is known, the eUAV flies to the target and back to the hub, by:
 - performing a vertical take-off to the desired cruise altitude,
 - cruising at the set altitude to the target,
 - vertically landing at the target,
 - resting for a short time,
 - executing a vertical take-off to the inputted cruise altitude,
 - cruising at the set altitude back to the charging hub, and,
 - vertically landing at the hub.
- 4. A new target mission is then generated (continue to loop from step 1).

The maximum distance that the eUAV is able to fly is limited by the battery capacity. In the model set-up for this research, the maximum distance is defined conservatively to ensure that the battery is always able to safely complete each flight. The flight distances covered in this research are briefly elaborated on in the next section.

3.2 Flight Distance

In this study, the distance that the eUAV flies is varied to review the impact of DOD on battery degradation for eUAV applications. From literature [30, 91] it is found that smaller DOD ranges result in less severe ageing for batteries cycled under constant discharge protocols. For eUAV battery operations, there are various manners to regulate the DOD such as altering the distance covered, speed flown or payload carried. However, because an amendment in speed or payload is paired with other ageing side-effects due to a change in battery load profiles. Thus, it is chosen to exclusively adjust the cruise distance flown.

The eUAV completes four sets of different ranges to explore the effect of varying the DOD: *mixed, short, medium* and *long* range. The target distance framework is further elaborated on in section 6.

3.3 Flight Data

In the DJI Matrice 100 quadcopter data [65] several flights are monitored exploring the influence of wind, cruise altitude, ground speed and payload weight on the eUAV performance.

The DJI Matrice has a 22.2 V Lithium battery with 4.5 Ah capacity. The quadcopter has a maximum cruise speed equal to 17 ms^{-1} and is able to fly 22 minutes provided it is not carrying a payload.

In order to simulate the battery flights in PyBaMM, the rates of current the battery provides the quadcopter during take-off, cruise and landing are required. Secondly, the time duration of a take-off and landing manoeuvre are quantified. A summary of the mean inputs per flight part is given in Table 1. Both the current and time values have a standard deviation equal to 2.6% [65]. Note that the cruise time is not fixed, as this depends on the target distance that is stochastically generated before each flight.

In order to obtain these data inputs, the DJI Matrice 100 data set is filtered on altitude, payload weight and ground speed. The lowest altitude from the DJI Matrice 100 tests, equal to 25 m, is chosen to allow for a longer residual flight time to explore the influence of flying different DOD ranges. Additionally, following the same reasoning, the payload is set to 0 kg. Finally, the average ground speed of 8 ms^{-1} is used.

Part	Current		Time duration	
	Symbol	Value [A]	Symbol	Value [s]
Take-off	$I_{t/o}$	23.5	$t_{t/o}$	12.0
Cruise	I_c	21.0	t_c	variable
Landing	I_l	19.5	t_l	25.0

Table 1: Flight profile input values

4 Battery Simulation

Now that the mission profiles are defined, this section presents the battery simulation set-up used to cycle the battery throughout the flights. To evaluate the performance of the mission-based battery health management strategy, it is compared to a SOC 100% and SOC 80% strategy. First, section 4.1 describes the general battery model used for simulations in Python. Then,



Figure 1: Schematic overview of SOC 100%, SOC 80% and mission-based strategy battery models

the SOC 100%, SOC 80% and mission-based battery health management strategies are elaborated on in section 4.2.

4.1 Battery Modelling

The battery is modelled with the use of the Python package PyBaMM (version 21.10) [77]. To increase computational speed, the battery cells are represented by a Single Particle Model (SPM). It is chosen to replicate the Nickel, Manganese and Cobalt (NMC) Lithium battery using the parameter set provided by Mohtat et al. [55] as this battery cell has similar properties to the DJI Matrice battery used to fly the eUAV described in section 3.3. An overview of the battery parameter values defined by Mohtat et al. [55] is given in Table 21 in Appendix A.

In this model, the Lithium pouch cell from Mohtat et al. [55] has a 4.2 V maximum voltage (V_{max}) and a 3.0 V minimum voltage (V_{min}) . The nominal cell capacity is 5.0 Ah. This is slightly more than the 4.5 Ah DJI Matrice battery, enabling the eUAV to fly longer. To reach the 22.2 V delivered by the DJI Matrice batteries, multiple Lithium NMC pouch cells can be placed in series to form a battery pack. To rule out ageing influences between battery cells, however, this model only reviews the performance of a single cell.

Battery ageing is modelled with a Solid Electrolyte Interphase (SEI) kinetic rate equal to $1 \cdot 10^{-14} ms^{-1}$ derived by Yang et al. [92]. Here, the ethylene carbonate reaction is limited following the SEI formation reaction. The battery's EOL is defined when SOH is

80% to avoid a critical voltage drop. The internal resistance is not defined to be the limiting factor as test runs showed that this parameter increased with a maximum of +20% upon reaching 80% SOH.

As PyBaMM is deterministic, several small perturbances are inputted into the model to mirror real-life parameter variation observed in literature. Table 22 in Appendix A lists these non-deterministic errors that applied to the battery chemistry and BMS readings. These stochastic inputs are also briefly addressed in section 6 and further explored when validating the model in section 8.

A schematic flowchart of the steps the battery model undergoes is given in Figure 1. This overview includes three separate steps that are executed depending on the battery health management strategy that is in place. These three strategies are: (1) SOC 100%, (2) SOC 80% and (3) mission-based. The battery model is almost identical for these strategies, except for the *experiment* that the battery executes for each flight. A detailed description of experiments 1, 2 and 3 is given in section 4.2. Below is a brief description of each step in depicted in Figure 1:

- 1. New battery a new battery is inputted into the model to be tested until it reaches SOH 80%. The number of batteries that are tested for each strategy is determined by the MC simulation setup which is further discussed in section 6.
- 2. New target while SOH $\geq 80\%$, new targets are generated for the battery to fly to.

- 3. Experiment depending on the battery health management strategy (SOC 100%, SOC 80% or mission-based), the battery is either first charged and then flown, or flown directly.
- 4. **SOH** \geq 80% SOH is checked after each flight:
 - i if SOH $\geq 80\%$, a new target is generated at d_{target} distance and the battery undergoes another cycle.
 - ii if SOH $\leq 80\%$, this battery is discarded and a new battery is inputted in the model to be tested.

Translating the flowchart steps into a simulation model, the skeleton of the battery model developed in Python using PyBaMM is given in Algorithm 1. The following notes apply:

- The normally distributed input values in lines 5, 7, 8 and 17 refer to the specifications listed in Table 22 in Appendix A.
- Flights, *f*, are counted for each mission completed.
- During each flight, the *experiment* run for the simulation in line 27 depends on the battery health management strategy that is in place. Note that in this context, 'experiment' refers to the charge/discharge cycle that the battery undergoes in PyBaMM. The variation in types of simulation experiments are elaborated on in section 4.2.
- After an experiment has been simulated, sim(), the battery's inner state has been modified from $battery_f$, start to $battery_f$, end.
- Finally, when the battery has reached SOH = 80% the *output* solution is a set of flights executed by the inputted battery. The data includes detailed battery state information such as voltage, current and capacity levels for each flight.

4.2 Battery Health Management Strategies

As stated in the previous section, the *experiment* a battery performs during each cycle depends on the strategy that is being evaluated. In this section, the three battery health management strategies (1) SOC 100%, (2) SOC 80% and (3) mission-based are described in detail. The frameworks of the experiments associated with each strategy is first given in section 4.2.1. Thereafter, the methodology of the estimation of the required amount of SOC to safely complete a mission is presented in section 4.2.2.

4.2.1 PyBaMM Experiments per Strategy

The definitions of the *experiment* for the SOC 100%, SOC 80% and mission-based strategy are presented in Algorithms 2, 3 and 4, respectively. These experiments are inputted into the main battery model in Algorithm 1 when 'flying = True' which starts in line 21.

Algorithm 1 Skeleton of the battery model using Py-BaMM

- 1: Input model $\leftarrow SPM$
- 2: Input model \leftarrow ec reaction limited
- 3: **Input** parameter \leftarrow *Lithium-ion Mohtat2020*
- 4: Input parameter \leftarrow SEI kinetic rate Yang2017
- 5: **Input** parameter $\leftarrow capacity_{f=0}, normal(\mu, \sigma)$
- 6: **Input** electrode \leftarrow *Lithium-ion ElectrodeSOH*
- 7: Input electrode $\leftarrow Vmin, normal(\mu, \sigma)$
- 8: Input electrode $\leftarrow Vmax, normal(\mu, \sigma)$

```
9: f = 0
```

18:

19: 20:

31:

- 10: battery_{f=0} \leftarrow model, parameter, electrode
- 11: while SOH $\geq 80\%$ do
- 12: new mission = True
- 13: flying = False
- 14: mission complete = False
- 15: **if** new mission = True **then**
- 16: $d_{target} \leftarrow random \ target \ within \ range$ 17: parameter $\leftarrow SEI \ kinetic \ rate \ normal(\mu, \sigma)$

 $new_mission = False$ flying = True end if

```
if flying = True then
21:
          if f = f_0 then
22:
23:
             battery_{f,start} \leftarrow battery_{f=0}
24:
          else
             battery_{f,start} \leftarrow battery_{f-1,end}
25:
          end if
26:
27:
          experiment \leftarrow d_{target}
28:
          battery_{f,end} \leftarrow sim(battery_{f,start}, experiment)
          f \leftarrow f + 1
          flying = False
29:
          mission complete = True
30:
```

```
end if
```

```
32: if mission_complete = True then
33: SOH \leftarrow capacity_{f,end}/capacity_{f=0}
```

```
34: mission complete = False
```

```
35: new mission = True
```

```
36: end if
```

```
37: end while
```

38: **Output** model \leftarrow solution

For the SOC 100%, SOC 80% and mission-based strategy, the experiments includes charging ('Charge') and in flight ('Fly') discharging commands that PyBaMM uses to cycle the battery. These two processes are further elaborated on below. Note that both the SOC 80% and mission-based battery

health management strategy allow the eUAV to fly directly if it has sufficient SOC from the previous flight.

Charge

The battery cell is charged following a standard Constant Current Constant Voltage (CC/CV) protocol. The CC/CV charging steps inputted in the *experiment* are presented in Algorithm 5. The SOC_{start} values derived for the SOC 100%, SOC 80% and mission-based strategy are first converted to V_{start} V, as PyBaMM requires voltage inputs to charge the battery. Then, the battery is charged at 1C until V_{start} V (CC step) and held at V_{start} V until C/50 (CV step).

Algorithm 2 SOC 100% battery strategy			
- Experiment 1			
1: $SOC_{start} \leftarrow 100\%$			
2: if flying = $True$ then			
3: experiment $\leftarrow Charge(SOC_{start})$			
4: experiment $\leftarrow Fly(d_{target})$			
5: end if			

Algorithm 3 SOC 80% battery strategy - *Experiment 2*

1: if flying = True then Estimate $SOC_{req} \leftarrow d_{target}, SOH$ 2: if $SOC_{reg} > 80\%$ then 3: $SOC_{start} \leftarrow 100\%$ 4: experiment $\leftarrow Charge(SOC_{start})$ 5:experiment $\leftarrow Fly(d_{target})$ 6: else if $SOC_{req} \le 80\%$ and SOC(t) < 80% then 7: $SOC_{start} \leftarrow 80\%$ 8: experiment $\leftarrow Charge(SOC_{start})$ 9: experiment $\leftarrow Fly(d_{target})$ 10: 11: else 12:experiment $\leftarrow Fly(d_{target})$ end if 13: 14: end if

Algorithm 4 Mission-based battery strategy - *Experiment 3*

1: if flying = True then Estimate $SOC_{req} \leftarrow d_{target}, SOH$ 2: 3: if $SOC(t) < SOC_{req}$ then 4: $SOC_{start} \leftarrow SOC_{req}$ experiment $\leftarrow Charge(SOC_{start})$ 5: experiment $\leftarrow Fly(d_{target})$ 6: 7: else experiment $\leftarrow Fly(d_{target})$ 8: end if 9: 10: end if

As described in the mission profile part in section 3, a flight consists of multiple steps in order to fly from the hub, to the target and back to the hub. These steps are presented in Algorithm 6. The target distance (d_{target}) is converted to cruise time (t_c) with the use of the set 8 ms^{-1} ground speed. The current throughput rates applied during take-off $(I_{t/o})$, cruise (I_c) and landing (I_l) , as well as the times to perform take-off $(t_{t/o})$ and landing (t_l) are in line with the DJI Matrice values observed in literature [65] as listed in Table 1 which is stochastically generated with a standard deviation equal to 2.6% for each flight.

Algorithm	5	CC/CV	charge in	n <i>experiment</i>	
-----------	----------	---------------------------	-----------	---------------------	--

1: $V_{start} \leftarrow SOC_{start}$

2: Charge $(V_{start}) \leftarrow \left\{ \begin{array}{l} Charge \ at \ 1C \ until \ V_{start}V, \\ Hold \ at \ V_{start}V \ until \ C/50 \end{array} \right\}$

Algorithm 6 Fly in experiment

1: $t_c \leftarrow d_{target}$

2: Fly(d_{target}) \leftarrow $\begin{cases}
Discharge at 0.1A \text{ for } 3s, \\
Discharge at I_{t/o} A \text{ for } t_{t/o} s, \\
Discharge at I_c A \text{ for } t_c s, \\
Discharge at I_l A \text{ for } t_l s, \\
Discharge at 0.1A \text{ for } 3s, \\
Rest \text{ for } 3 \text{ minutes}
\end{cases}$

4.2.2 Estimation of Required State of Charge

Both the SOC 80% and mission-based strategy estimate the required SOC for the eUAV to complete the flight before flying. The establishment of this estimation is done using Linear Regression (LR) and Multiple Linear Regression (MLR). Because PyBaMM requires voltage as an input for charging practices, it is required to translate SOC to voltage levels, whilst keeping in mind that the voltage relation changes as the battery ages [72].

Equation 1 gives the formula used in the battery health management models to derive the required voltage level (V_{req}) for the SOC 80% and mission-based battery health management strategy. This method is based on an approach presented by Viswanathan et al. [84]. The V_{req} is the voltage level to which the battery must be charged to safely complete the mission. In this equation, $V_{SOC 30\%}$ is the voltage at which the battery has 30% SOC. The expected DOD voltage based on the inputted target distance is represented by V_{exp} .

$$V_{req} = V_{SOC\ 30\%} + V_{exp} \tag{1}$$

Voltage for 30% SOC ($V_{SOC 30\%}$) The level at which $V_{SOC 30\%}$ is incorporated because 30% SOC is the minimum SOC threshold that the battery must have to ensure safe flight [37]. The determination of the $V_{SOC 30\%}$ level is given in Algorithm 7.

The number of batteries tested to determine the $V_{SOC\ 30\%}$ levels for different SOH states, along with the corresponding number of data points, is listed in Table 2. The simulations are run with the battery and BMS specifications listed in Table 21 and 22 in Appendix A. To capture the $V_{SOC\ 30\%}$ values on the voltage curves as the battery ages, the battery is cycled from its minimum to its maximum voltage until SOH = 80%. The battery's SOH is reviewed at the start and end of each cycle, c. The cycling experiment in line 17 is given in Algorithm 8. To ensure that the battery ageing process is similar to the battery model used for flying, the same CC/CV charging approach is used to charge the batteries.

After the model output values for $V_{SOC~30\%}$ from 100 batteries at different SOH stages have been gathered, a LR is performed through regression in Python. For this LR, the battery's SOH is inputted due to the fact that the batteries' voltage level curves decrease as the battery degrades. Equation 2 is yielded to calculate $V_{SOC~30\%}$. This LR equation has an R^2 value of 0.432.

$$V_{SOC\ 30\%} = 0.397 \cdot SOH + 3.30 \tag{2}$$

Expected Voltage (V_{exp})

Secondly, the V_{exp} is added to the determined $V_{SOC 30\%}$. In order to derive the voltage DOD for a variety of distances flown, the model described in Algorithm 1 is used. Naturally, a higher V_{exp} relates to a longer flight distance. Again, this voltage model is simulated with the battery and BMS inputs listed in Table 21 and 22 in Appendix A are simulated. The total number of batteries tested is listed in Table 2. In this model, the SOC 100% experiment described in Algorithm 2 is inputted into the battery simulations. The number of data points outputted from these runs is significantly more compared to the $V_{SOC 30\%}$ model due to the longer lifetime as a result of a smaller DOD.

For V_{exp} , two variables are fed into the MLR, namely SOH and target distance. The SOH is inputted in order to account for the change in voltage curves as the battery ages, while the target distance (d_{target}) is used to determine what the DOD of the battery voltage will be. The regression results in Equation 3 which can be used to determine V_{exp} . The R^2 value of this MLR equation is 0.857.

 $V_{exp} = -0.0861 \cdot SOH + 0.0003 \cdot d_{target} + 0.195 \quad (3)$

Voltage level	Number of batteries	Data points
$V_{SOC\ 30\%}$	1000	4442
V_{exp}	100	9067

Table 2: Voltage level tests

Algorithm 7 LR required SOC

- Model for Voltage at which SOC = 30% (V_{SOC 30%})

- 1: Input model $\leftarrow SPM$
- 2: **Input** model \leftarrow *ec* reaction limited
- 3: **Input** parameter \leftarrow *Lithium-ion Mohtat2020*
- 4: Input parameter \leftarrow SEI kinetic rate Yang2017
- 5: **Input** parameter $\leftarrow capacity_{f=0}, normal(\mu, \sigma)$
- 6: **Input** electrode \leftarrow *Lithium-ion ElectrodeSOH*
- 7: Input electrode $\leftarrow Vmin, normal(\mu, \sigma)$
- 8: Input electrode $\leftarrow Vmax, normal(\mu, \sigma)$

```
9: c = 0
```

10: battery_{f=0} \leftarrow model, parameter, electrode

11: while SOH $\geq 80\%$ do

- 12: **if** $c = c_0$ **then**
- 13: battery_{c,start} \leftarrow battery_{c=0}
- 14: else
- 15: battery_{c,start} \leftarrow battery_{c-1,end}
- 16: **end if**
- 17: battery_{c,end} \leftarrow sim(battery_{c,start}, experiment) $c \leftarrow c + 1$
- 18: SOH $\leftarrow capacity_{c,end}/capacity_{c=0}$

19: end while

20: **Output** model \leftarrow solution

Algorithm 8 LR required SOC - *Experiment for* V_{SOC 30%}



5 Sustainability and Cost Assessment

After a MC simulation of the batteries has been run, an assessment is performed comparing the SOC 100%, SOC 80% and mission-based eUAV battery health management strategies. In total, three pillars are assessed: *environmental impact, financial costs*, and, *efficiency*. These factors are explained in more detail in section 5.1, 5.2 5.3, respectively. The assessment of the results evaluates these three pillar separately as a trade-off requires additional stakeholder information to determine the weight of each parameter.

5.1 Environmental Impact

The environmental impact related to longevity and energy consumption are reviewed. These are sustainability factors derived from literature [28, 31, 40] and chosen due to the availability of data. The environmental impact is quantified through carbon emissions, as these make up for 80% of a battery's Global Warming Potential (GWP) score [44]. The calculations of the carbon emissions are presented hereafter.

The carbon emissions that arise from the BOL production processes $(CO_{2,BOL} [kg])$ account for the waste and longevity sustainability component. These emissions are determined using Equation 4. In this equation, battery's energy at BOL $(E_{BOL} [kWh])$. The average E_{BOL} for the batteries inputted in this model in PyBaMM is 0.0190 kWh. The BOL carbon emissions $(CO_{2, BOL kWh} [kg/kWh])$ equal 143.7 kg of carbon per kWh battery capacity. This is derived from averaged BOL emission data for lithium batteries with NMC chemistry as the values of the BOL CO_2 emissions strongly vary per source [35, 52]. The EOL emissions are disregarded due to a lack of consistency in data and situational influences [38, 46].

$$CO_{2, BOL} = E_{BOL} \cdot CO_{2, BOL kWh} \tag{4}$$

During operation, the energy consumption throughout the activity of charging is closely monitored and translated to carbon emissions $(CO_{2,charge} [kg])$ with Equation 5. Here, the total charging power $(P_{total} [kWh])$ is reviewed for each battery health management strategy. The efficiency of transferring electricity (η) from the power plant to the drone motors is set to 0.78 [32, 76]. Moreover, charging carbon emissions $(CO_{2, charge \ kWh} [kg/kWh])$ associated with a mixed-use of green and grey electricity in the Netherlands are implemented into the model. This results in 0.49 kg CO_2 per kWh electricity used. [13, 15].

$$CO_{2, \ charge} = P_{total} \cdot \frac{1}{\eta} \cdot CO_{2, \ charge \ kWh}$$
 (5)

5.2 Financial Costs

Secondly, similar to the environmental impact, the financial costs of a battery at BOL, throughout operation and EOL can be considered for assessment [81]. EOL costs, however, are again disregarded as these are complex to determine in view of inconsistency of data and battery specification dependencies [38, 66]. For the SOC 80% and mission-based strategy, an investment cost also applies to initially establish, verify and validate the predictive approach [26]. In this study, however, the investment cost is neglected due to a lack of data. Consequently, the costs' ROI or Payback Period are also not assessed. By neglecting battery health management investment costs, the financial assessment of the mission-based and SOC 80% strategy is more optimistic. The equations to evaluate the remaining costs for the battery at BOL and during operation are discussed below.

The battery BOL costs $(Cost_{BOL} [\in])$ can be determined by either through an analogy approach reviewing average costs per kWh or with a parametric method by averaging Off-the-Shelf (OTS) products. As

costs per kWh vary [48], it is chosen to use the offthe-shelf price of the Lithium batteries used to model the eUAV flight mission profiles in section 3, which is equivalent to \in 199 [21].

To operate an eUAV battery, operation and maintenance costs mainly consist of energy consumption costs during charging [48]. To determine the charging costs ($Cost_{charge} \ [e]$) Equation 6 is used with the average electricity cost ($Cost_{charge \ kWh} \ [e / kWh]$) in the Netherlands as of 1 May 2021 [59], equal to $\in 0.23$ per kWh. In this equation P_{total} and η refer to the same charging power in kWh and electricity transfer process efficiency, respectively, as used in Equation 5.

$$Cost_{charge} = P_{total} \cdot \frac{1}{\eta} \cdot Cost_{charge\ kWh}$$
(6)

5.3 Efficiency

The last assessment parameter used to assess the battery health management strategies is efficiency which is related to the availability of the battery [62, 81]. Here, the 'downtime' of the battery health management strategies during the activity of charging are compared to the flight times. Naturally, a low downtime is desired, yielding a higher efficiency. To determine the efficiency of a strategy, Equation 7 is applied. Here, $Total \ flight \ time$ refers to the sum of the time the battery is used to fly, and $Total \ time$ represents the sum of the flying and charging time each battery undergoes.

$$Efficiency = \frac{Total \ flight \ time}{Total \ time} \tag{7}$$

Depending on the stakeholders involved, efficiency could be reflected in operational costs. This is, however, case-specific and beyond the scope of this study.

6 Experimental Set-up

Sections section 3 to 5 provided background information for the mission profiles, battery model and sustainability and cost assessment framework. In this section, the experimental set-up of this research is elaborated on. Details concerning the stochastic generation of mission targets is first presented in section 6.1, followed by an overview of the MC simulation set-up in section 6.2.

6.1 Mission Targets

During a flight, the eUAV is modelled to fly from the hub to a mission target and back. The minimum distance (d_{min}) to a target to is positioned 20 *m* from the hub such that the eUAV approximately covers the same distance horizontally as vertically. The maximum distance that the eUAV is able to fly is governed by the battery characteristics and flight profiles. The battery in PyBaMM has a capacity of 5.0 *Ah*. Following the flight instructions in which the battery provides sufficient energy for the eUAV to fly to a randomly assigned target location and back to the hub while also maintaining a SOC level above the 30% threshold, the maximum target distance (d_{max}) is defined to be 1000 m.

To analyse the influence of the battery DOD cycled through by each eUAV battery, four target distance ranges are defined. An overview of the minimum and maximum distance $(d_{min} \text{ and } d_{max})$ associated with each range is given in Table 3. The minimum distance of the *short* range equals the minimum distance inputted into the *mixed* battery model, while the maximum distance of the *long* distance is the same as the maximum range flown by the eUAV for the *mixed* set of missions. For each range, targets are generated following a uniform distribution such that each mission target has an equal chance of being visited.

Range	d_{min} [m]	d_{max} [m]
Mixed	20	1000
\mathbf{Short}	20	346
Medium	347	673
Long	674	1000

Table 3: Target distance ranges

6.2 Monte Carlo

To run the MC simulations, the battery model is tested using the SOC 100%, SOC 80% and mission-based strategy. Table 4 provides an overview of the number of batteries tested per strategy and range.

Range	SOC 100%	SOC 80%	Mission- based
Mixed	1000	1000	1000
Short	300	300	300
Medium	300	300	300
Long	300	300	300

Table 4: Number of batteries tested per strategy and range

The *mixed* target distance is initially tested for 1000 batteries. A statistical analysis presented in supporting work [86], is carried out to verify if the MC simulations of the independent battery runs converge to stable means. The total sum of target distances flown by each battery is reviewed as missions targets are stochastically generated as an 'input' to the battery model. The results show that the total sum of target distances flown by each battery for the SOC 100%, SOC 80% and mission-based strategy for the *mixed* range stabilises after approximately 300 runs. Hence, it can be concluded that sufficient MC simulations have been run. For the *short, medium* and *long* range tests, the MC simulations are run 300 times to decrease the total run time.

For the MC simulations for the *mixed*, *short*, *medium* and *long* range tests using the SOC 100%, SOC 80% and mission-based strategy, the following stochastic inputs are incorporated in the runs:

The MC incorporates stochastic inputs to represent battery manufacturing impurities in new batteries [6, 22, 61, 89] and in-flight variations in battery power usage and flying times [65].

• Initial battery parameters

Each battery that is newly inputted into the model, has the following stochastic initial properties as listed in Table 22 in Appendix A: initial capacity, initial maximum voltage and initial minimum voltage.

• In-flight variations

Every battery cycles through random flights depending on the target locations generated. Each mission, the current and time required to complete a take-off, cruise and landing manoeuvre varies around the mean value presented in Table 1 with a standard deviation equal to 2.6% [65]. Furthermore, the SEI kinetic rate constant depends on the battery health and is normally distributed using the inputs provided in Table 22 in Appendix A. Finally, the BMS reading errors for voltage, SOC and SOH which are presented in Table 22 in Appendix A, vary each flight affecting for example the determination of EOL at 80% SOH.

6.3 Sustainability and Cost Assessment

To assess the environmental impact and financial costs of the battery models, the output is reviewed for 'battery lifetime' and 'battery usage'. A brief description of these two characteristics is given below:

• Battery lifetime

Here, the impact of battery lifetime concerning BOL practices is assessed. From the sustainability and cost assessment parameters presented in section 5, the review of battery lifetime performance includes $CO_{2, BOL}$ and $Cost_{BOL}$. The EOL practices are disregarded due to lack of data.

In order to ensure that the SOC 100%, SOC 80% and mission-based output values are on the same scale, the BOL emissions and costs are expressed per total flying time flown per battery ($\widetilde{CO}_{2, BOL}$ [kg/s] and \widetilde{Cost}_{BOL} [ϵ/s]). Both these two parameters relate to the number of batteries used per flying time (\widetilde{B}), which will be discussed in the results in section 7.

• Battery usage

The evaluation of battery usage focuses on the performance during operation. From section 5 the parameters $CO_{2, charge}$ and $Cost_{charge}$ are incorporated to review the impact of charging. Also, the efficiency of flying versus downtime is included.

Again, to effectively compare the SOC 100%, SOC 80% and mission-based results, the emissions and costs parameters are reviewed per flying time $(\widetilde{CO}_{2, \ charge} \ [kg/s]$ and $\widetilde{Cost}_{charge} \ [€/s])$. These two metrics both relate to the total charge used per flying time per battery $(\widetilde{P}_{total} \ [kWh/s])$, which

is used to evaluate the battery usage performance in the next section. The flying efficiency results for battery usage can be compared directly.

7 Results

In this section, the results for battery health management strategies are presented. The results of the 1000 batteries tested for the *mixed* range are first discussed in section 7.1, after which the results of the 300 batteries simulated for the *short, medium* and *long* range are elaborated on in section 7.2. Finally, conclusions regarding the effect of regulating the average SOC and DOD on sustainability and cost performance is discussed in section 7.3.

Parameter	Mission-based vs. SOC 100%	Mission-based vs. SOC 80%	
\widetilde{B}	-20.1%	-5.7%	
\widetilde{P}_{total}	-6.4%	-5.2%	
Efficiency	+2.8%	-6.9%	

Table 5: Relative mean Battery Usage and Battery Lifetime performance of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range



Figure 2: Sustainability and Cost Assessment - Battery Lifetime results of 1000 batteries tested through Monte Carlo simulation for the SOC 100%, SOC 80% and mission-based strategy for mixed range

7.1 Mixed Range

First, section 7.1.1 elaborates on the evaluation with respect to battery lifetime in which the BOL emissions and costs are assessed. Secondly, section 7.1.2 presents the outcome of the battery usage assessment concerning emissions and costs of charging as well as the flying efficiency results. Sustainability and Cost Assessment (Battery Usage) for mixed range



Figure 3: Sustainability and Cost Assessment - Battery Usage results of 1000 batteries tested through Monte Carlo simulation for the SOC 100%, SOC 80% and mission-based strategy for mixed range

7.1.1 Battery Lifetime

It is expected that the mission-based health management strategy yields a longer battery lifetime due to the lower average SOC level. This implies that this model requires fewer batteries per flying distance flown (\tilde{B}) before reaching EOL at 80% SOH. A longer lifetime results in a relative decrease in BOL emissions and costs.

In order to compare if there is a significant difference in the average battery lifetime for the SOC 100%, SOC 80% and mission-based strategy, the hypotheses in Table 6 and Table 7 are tested for batteries used per flying time (\tilde{B}). From a statistical analysis performed in supporting work [86] it is concluded that the \tilde{B} output of the models follow a normal distribution. Hence, the hypotheses are tested for $\alpha = 1\%$ using an unpaired one-sided T-test.

Mission-based vs. SOC 100% - mixed range
$H_0: \widetilde{B}_{mission-based} \geq \widetilde{B}_{SOC \ 100\%}$
H1: \widetilde{B} mission-based < \widetilde{B} SOC 100%

Table 6: **Hypothesis** \tilde{B} - Mission-based vs. SOC 100% strategy for mixed range

The relative mean \tilde{B} performance from the MC simulations for the SOC 100% and mission-based strategy is displayed in Table 5. When comparing the independent samples of \tilde{B} paired with both strategies, it is observed that the mission-based approach requires approximately **20.1%** less \tilde{B} with 100% confidence [86]. Hence, H_0 for \tilde{B} in Table 6 is rejected.

From these results, it can be concluded that the mission-based battery health management strategy yields a longer battery lifetime with respect to the SOC 100% method. The battery life elongation as a result of applying the mission-based strategy is attributed to the fact that the battery is cycled around a lower average SOC.

Mission-based vs. SOC 80% - mixed range
$H_0: \widetilde{B}_{mission-based} \geq \widetilde{B}_{SOC 80\%}$
$H_1: B_{mission-based} < B_{SOC 80\%}$

Table 7: **Hypothesis** \widetilde{B} - Mission-based vs. SOC 80% strategy for mixed range

The relative average \tilde{B} from the SOC 80% and missionbased strategy MC simulations is displayed in Table 5. When comparing the \tilde{B} for these two strategies and reviewing the statistical analysis presented in supporting work [86], it is concluded with 100% confidence that the mission-based strategy has 5.7% reduced \tilde{B} . Hence, H_0 for \tilde{B} in Table 7 is rejected. Again, the conclusion can be drawn that the mission-based battery health management strategy results in battery lifetime elongation compared to the SOC 80% method.

Translating \widetilde{B} into $\widetilde{CO}_{2, BOL}$ and \widetilde{Cost}_{BOL} it can be reviewed how the BOL results for the SOC 100%, SOC 80% and mission-based strategy compare. The $\widetilde{CO}_{2, BOL}$ and \widetilde{Cost}_{BOL} results for the strategies are depicted in Figure 2.

7.1.2 Battery Usage

During the operation, the batteries' charging CO_2 emissions and costs and flying efficiency are assessed. The mission-based battery health management strategy is anticipated to perform better than the SOC 100% strategy for all three parameters. Below, the amount of energy required to charge the batteries per flying distance flown (\tilde{P}_{total}) and efficiency of the flying time to the total time are assessed.

To evaluate the model outputs for the *mixed* range, the following sets of hypotheses listed in Table 8 to 11 are defined for \tilde{P}_{total} and *Efficiency* per battery. From supporting work [86] a statistical analysis shows that both outputs follow a normal distribution for the SOC 100%, SOC 80% and mission-based models. The hypotheses listed below are tested for $\alpha = 1\%$ with an unpaired one-sided T-test.

Mission-based vs. SOC 100% - mixed range
$H_0: \widetilde{P}_{total, mission-based} \geq \widetilde{P}_{total, SOC 100\%}$
$H_1: \widetilde{P}_{total, mission-based} < \widetilde{P}_{total, SOC 100\%}$

Table 8: **Hypothesis** \widetilde{P}_{total} - Mission-based vs. SOC 100% strategy for mixed range

Mission-based vs. SOC 100% - $\mathbf{mixed\ range}$
$H_0:$ Efficiency mission-based \leq Efficiency SOC 100%
H_1 : Efficiency mission-based > Efficiency SOC 100%

Table 9: **Hypothesis** *Efficiency* - Mission-based vs. SOC 100% strategy for mixed range

From the relative outcome of the MC simulations presented in Table 5, it can be concluded that the mission-

based battery health management strategy results in a lower \tilde{P}_{total} and higher flying efficiency. The missionbased \tilde{P}_{total} is reduced with **6.4%** of compared to the SOC 100% approach, while the efficiency increases by **2.8%**. The improved battery usage performance is the result of the fact that the battery is merely charged to the estimated amount of charge. Both the samples for \tilde{P}_{total} and *Efficiency* differ with 100% confidence as presented in supporting work [86], implying that both H_0 hypotheses in Table 8 and Table 9 are rejected.

Mission-based vs. SOC 80% - $\mathbf{mixed\ range}$
$H_0: \widetilde{P}_{total, mission-based} \ge \widetilde{P}_{total, SOC 80\%}$
$H_1: \widetilde{P}_{total, \ mission-based} < \widetilde{P}_{total, \ SOC \ 80\%}$

Table 10: **Hypothesis** \widetilde{P}_{total} - Mission-based vs. SOC 80% strategy for mixed range

Mission-based vs. SOC 80% - $\mathbf{mixed\ range}$
$H_0: Efficiency_{mission-based} \leq Efficiency_{SOC 80\%}$
H_1 : Efficiency mission-based > Efficiency SOC 80%

Table 11: **Hypothesis** *Efficiency* - Mission-based vs. SOC 80% strategy for mixed range

The average results of the MC simulations presented in Table 5 show the mission-based battery health management strategy results in a lower total power used for charge per mission time flown (\tilde{P}_{total}). The \tilde{P}_{total} for the mission-based approach is reduced with **5.2%** compared to the SOC 80% strategy. However, despite this improved charging performance, the mission-based outputs a **6.9%** lower efficiency with respect to the SOC 80% strategy. This is explained by the fact that the strategy skips charging when $\text{SOC}_{req} \leq 80\%$ and $\text{SOC}(t) \geq 80\%$, which is frequently yielded after having charged to SOC 100% for the previous flight. Therefore, the overall conclusion stands that the H_0 for \tilde{P}_{total} in Table 10 is rejected with 100% confidence [86], while the H_0 for Efficiency in Table 11 is failed to be rejected.

For the sustainability and cost assessment of each battery, the sum of charging power per time flown (\tilde{P}_{total}) is translated in to $\tilde{CO}_{2, \ charge}$ and \tilde{Cost}_{charge} . Moreover, the flying efficiency is directly assessed. The battery usage results per battery run through the MC simulations for the SOC 100%, SOC 80% and missionbased strategy are displayed in Figure 3.

7.2 Short, Medium and Long Range

Reviewing different distance ranges flown, the missionbased battery strategy is expected to perform better than the SOC 100% and SOC 80% model for both 'battery lifetime' and 'battery usage' related parameters. Moreover, the short range distance is expected to yield the best results due to minimum battery degradation influenced by DOD.

In a statistical analysis presented in supporting work [86], it is concluded that the results for the SOC 100%,

	Short		Medium		Long	
Parameter	Mission-based vs. SOC 100%	Mission-based vs. SOC 80%	Mission-based vs. SOC 100%	Mission-based vs. SOC 80%	Mission-based vs. SOC 100%	Mission-based vs. SOC 80%
Ĩ	-34.3%	-15.1%	-19.9%	+5.5%	-10.9%	-10.8%
\widetilde{P}_{total}	-9.7%	-0.8%	-7.4%	-6.1%	-4.5%	-4.5%
Efficiency	+6.8%	+2.6%	+1.5%	-15.6%	+0.6%	+0.5%

Table 12: Relative mean Battery Usage and Battery Lifetime performance of 300 batteries tested through Monte Carlo simulation comparing the SOC 100%, SOC 80% and mission-based strategy per short, medium and long range



Figure 4: Sustainability and Cost Assessment - Battery Lifetime results of 300 batteries tested through Monte Carlo simulation for the SOC 100%, SOC 80% and mission-based strategy for short, medium and long range



Figure 5: Sustainability and Cost Assessment - Battery Usage results of 300 batteries tested through Monte Carlo simulation the for SOC 100%, SOC 80% and mission-based strategy for short, medium and long range
SOC 80% and mission-based strategy are normally distributed and can thus be evaluated with an unpaired one-sided T-test. The hypotheses for the short, medium and long range results are tested for $\alpha = 1\%$. Here, *Output* refers to the performance of \tilde{B} , \tilde{P}_{total} and *Efficiency*. An improved performance corresponds to a lower \tilde{B} and \tilde{P}_{total} , and higher *Efficiency* output.

The results of the mission-based strategy compared to the SOC 100% approach are first discussed in section 7.2.1. Subsequently, the results of the missionbased compared to the SOC 100% strategy are evaluated in section 7.2.2. Finally, the results of the different ranges are compared in section 7.2.3.

7.2.1 Mission-based vs. SOC 100%

To review if the mission-based strategy is better than the SOC 100% approach, hypotheses for the *short*, *medium* and *long* range are presented in Table 13, 14 and 15, respectively.

Mission-based vs. SOC 100% - short range
$H_0: Output_{mission-based} \leq Output_{SOC 100\%}$
$H_1: Output mission-based > Output SOC 100\%$

Table 13: Hypothesis Output - Mission-based vs.SOC 100% strategy for short range

Mission-based vs. SOC 100% - medium range
$H_0: Output mission-based \leq Output SOC 100\%$
$H_1: Output \ mission-based > Output \ SOC \ 100\%$

Table 14: Hypothesis Output - Mission-based vs.SOC 100% strategy for medium range

Mission-based vs. SOC 100% - \log range
$H_0: Output _{mission-based} \leq Output _{SOC 100\%}$
H_1 : Output mission-based > Output SOC 100%

Table 15: Hypothesis Output - Mission-based vs.SOC 100% strategy for long range

From the relative mean battery lifetime and usage results presented in Table 12, Figure 4 and 5 the following conclusions are drawn. The mission-based approach yielded better results than the SOC 100% for all three distance ranges.

Short range

Battery lifetime for the mission-based strategy is improved significantly as it shows the lowest \tilde{B} . For battery usage, the P_{total} is reduced with and efficiency is increased. Compared to the *medium* and *long* range, it becomes apparent that the battery lifetime and usage is most affected in the *short* range. This is the result of the shorter distances flown that result in a longer lifetime and minimal charging activity corresponding to a lower \tilde{P}_{total} and higher efficiency. Concluding these results and a T-test presented in supporting work [86], the H_0 for the **short** range in Table 13 can be rejected with 100% confidence.

Medium range

With the mission-based strategy, the battery lifetime and usage yield improved results compared to the SOC 100% strategy. Here, similar values to the performance in *mixed* distance are observed. Based on the output of these results and a T-test [86], the H_0 for the **medium** range in Table 14 is rejected with 100% confidence.

Long range

The mission-based strategy yields less \tilde{B} than the SOC 100% approach. Furthermore, the \tilde{P}_{total} is improved and an efficiency that is also a little higher. Contrarily to the other distance ranges, though, the difference in performance of the mission-based and SOC 100% strategy for the *long* distance is less evident. This is due to the fact that when flying to targets that are positioned further away, the mission-based strategy is required to charge to a SOC level that is closer to 100%. From these results and the T-test output [86], H_0 for **long** range in Table 15 can successfully be rejected with 100% confidence.

7.2.2 Mission-based vs. SOC 80%

The hypotheses to evaluate the performance of the mission-based strategy compared to the SOC 80% approach for the *short, medium* and *long* range are presented in Table 16, 17 and 18, respectively.

Mission-based vs. SOC 80% - short range
$H_0: Output mission-based \leq Output SOC 80\%$
H_1 : Output mission-based > Output SOC 80%

Table 16: **Hypothesis** *Output* - Mission-based vs. SOC 80% strategy for short range

Miss	sion-based vs. SOC 80% - medium range
H_0 :	$Output mission-based \leq Output SOC 80\%$
H_1 :	Output mission-based > Output SOC 80%

Table 17: Hypothesis Output - Mission-based vs.SOC 80% strategy for medium range

Mission-based vs. SOC 80% - \log range
$H_0: Output _{mission-based} \leq Output _{SOC 80\%}$
$H_1: Output mission-based > Output SOC 80\%$

Table 18: **Hypothesis** *Output* - Mission-based vs. SOC 80% strategy for long range

Comparing the relative mean battery lifetime and usage results presented in Table 12, Figure 4 and 5 the following conclusions are drawn for the mission-based and SOC 80% strategy.

Short range

The mission-based strategy shows a better \tilde{B} than the SOC 80% strategy. For battery usage, the \tilde{P}_{total} is only slightly reduced, which suggest that both strategies charge to similar SOC levels. The efficiency of the mission-based strategy is however still higher because the average required SOC for short ranges is very low, resulting in the battery regularly skipping charging before a flight. From these results and a T-test elaborated on in supporting work [86], H_0 for the **short** range in Table 16 is rejected with 100% confidence.

Medium range

For the medium range, the mission-based strategy is not superior to the SOC 80% strategy, which goes against the hypothesis formulated above. The results show that the SOC 80% strategy yields a better B and efficiency. An explanation for this could be the fact that the eUAV battery regularly skips charging due to the toggle between SOC 100% and 80%. As a result, battery degradation is less severe and efficiency is significantly increased. When analysing the charging power P_{total} , however, the mission-based strategy outperforms the SOC 80% approach. To conclude, the H_0 for the **medium** in Table 17 is only partially rejected, as only the mission-based strategy only outperformed SOC 80% for P_{total} during operation. The confidence levels for the T-tests are presented in supporting work [86].

Long range

The mission-based strategy results in a lower \tilde{B} compared to the SOC 100% approach. Also, the \tilde{P}_{total} is and efficiency are improved. Based on the T-test results [86], H_0 for the **long** range in Table 18 is rejected with 100% confidence. The results show that for *long* distance, the SOC 80% strategy performs very similar to the SOC 100% approach. This is due to the fact that 80% SOC is insufficient to safely complete a mission to the target in the *long* range, forcing the battery to almost always charge to 100%.

7.2.3 Distances

Per strategy, the performance of *short* compared to *medium* and *long* distance is evaluated to review the influence of DOD. The hypotheses for testing the impact of the distance are presented in Table 20. It is expected that the *short* distance outperforms the *medium* and *long* range model due to battery lifetime elongation as a result of a smaller DOD.

Distance - short vs. medium and long range
$H_0: Output short \leq Output medium, long$
$H_1: Output \ _{short} > Output \ _{medium, \ long}$

Table 20:HypothesisOutput-Distance short vs.medium and long range

Reviewing the battery lifetime and usage sustainability and cost assessment results in Table 19, Figure 4 and 5, several conclusions are drawn.

Battery lifetime

When comparing the battery lifetime output within each strategy, it is observed that the *short* range yields the best battery lifetime performance compared to *medium* and *long* target distance range. This is in line with the expectations and shows that a smaller DOD results in a longer battery life. Furthermore, it becomes apparent that decreasing the DOD elongates the lifetime most significantly for mission-based strategy compared to the SOC 100% and SOC 80% strategy because the batteries with the mission-based approach are cycled around a lower average SOC.

Battery usage

For battery usage, the results show that P_{total} yields the lowest results for the *short* distance, which supports the hypothesis stated above. The amount of charge used during battery lifetime decreases with approximately the same order of magnitude for each strategy. The reason that the charge required for the total mission time flown decreases, originates from the fact that the battery suffers from less severe battery ageing, implying that a eUAV battery is able to fly longer. For the flying efficiency for the SOC 80% strategy, however, the value is highest for the *medium* range, instead of the *short* range. The high efficiency for the SOC 80% range is attributed to the SOC 100% and 80% toggle mentioned previously.

Concluding the findings above, the H_0 in Table 20 for **distance** is partially rejected, because the flying efficiency with the SOC 80% strategy for *medium* distance outperforms the other ranges. The T-test results are presented in supporting work [86].

7.3 Discussion

A concluding discussion of the results of this study is given in the following sections. The benefit of applying a mission-based battery health management strategy by minimising the average SOC and DOD is first elaborated on in section 7.3.1 and 7.3.2, respectively. Then, general sustainability and cost assessment findings are discussed in section 7.3.3. Throughout the section, several recommendations for future studies are provided. A more elaborate overview of this study's limitations and suggestions for further analysis is given in supporting work [86].

7.3.1 Average State of Charge

From the *mixed* range simulations, it is concluded that the mission-based study which cycles the battery around a lower average SOC by charging to a beforehand estimated required SOC, yields better results. However, it must be noted that it cannot be ruled out that this performance is solely the result of cycling

	SOC	100%	SOC	80%	Mission-based	
Parameter	Short vs.	Short vs.				
	Medium	Long	Medium	Long	Medium	Long
\widetilde{B}	-7.0%	-8.8%	-5.2%	-29.4%	-23.7%	-32.8%
\widetilde{P}_{total}	-32.3%	-45.0%	-37.5%	-50.0%	-34.0%	-48.0%
Efficiency	+5.3%	+9.1%	-8.9%	+13.5%	+10.7%	+15.8%

Table 19: Relative mean Battery Usage and Battery Lifetime performance of 300 batteries tested through Monte Carlo simulation comparing short, medium and long range per strategy

around a lower SOC, as the influence of skipping charging must not be neglected. By applying the missionbased strategy, benefits are yielded across eUAV *battery lifetime* parameters with respect to emissions and costs.

Secondly, during eUAV operation for the *mixed* distance, the mission-based methodology yields the least power required for charging. This is due to the fact that the battery is merely charged to the estimated SOC required to complete the next flight, or is immediately used to fly if the battery already has sufficient charge. Regarding flying efficiency, however, the SOC 80% strategy performs best. This is the result of the SOC 80% strategy's toggle between SOC 80% for shorter and SOC 100% for longer distances. This leads to the eUAV regularly having sufficient SOC to complete the next mission without being charged beforehand. It could therefore be argued that the need for a mission-based model is challenged, considering the fact that the SOC 80% is relatively simple to apply.

Although the SOC 80% and mission-based strategy improve battery lifetime and usage performance, it could be argued that implementing these battery health management strategies is risk-prone due to the reduced available capacity for flight. For future studies it is therefore recommended to include a safety assessment parameter. Moreover, the reliability of the strategies could be increased by establishing a more advanced SOC estimation model that accounts for an elaborate set of factors such as wind, altitude, payload and flight speed.

7.3.2 Depth of Discharge

By varying the distance flown by the eUAV for a short, medium and long range, the effect of varying the DOD of that the battery cycles through is explored. From the model output, it is concluded that the mission-based battery health management strategy outperforms the SOC 100% strategy for all three distance ranges. The *battery lifetime* is most significantly affected for the *short* range, as this distance pairs with a smaller DOD, a lower average SOC and regular skipping of charging practices. During *battery usage*, the power for charging is reduced for the mission-based strategy as the battery suffers from less severe battery ageing, implying that a eUAV battery is able to fly longer. Lastly, the flying efficiency is increased due to the fact that the battery is either charged to a lower SOC, or can immediately be inputted to fly given it

already has sufficient charge from the previous flight.

Comparing the mission-based strategy to the SOC 80% approach, it is concluded that the SOC 80% yields benefits for the *medium* range. For this distance, the SOC 80% strategy resulted in a better *battery lifetime* yielding lower emissions and costs. Moreover, the flying efficiency was significantly increased for *battery usage*. This improved performance is the result of the SOC 80% regularly skipping the charging step when the eUAV has sufficient charge from the previous flight.

Given the difference in sustainability and cost assessment results for the *mixed*, *short*, *medium* and *long* range, a recommendation for future research is to review the optimal flying distance to maximise battery lifetime and usage performance.

Note that if the environmental and financial assessment results would be specified over target distance flown instead of total flying time, a different output is yielded. For the short range, this implies that a eUAV travels less cruise distance per flight, but still has to perform the same take-off and landing manoeuvres requiring relatively more charge per target distance flown. As a result, the results for the short range would not outperform the other ranges for battery lifetime and battery usage.

7.3.3 Sustainability and Cost Assessment

Comparing the battery lifetime to the battery usage results, it becomes apparent that the environmental and financial costs at BOL have the largest impact. For the mixed range, the $\widetilde{CO}_{2, BOL}$ emissions for approximately 91.1%, 89.8% and 89.8% of the total emissions for the SOC 100%, SOC 80% and mission-based battery health management strategy, respectively. For the economic costs, the \widetilde{Cost}_{BOL} approximately sum up to equal 99.9% of the total costs for all three strategies. This is in line with the findings described in literature, and suggest that battery lifetime could be considered prime focus point by operators.

The second discussion point revolves around the seemingly low battery usage efficiency rate across all strategies and distances flown. This output is attributed to the CC/CV charging protocol that is inputted into the model. Due to the applied charging C-rate and method, the charging time is approximately twice as long as the discharging practices, yielding a low flying efficiency rate. Applying a different charging methodology would influence the magnitude of this parameter, which must be taken into account when us-

ing the same framework to other battery models.

Finally, when analysing the distributions of the battery lifetime and usage graphs depicting the charging emissions, costs and flying efficiency, the following points stand out. For the emissions and costs boxplots, similar distributions for the SOC 100%, SOC 80% and mission-based strategy are observed. For flying efficiency, however, the SOC 80% and mission-based strategy show a wider spread of flying efficiency results compared to the SOC 100% approach. This is caused to the SOC 80% and mission-based strategy set-up in which the battery is not charged before flight if it has sufficient SOC from the previous mission. This becomes strongly apparent for short range flights which require less charge to safely complete a mission.

8 Validation

Validation of the battery model is done through a Sensitivity Analysis (SA). Due to the absence of real data to compare the model output, a SA is one of the few methods to validate functional frameworks [10, 39, 67]. By running a SA, the robustness of a model is checked. For future studies, it is recommended to validate the model with real eUAV battery data.

The result of a change in inputs is reviewed both with respect to *logic* and *sensitivity*. A One-at-a-Time (OAT) analysis [10] is performed to review how the output responds to a change of a single input while keeping the other inputs constant. The inputs that are varied are those that are stochastically generated for each battery and flight, listed in Table 22 in Appendix A. Each input is tested OAT for their minimum and maximum value. Depending on their distribution, this is either equal to the minimum or maximum value of the uniform distribution, or $\pm 95\%$ confidence interval value for normally distributed variables.

In the interest of speeding up the computational time, the SOC 100% methodology is used to validate the battery model. Moreover, the MC is run for 300 runs for each sensitivity parameter as the mean target distance covered by each eUAV battery stabilised for this number of simulations. A verification of this is given in statistical analysis in supporting work [86].

In Figure 6 and 7, the tornado plots of the SA results for battery lifetime and usage are depicted. The mean results of the model outputs for the minimum or maximum sensitivity parameters varied OAT are compared to the average results for a model in which all sensitivity parameter inputs are fixed at their mean. These validation outputs for battery lifetime are first discussed in section 8.1, followed by the battery usage results in section 8.2.

8.1 Battery Lifetime

For battery lifetime, Figure 6 shows that the minimum and maximum settings for SOH have the biggest influence on the batteries used per flying distance flown (\tilde{B}) . This tells us that with respect to lifetime, the model is



Figure 6: Sensitivity Analysis for Battery Lifetime - relative change in *output* of \widetilde{B} (battery used per flying time) for SOC 100% strategy

more sensitive to the BMS error than the other stochastic inputs. The relation between the SOH sensitivity parameter and lifetime is in line with expectations. If the BMS consequently reads a lower SOH than the actual value, the battery is terminated at an earlier stage. Contrarily, if a higher SOH is gone by, the battery is terminated later than the real SOH 80% threshold, resulting in an elongated lifetime. For \tilde{B} , the minimum SOH input value results in a larger change in output deviation than the maximum value. This is because the flying time sums up to equal a relatively larger fraction for a shorter compared to at a longer lifetime which is governed by the CC/CV charging protocol.

For the remaining sensitivity parameters, the battery lifetime results display logical results. The battery performs more flights per battery when for minimum current throughput and flying time manoeuvre duration, yielding a lower \tilde{B} . This is due to a decrease in ageing severeness as a result of lower current rates during discharge and smaller DOD. Moreover, lower SEI kinetic rate values result in longer battery life. On the contrary, the eUAV battery is able to complete more flights on average when the maximum initial capacity is inputted into the model. Finally, the initial voltage parameter shows a negligible difference when comparing the minimum and maximum inputted values to the mean.

8.2 Battery Usage

The \dot{P}_{total} and flying efficiency SA tornado plots are shown in Figure 7a and 7b, respectively. The average output of results varies marginally for battery usage. The influence of each sensitivity parameter is discussed below.

The flying current and time duration sensitivity values have a large influence on the resulting \tilde{P}_{total} and flying efficiency. This is logical, as battery ageing is expected to be less severe for lower current rates and DOD levels, resulting in a longer lifetime which pairs with smaller \tilde{P}_{total} values. For flying efficiency, minimum current and time values yield a higher efficiency. This is attributed to the fact that the batteries require



Figure 7: Sensitivity Analysis for Battery Usage - relative change in *output* for SOC 100% strategy

less charging time on average, which results in a higher flying efficiency because charging takes relatively long due to a lower current rate during the CC phase.

The maximum SOH reading error implies that batteries are operated longer which yields a higher total flying time per battery. Hence, the \tilde{P}_{total} for each battery decreases. However, the flying efficiency is also decreased, as more flights are flown which pair with relatively longer charging time. Furthermore, from the tornado plots it is concluded that the \tilde{P}_{total} and flying efficiency is hardly related to the SEI kinetic rate.

The initial capacity of the eUAV battery has a relatively large influence on the flying efficiency output. The reason for this is not trivial and requires further analysis. For \tilde{P}_{total} , a higher initial capacity results in more charging power required to reach a fully charged state. This is linked to the longer lifetime presented in the SA plot for \tilde{B} . Finally, the initial voltage level has a negligible influence on the SA results.

9 Conclusions and Recommendations

This research reviews the environmental and costbenefit of applying a mission-based battery health management strategy to Lithium batteries used in an electric Unmanned Aerial Vehicle (eUAV). By applying the mission-based strategy, the eUAV battery is charged to the estimated State of Charge (SOC) level required to successfully complete a flight. The performance of the mission-based strategy is compared to two other approaches where the battery is either charged to 100% SOC or SOC 80%. The strategies are tested for several different flight ranges (*mixed, short, medium* and *long*) to explore the influence of varying the Depthof-Discharge (DOD). The motivation to research this topic originated from the hypothesis that cycling a battery at a lower average level of SOC and smaller DOD ranges results in a longer battery lifetime, which is especially important from a sustainability perspective.

The research is carried out with a Nickel, Manganese and Cobalt (NMC) Lithium battery model simulated with the Python Battery Mathematical Modelling (Py-BaMM) package. A Monte Carlo (MC) simulation is used to review how the model responds to uncertainties in initial battery characteristics and in-flight variations. The eUAV battery performs flights including a take-off, cruise and landing manoeuvre, during which it flies there and back to stochastically generated targets. Ultimately, the performance of the mission-based strategy is compared to the SOC 100% and SOC 80% approach by executing a sustainability and cost assessment. Here, the environmental and financial costs related to the batteries' lifetime and usage are assessed. Furthermore the efficiency of flying is evaluated.

Comparing the mission-based strategy to the SOC 100% and SOC 80% model for the *mixed* distance, a decrease in emissions and costs is observed. Regarding flying efficiency that aims to minimise downtime, however, the SOC 80% performs best. Additionally, like the mission-based strategy, the SOC 80% approach outperforms the SOC 100% strategy for all assessment parameters. These results could challenge the significance of implementing a mission-based battery health management strategy, as the SOC 80% method is relatively simple to establish. Reviewing the results of the *short, medium* and *long* range flights, it is concluded that battery lifetime and usage results are significantly improved for a battery flying short range flights.

Although a more advanced model is advised before applying this model to real-life eUAV applications, this study provides insights with respect to how a missionbased battery health management strategy could yield environmental, financial and flying efficiency benefits. Addressing the shortcomings described in section 7 and 8, the foremost focus point for future recommendations of the model include the incorporation of a safety assessment parameter. Moreover, it is advised to extend the model by including inputs such as wind, speed, altitude and payload settings. Finally, it is recommended to validate the model with real-life experiments. Here, it is especially advised to review the Battery Management System's (BMS) accuracy of determining the battery's SOH, as this sensitivity parameter resulted in the most significant relative output changes.

To conclude, this study analysed the benefit of applying a mission-based battery health management strategy with the use of a simulated Lithium battery model in PyBaMM. The mission-based methodology, battery model and sustainability and cost assessment framework can support operators in evaluating and comparing the performance of different battery health management strategies.

References

- National Aeronautics and Space Administration. Prognostics center of excellence datasets. https://ti.arc.nasa.gov/tech/dash/groups/ pcoe/prognostic-data-repository. Retrieved: 16-04-2021.
- [2] Jean-Philippe Aurambout, Konstantinos Gkoumas, and Biagio Ciuffo. Last mile delivery by drones: an estimation of viable market potential and access to citizens across european cities. *European Transport Research Review*, 11, 12 2019. doi: 10.1186/s12544-019-0368-2.
- [3] Reg Austin. Unmanned Aircraft Systems. John Wiley and Sons Ltd., 1st edition, 2010. ISBN 9780470058190.
- [4] Anthony Barré, Benjamin Deguilhem, Sébastien Grolleau, Mathias Gérard, Frédéric Suard, and Delphine Riu. A review on lithium-ion battery ageing mechanisms and estimations for automotive applications. *Journal of Power Sources*, 241: 680–689, nov 2013. ISSN 03787753. doi: 10.1016/ j.jpowsour.2013.05.040.
- [5] Rushikesh Battulwar, Garrett Winkelmaier, Jorge Valencia, Masoud Zare Naghadehi, Bijan Peik, Behrooz Abbasi, Bahram Parvin, and Javad Sattarvand. A practical methodology for generating high-resolution 3d models of open-pit slopes using uavs: Flight path planning and optimization. *Remote Sensing*, 12(14), 2020. ISSN 2072-4292. doi: 10.3390/rs12142283.
- [6] Thorsten Baumhöfer, Manuel Brühl, Susanne Rothgang, and Dirk Uwe Sauer. Production caused variation in capacity aging trend and correlation to initial cell performance. *Journal of Power Sources*, 247:332–338, 2014. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2013.08.108.
- [7] Christoph Birkl, Matthew Roberts, Euan Mc-Turk, Peter Bruce, and David Howey. Degradation diagnostics for lithium ion cells. *Journal of Power Sources*, 341:373–386, feb 2017. ISSN 03787753. doi: 10.1016/j.jpowsour.2016.12.011.
- [8] Robert Cameron Bolam, Yuriy Vagapov, and Alecksey Anuchin. A review of electrical motor topologies for aircraft propulsion. In 2020 55th International Universities Power Engineering Conference (UPEC), pages 1–6, 2020. doi: 10.1109/UPEC49904.2020.9209783.
- [9] Brian Bole, Chetan S Kulkarni, and Matthew Daigle. Adaptation of an Electrochemistry-based

Li-Ion Battery Model to Account for Deterioration Observed Under Randomized Use. In Annual Conference of the Prognostics and Health Management Society 2014, pages 502–510, 2014.

- [10] Emanuele Borgonovo and Elmar Plischke. Sensitivity analysis: a review of recent advances. *European Journal of Operational Research*, 248(3): 869–887, 2016.
- [11] Alexander Busse, Joachim Metternich, and Eberhard Abele. Evaluating the Benefits of Predictive Maintenance in Production: A Holistic Approach for Cost-Benefit-Analysis. In Advances in Production Research, pages 690–704. Springer International Publishing, nov 2019. doi: 10.1007/978-3-030-03451-1_67.
- [12] Isidor Buchmann Cadex Electronics Inc. Battery university. https://batteryuniversity.com/. Retrieved: 19-04-2021.
- [13] Centraal Bureau van Statistieken (CBS). 11 procent energieverbruik in 2020 afkomstig uit hernieuwbare bronnen. https: //www.cbs.nl/nl-nl/nieuws/2021/22/11procent-energieverbruik-in-2020afkomstig-uit-hernieuwbare-bronnen. Retrieved: 26-11-2021.
- [14] Wen-Yeau Chang. The State of Charge Estimating Methods for Battery: A Review. International Scholarly Research Notices Applied Mathematics, 2013:1–7, jul 2013. doi: 10.1155/2013/953792.
- [15] Climate Neutral Group. Review emissiefactoren 2021. https://www.climateneutralgroup.com/ wp-content/uploads/2021/05/210422-CNG-Review-emissiefactoren-2020-2021-NL.pdf, Nederland. Retrieved: 26-11-2021.
- [16] Luis Corral, Ilenia Fronza, Nabil El Ioini, and Aristea Ibershimi. A measurement tool to track drones battery consumption during flights. In *International Conference on Mobile Web and Information Systems*, 08 2016. ISBN 978-3-319-44214-3. doi: 10.1007/978-3-319-44215-0 28.
- [17] Quach Cuong Chi, Brian Bole, Edward Hogge, Sixto Vazquez, Matthew Daigle, José Celaya, Adam Weber, and Kai Goebel. Battery Charge Depletion Prediction on an Electric Aircraft. In Annual Conference of the Prognostics and Health Management Society 2013, pages 1–11, 2013.
- [18] Leszek Cwojdziński and Mirosław Adamski. Power units and power supply systems in UAV. Aviation, 18(1):1–8, jan 2014. ISSN 16487788. doi: 10.3846/16487788.2014.865938.
- [19] Dominik Franjo Dominković, Ivan Bačeković, Allan Schrøder Pedersen, and Goran Krajačić. The future of transportation in sustainable energy systems: Opportunities and barriers in a clean energy

transition. *Renewable and Sustainable Energy Reviews*, 82:1823–1838, 2018. ISSN 1364-0321. doi: 10.1016/j.rser.2017.06.117.

- [20] Gonçalo Dos Reis, Calum Strange, Mohit Yadav, and Shawn Li. Lithium-ion battery data and where to find it. *Energy and AI*, 5:100081, 2021. ISSN 2666-5468. doi: 10.1016/j.egyai.2021. 100081.
- [21] Droneshop.nl. Dji matrice 100 tb47d battery. https://droneshop.nl/dji-matrice-100-tb47d-intelligent-flight-battery. Retrieved: 18-01-2022.
- [22] Matthieu Dubarry, Cyril Truchot, Mikaël Cugnet, Bor Yann Liaw, Kevin Gering, Sergiy Sazhin, David Jamison, and Christopher Michelbacher. Evaluation of commercial lithium-ion cells based on composite positive electrode for plug-in hybrid electric vehicle applications. part i: Initial characterizations. Journal of Power Sources, 196 (23):10328–10335, 2011. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2011.08.077.
- [23] Nick Eleftheroglou, Sina Sharif Mansouri, Theodoros Loutas, Petros Karvelis, George Georgoulas, George Nikolakopoulos, and Dimitrios Zarouchas. Intelligent data-driven prognostic methodologies for the real-time remaining useful life until the end-of-discharge estimation of the Lithium-Polymer batteries of unmanned aerial vehicles with uncertainty quantification. Applied Energy, 254:113677, nov 2019. ISSN 03062619. doi: 10.1016/j.apenergy.2019.113677.
- [24] Eurostat. Electricity price statistics. https://ec.europa.eu/eurostat/statisticsexplained/. Retrieved: 26-11-2021.
- [25] Alexander Farmann, Wladislaw Waag, Andrea Marongiu, and Dirk Uwe Sauer. Critical review of on-board capacity estimation techniques for lithium-ion batteries in electric and hybrid electric vehicles. *Journal of Power Sources*, 281:114–130, 2015. ISSN 03787753. doi: 10.1016/j.jpowsour. 2015.01.129.
- [26] Kiri Feldman, Taoufik Jazouli, Peter Sandborn, and Senior Member. A Methodology for Determining the Return on Investment Associated with Prognostics and Health Management. *IEEE Transactions on Reliability*, 58(2):305–316, 2009.
- [27] Center for Advanced Life Cycle Engineering. Battery data. https://calce.umd.edu/batterydata. Retrieved: 25-11-2021.
- [28] Chiara Franciosi, Benoit Iung, Salvatore Miranda, and Stefano Riemma. Maintenance for Sustainability in the Industry 4.0 context: a Scoping Literature Review. *IFAC-PapersOnLine*, 51(11):903– 908, jan 2018. ISSN 24058963. doi: 10.1016/j. ifacol.2018.08.459.

- [29] Boris Galkin, Jacek Kibilda, and Luiz A. DaSilva. Uavs as mobile infrastructure: Addressing battery lifetime. *IEEE Communications Magazine*, 57(6):132–137, 2019. doi: 10.1109/MCOM.2019. 1800545.
- [30] Yang Gao, Jiuchun Jiang, Caiping Zhang, Weige Zhang, and Yan Jiang. Aging mechanisms under different state-of-charge ranges and the multiindicators system of state-of-health for lithiumion battery with Li(NiMnCo)O2 cathode. *Journal* of Power Sources, 400:641–651, oct 2018. ISSN 03787753. doi: 10.1016/j.jpowsour.2018.07.018.
- [31] Farnaz Ghazi Nezami and Mehmet Yildirim. A sustainability approach for selecting maintenance strategy. *International Journal of Sustainable En*gineering ISSN:, 6(4):332–343, 2013. ISSN 1939-7046. doi: 10.1080/19397038.2013.765928.
- [32] Anne Goodchild and Jordan Toy. Delivery by drone: An evaluation of unmanned aerial vehicle technology in reducing co2 emissions in the delivery service industry. *Transportation Research Part* D: Transport and Environment, 61:58–67, 2018. ISSN 1361-9209. doi: 10.1016/j.trd.2017.02.017. Innovative Approaches to Improve the Environmental Performance of Supply Chains and Freight Transportation Systems.
- [33] Clare Grey and Jean-Marie Tarascon. Sustainability and in situ monitoring in battery development, dec 2017. ISSN 14764660.
- [34] Jeroen Guinée. Life cycle assessment: an operational guide to the iso standards; lca in perspective; guide; operational annex to guide. Centre for Environmental Science, Leiden University, The Netherlands, 2001.
- [35] Han Hao, Zhexuan Mu, Shuhua Jiang, Zongwei Liu, and Fuquan Zhao. Ghg emissions from the production of lithium-ion batteries for electric vehicles in china. *Sustainability*, 9(4), 2017. ISSN 2071-1050. doi: 10.3390/su9040504.
- [36] Khaled Hatamleh, Ou Ma, and Robert Paz. A uav model parameter identification method. *International Journal of Information Acquisition*, 06, 08 2010. doi: 10.2514/6.2010-7649.
- [37] Edward Hogge, Chetan S Kulkarni, Sixto Vazquez, Kyle Smalling, Thomas Strom, Boyd Hill, and Cuong Quach. Flight Tests of a Remaining Flying Time Prediction System for Small Electric Aircraft in the Presence of Faults. In Conference of the Prognostics and Health Management Society, pages 1–12, 2017.
- [38] Bin Huang, Zhefei Pan, Xiangyu Su, and Liang An. Recycling of lithium-ion batteries: Recent advances and perspectives. *Journal of Power Sources*, 399:274–286, 2018. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2018.07.116.

- [39] Bertrand Iooss and Andrea Saltelli. Introduction to sensitivity analysis. *Handbook of uncertainty quantification*, pages 1103–1122, 2017.
- [40] Benoit Iung and Eric Levrat. Advanced maintenance services for promoting sustainability. Procedia CIRP, 22:15–22, 2014. ISSN 2212-8271. doi: 10.1016/j.procir.2014.07.018. Proceedings of the 3rd International Conference in Through-life Engineering Services.
- [41] Peter Keil, Simon Schuster, Jörn Wilhelm, Julian Travi, Andreas Hauser, Ralph C. Karl, and Andreas Jossen. Calendar Aging of Lithium-Ion Batteries. Journal of The Electrochemical Society, 163 (9):A1872–A1880, jul 2016. ISSN 0013-4651. doi: 10.1149/2.0411609jes.
- [42] Renee Kent and Dennis Murphy. Health Monitoring System Technology Assessments - Cost Benefits Analysis. Technical report, ARINC - Aeronautical Radio, Incorporated, 2000.
- [43] Jiwon Kim, Yonghun Choi, Seunghyeok Jeon, Jaeyun Kang, and Hojung Cha. Optrone: Maximizing performance and energy resources of drone batteries. *IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems*, 39 (11):3931–3943, 2020. doi: 10.1109/TCAD.2020. 3012790.
- [44] Jarotwan Koiwanit. Analysis of environmental impacts of drone delivery on an online shopping system. Advances in Climate Change Research, 9(3): 201–207, 2018. ISSN 1674-9278. doi: 10.1016/j. accre.2018.09.001.
- [45] Sandia National Laboratories. R&d data repository. https://www.sandia.gov/ energystoragesafety-ssl/researchdevelopment/research-data-repository/. Retrieved: 25-11-2021.
- [46] Laura Lander, Tom Cleaver, Mohammad Ali Rajaeifar, Viet Nguyen-Tien, Robert J.R. Elliott, Oliver Heidrich, Emma Kendrick, Jacqueline Sophie Edge, and Gregory Offer. Financial viability of electric vehicle lithium-ion battery recycling. *iScience*, 24(7):102787, 2021. ISSN 2589-0042. doi: 10.1016/j.isci.2021.102787.
- [47] Jie Li, Kasim Adewuyi, Nima Lotfi, Robert Landers, and Jonghyun Park. A single particle model with chemical/mechanical degradation physics for lithium ion battery state of health (soh) estimation. Applied Energy, 212:1178–1190, 2018. ISSN 0306-2619. doi: 10.1016/j.apenergy.2018.01.011.
- [48] Xin Li, Konstantinos J. Chalvatzis, and Phedeas Stephanides. Innovative energy islands: Life-cycle cost-benefit analysis for battery energy storage. *Sustainability*, 10(10), 2018. ISSN 2071-1050. doi: 10.3390/su10103371.

- [49] Yi Li, Kailong Liu, Aoife Foley, Alana Zülke, Maitane Berecibar, Elise Nanini-Maury, Joeri Van Mierlo, and Harry Hoster. Data-driven health estimation and lifetime prediction of lithium-ion batteries: A review, oct 2019. ISSN 18790690.
- [50] Kailong Liu, Xiaosong Hu, Zhile Yang, Yi Xie, and Shengzhong Feng. Lithium-ion battery charging management considering economic costs of electrical energy loss and battery degradation. *Energy Conversion and Management*, 195:167–179, 2019. ISSN 0196-8904. doi: 10.1016/j.enconman.2019. 04.065.
- [51] Maxim Lu, Alex James, and Mehdi Bagheri. Unmanned aerial vehicle (uav) charging from powerlines. In 2017 IEEE PES Asia-Pacific Power and Energy Engineering Conference (APPEEC), pages 1–6, 2017. doi: 10.1109/APPEEC.2017. 8308912.
- [52] Nicholas Lutsey and Dale Hall. Effects of battery manufacturing on electric vehicle life-cycle greenhouse gas emissions. *International Council* on Clean Transportation, 02 2018.
- [53] Shuai Ma, Modi Jiang, Peng Tao, Chengyi Song, Jianbo Wu, Jun Wang, Tao Deng, and Wen Shang. Temperature effect and thermal impact in lithiumion batteries: A review. *Progress in Natural Science: Materials International*, 28(6):653–666, 2018. ISSN 1002-0071. doi: 10.1016/j.pnsc.2018. 11.002.
- [54] Sina Sharif Mansouri, Petros Karvelis, George Georgoulas, and George Nikolakopoulos. Remaining Useful Battery Life Prediction for UAVs based on Machine Learning. *IFAC-PapersOnLine*, 50 (1):4727–4732, jul 2017. ISSN 24058963. doi: 10.1016/j.ifacol.2017.08.863.
- [55] Peyman Mohtat, Suhak Lee, Valentin Sulzer, Jason B. Siegel, and Anna G. Stefanopoulou. Differential Expansion and Voltage Model for Li-ion Batteries at Practical Charging Rates. *Journal of The Electrochemical Society*, 167(11):110561, 2020. doi: 10.1149/1945-7111/aba5d1.
- [56] Melissa Morris and Sabri Tosunoglu. Comparison of Rechargeable Battery Technologies. In ASME Early Career Technical Conference (ECTC), 2012.
- [57] Mohammad Moshref-Javadi, Ahmad Hemmati, and Matthias Winkenbach. A truck and drones model for last-mile delivery: A mathematical model and heuristic approach. *Applied Mathematical Modelling*, 80:290–318, 2020. ISSN 0307-904X. doi: 10.1016/j.apm.2019.11.020.
- [58] Nirmal-Kumar C. Nair and Niraj Garimella. Battery energy storage systems: Assessment for small-scale renewable energy integration. *Energy and Buildings*, 42(11):2124–2130, 2010. ISSN 0378-7788. doi: 10.1016/j.enbuild.2010.07.002.

- [59] Nationaal Instituut voor Budgetvoorlichting. Energie en water. https://www.nibud.nl/ consumenten/energie-en-water/. Retrieved: 18-01-2022.
- [60] Maria Palacín. Recent advances in rechargeable battery materials: a chemist's perspective. *Chemical Society Reviews*, 38(9):2565–2575, 2009. doi: 10.1039/b820555h.
- [61] Sebastian Paul, Christian Diegelmann, Herbert Kabza, and Werner Tillmetz. Analysis of ageing inhomogeneities in lithium-ion battery systems. *Journal of Power Sources*, 239:642–650, 2013. ISSN 0378-7753. doi: 10.1016/j.jpowsour. 2013.01.068.
- [62] Michael Pecht and Anthony Rafanelli. Product Reliability, Maintainability, and Supportability Handbook. 2. CRC Press, 2009. ISBN 9780849398797. doi: 10.1115/1.2792151.
- [63] Ahmad Ali Pohya, Kai Wicke, and Johannes Hartmann. Comparison of direct operating cost and life cycle cost-benefit methods in aircraft technology assessment. In AIAA Aerospace Sciences Meeting, 2018. American Institute of Aeronautics and Astronautics Inc, AIAA, January 2018. ISBN 9781624105241. doi: 10.2514/6.2018-0282.
- [64] Seyed Mohammad Rezvanizaniani, Zongchang Liu, Yan Chen, and Jay Lee. Review and recent advances in battery health monitoring and prognostics technologies for electric vehicle (ev) safety and mobility. *Journal of Power Sources*, 256:110–124, 2014. ISSN 0378-7753. doi: 10.1016/ j.jpowsour.2014.01.085.
- [65] Thiago Rodrigues, Jay Patrikar, Arnav Choudhry, Jacob Feldgoise, Vaibhav Arcot, Aradhana Gahlaut, Sophia Lau, Brady Moon, Bastian Wagner, H. Scott Matthews, Sebastian Scherer, and Constantine Samaras. In-flight positional and energy use data set of a dji matrice 100 quadcopter for small package delivery. *Scientific Data*, 8, 06 2021. doi: 10.1038/s41597-021-00930-x.
- [66] Mia Romare and Lisbeth Dahllöf. The life cycle energy consumption and greenhouse gas emissions from lithium-ion batteries. *IVL Swedish Environmental Research Institute*, C243, 2017.
- [67] Bill Roungas, Sebastiaan Meijer, and Alexander Verbraeck. A framework for simulation validation & verification method selection. In Arash Ramezani, Edward Williams, and Marek Bauer, editors, *Proceedings of the 9th International Conference on Advances in System Simulation, SIMUL 2017*, pages 35–40, 2017. The 9th International Conference on Advances in System Simulation SIMUL 2017, SIMUL 2017 ; Conference date: 08-10-2017 Through 12-12-2017.

- [68] Bhaskar Saha, Edwin Koshimoto, Cuong Quach, Edward Hogge, Thomas H. Strom, Boyd Hill, Sixto Vazquez, and Kai Goebel. Battery health management system for electric UAVs. In *IEEE Aerospace Conference Proceedings*, pages 1–9, 2011. ISBN 9781424473502. doi: 10.1109/AERO. 2011.5747587.
- [69] Bhaskar Saha, Edwin Koshimoto, Cuong C. Quach, Sixto Vazquez, Edward Hogge, Thomas Strom, Boyd Hill, and Kai Goebel. Predicting battery life for electric UAVs. In AIAA Infotech at Aerospace Conference and Exhibit 2011, 2011. ISBN 9781600869440. doi: 10.2514/6.2011-1517.
- [70] Abhinav Saxena, José Celaya, Edward Balaban, Kai Goebel, Bhaskar Saha, Sankalita Saha, and Mark Schwabacher. Metrics for evaluating performance of prognostic techniques. In 2008 International Conference on Prognostics and Health Management, PHM 2008, 2008. ISBN 9781424419357. doi: 10.1109/PHM.2008.4711436.
- [71] Abhinav Saxena, José Celaya, Bhaskar Saha, Sankalita Saha, and Kai Goebel. Evaluating algorithm performance metrics tailored for prognostics. In *IEEE Aerospace Conference*, pages 1–13, 2009. ISBN 9781424426225. doi: 10.1109/AERO. 2009.4839666.
- [72] Abhinav Saxena, José Celaya, Indranil Roychoudhury, Bhaskar Saha, Sankalita Saha, and Kai Goebel. Designing Data-Driven Battery Prognostic Approaches for Variable Loading Profiles: Some Lessons Learned. *First European Conference of the Prognostics and Health Management Society 2012*, page 10, 2012.
- [73] Ping Shen, Minggao Ouyang, Xuebing Han, Xuning Feng, Languang Lu, and Li Jianqiu. Error analysis of the model-based state-of-charge observer for lithium-ion batteries. *IEEE Transactions on Vehicular Technology*, PP:1–1, 06 2018. doi: 10.1109/TVT.2018.2842820.
- [74] Haotian Shi, Shunli Wang, Carlos Fernandez, Chunmei Yu, Xiaoxia Li, and Chuanyun Zou. Adaptive iterative working state prediction based on the double unscented transformation and dynamic functioning for unmanned aerial vehicle lithium-ion batteries. *Measurement and Control*, 53(9-10):1760–1773, 2020. doi: 10.1177/ 0020294020923057.
- [75] Gina Sierra, Marcos Orchard, Kai Goebel, and Chetan Kulkarni. Battery health management for small-size rotary-wing electric unmanned aerial vehicles: An efficient approach for constrained computing platforms. *Reliability Engineering and System Safety*, 182:166–178, feb 2019. ISSN 09518320. doi: 10.1016/j.ress.2018.04.030.
- [76] Joshuah K Stolaroff, Constantine Samaras, Emma R O'Neill, Alia Lubers, Alexandra S

Mitchell, and Daniel Ceperley. Energy use and life cycle greenhouse gas emissions of drones for commercial package delivery. *Nature communications*, 9(1):1–13, 2018.

- [77] Valentin Sulzer, Scott G. Marquis, Robert Timms, Martin Robinson, and S. Jon Chapman. Python Battery Mathematical Modelling (PyBaMM). *Journal of Open Research Software*, 9(1):14, 2021. doi: 10.5334/jors.309.
- [78] Koji Suzuki, Paulo Kemper Filho, and James Morrison. Automatic battery replacement system for UAVs: Analysis and design. Journal of Intelligent and Robotic Systems: Theory and Applications, 65(1-4):563–586, jan 2012. ISSN 09210296. doi: 10.1007/s10846-011-9616-y.
- [79] Mohd Tariq, Ali Maswood, Chandana Jayampathi Gajanayake, and Amit Kumar. Gupta. Aircraft batteries: current trend towards more electric aircraft. *IET electrical systems in transportation*, 7: 93–103, 2017.
- [80] Rhea Valentina, Alexander Viehl, Oliver Bringmann, and Wolfgang Rosenstiel. Battery aging estimation for eco-driving strategy and electric vehicles sustainability. In 40th Annual Conference of the IEEE Industrial Electronics Society, pages 5622–5627. Institute of Electrical and Electronics Engineers Inc., feb 2014. ISBN 9781479940325. doi: 10.1109/IECON.2014.7049361.
- [81] Jorne Van den Bergh, Philippe De Bruecker, Jeroen Beliën, and Jonas Peeters. Aircraft maintenance operations: state of the art. Technical report, Katholieke Universiteit Leuven and Hogeschool-Universiteit Brussel, 2013.
- [82] Prakash Venugopal and Vigneswaran T. Stateof-health estimation of li-ion batteries in electric vehicle using indrnn under variable load condition. *Energies*, 12(22), 2019. ISSN 1996-1073. doi: 10. 3390/en12224338.
- [83] Venkatasubramanian Viswanathan and Matthew Knapp. Potential for electric aircraft. *Nature*, 2 (2):88–89, feb 2019. ISSN 23989629. doi: 10.1038/ s41893-019-0233-2.
- [84] Venkatasubramanian Viswanathan, Alan H Epstein, Yet-Ming Chiang, Esther Takeuchi, Marty Bradley, John Langford, and Michael Winter. The challenges and opportunities of battery-powered flight. *Nature*, 601(7894):519–525, 2022.
- [85] Vilayanur V. Viswanathan and Michael Kintner-Meyer. Second use of transportation batteries: Maximizing the value of batteries for transportation and grid services. *IEEE Transactions on Vehicular Technology*, 60(7):2963–2970, 2011. doi: 10.1109/TVT.2011.2160378.

- [86] Pippa Vossen. Sustainability and cost assessment of lithium battery health management strategies for electric unmanned aerial vehicles. MSc Thesis - Air Transport Operations, Delft University of Technology, 2022.
- [87] Wladislaw Waag, Christian Fleischer, and Dirk Uwe Sauer. Critical review of the methods for monitoring of lithium-ion batteries in electric and hybrid vehicles. *Journal of Power Sources*, 258:321–339, 2014. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2014.02.064.
- [88] Shunli Wang, Carlos Fernandez, Yongcun Fan, Juqiang Feng, Chunmei Yu, Kaifeng Huang, and Wei Xie. A novel safety assurance method based on the compound equivalent modeling and iterate reduce particle-adaptive kalman filtering for the unmanned aerial vehicle lithium ion batteries. *Energy Science & Engineering*, 8(5):1484–1500, 2020. doi: 10.1002/ese3.606.
- [89] Nick Williard, Wei He, Michael Osterman, and Michael Pecht. Predicting Remaining Capacity of Batteries for UAVs and Electric Vehicle Applications. In *IMAPS Advanced Technology Workshop* on High Reliability Microelectronics for Military Applications, 2011.
- [90] Bo Wu, Xisheng Jia, Lianghua Xia, Congying Lei, and Yabin Wang. Cost-benefit analysis of the application of prognostics and health management technology. In 2013 Proceedings of 2013 International Conference on Quality, Reliability, Risk, Maintenance, and Safety Engineering, pages 1923–1927, 2013. ISBN 9781479910144. doi: 10.1109/QR2MSE.2013.6625955.
- [91] Bolun Xu, Alexandre Oudalov, Andreas Ulbig, Göran Andersson, and Daniel S. Kirschen. Modeling of lithium-ion battery degradation for cell life assessment. *IEEE Transactions on Smart Grid*, 9(2):1131–1140, 2018. ISSN 19493053. doi: 10.1109/TSG.2016.2578950.
- [92] Xiao Guang Yang, Yongjun Leng, Guangsheng Zhang, Shanhai Ge, and Chao Yang Wang. Modeling of lithium plating induced aging of lithium-ion batteries: Transition from linear to nonlinear aging. Journal of Power Sources, 360:28–40, 2017. doi: 10.1016/j.jpowsour.2017.05.110.
- [93] Haitao Zhang, Ming Zhou, and Xudong Lan. State of charge estimation algorithm for unmanned aerial vehicle power-type lithium battery packs based on the extended kalman filter. *Energies*, 12(20), 2019. ISSN 1996-1073. doi: 10.3390/ en12203960.
- [94] Özgür Dündar, Mesut Bilici, and Tarık Ünler. Design and performance analyses of a fixed wing battery vtol uav. *Engineering Science and Technol*ogy, an International Journal, 23(5):1182–1193, 2020. ISSN 2215-0986. doi: 10.1016/j.jestch.2020. 02.002.

Appendices

Pouch cel	l					
Nominal capacity [Ah]	5.0					
Minimum voltage $[V]$	3.0					
Maximum voltage $[V]$	4.2					
Thickness [mm]	4.0					
Length [mm]	132					
Width [mm]	90					
Positive electrode						
Material	NMC:CB:PVDF (94:3:3)					
Number of double sided electrode sheet	ts 14					
Negative electrode						
Material	Graphite:PVDF (95:5)					
Number of double sided electrode sheet	ts 15					
Separator						
Material	Polyethylene (PE)					
Electrolyte						
Material	$1 \text{ M } LiPF_6$					
Organic solvent in electrolyte	2% EC:EMC (3:7)					

A Appendix A - Battery model specifications

Table 21: Lithium battery specifications [55]

Parameter	Source	$\mathbf{Distr.}^1$	Value			
		Initial bat	tery parameters			
Initial capacity [Ah]	[61]	Normal	$\mu=5,\sigma=1.3\%$			
Initial minimum voltage $[V]$	[22]	Normal	$\mu=3.0,\sigma=0.01\%$			
Initial maximum voltage $[V]$	[22]	Normal	$\mu=4.2,\sigma=0.01\%$			
		Battery ag	leing parameters			
SEI kinetic rate	[c]		$1.0 > \text{SOH} > 0.9: \ \mu = 1 \cdot 10^{-14}, \ \sigma = 1.5\%$			
constant $[ms^{-1}]$	[θ]	Normal	$0.9 > \text{SOH} > 0.8$: $\mu = 1 \cdot 10^{-14}$, $\sigma = -0.055^* \text{SOH} + 0.051$			
Battery Monitoring System						
Voltage $[V]$	[36, 88]	Uniform	Error margin $= \pm 1.0\%$			
SOC	[43, 74]	Uniform	$ m Error \ margin = \pm \ 2.0\%$			
SOH	[73, 82]	Uniform	$ m Error \ margin = \pm \ 2.5\%$			
1						

¹ Distr. = Distribution (probability)

Table 22: Battery and BMS non-deterministic error model inputs

This page intentionally left blank.

II

Literature Study previously graded under AE4020

1

Introduction

Maintenance is an important process in aviation to ensure safety and reliability requirements are met [41, 68]. Studying maintenance procedures has become a popular subject for many other reasons such as the minimisation of financial costs and improvement of efficiency [6, 73, 87, 102]. Commonly applied types of maintenance are Corrective Maintenance (CM), scheduled maintenance and Predictive Maintenance (PdM) [6, 124]. The most basic approach is CM, which is a reactive maintenance approach [41]. Scheduled maintenance is a proactive strategy that is structured by time- or usage-based intervals [53, 121], which is widely applied due to its simplicity and efficiency for planning extensive checks [1, 68]. Currently, PdM is increasingly applied within the maintenance domain. PdM, also known as Prognostics and Health Management (PHM) [119], is a preventive approach that monitors a component's performance to provide maintenance prognostics [74]. By incorporating predictive maintenance, the objective is to improve a system's reliability and safety while reducing monetary costs, increasing availability and elongating component lifetime [48, 111, 116]. To guide the decision-making process for which maintenance strategy is optimal for a given system, a Cost-Benefit Analysis (CBA) can be performed [54].

The aim of this study is to review the current literature on PHM practices in the aerospace industry, in order to set up a research that will contribute to this topic. Moreover, appropriate CBA methodologies are explored. This research is performed as part of an MSc thesis at the Faculty of Aerospace Engineering at Delft University of Technology. The thesis objective is to execute a CBA analysing the performance of a PHM approach. The fundamental building blocks of executing the thesis project are also covered in this report.

Part IIA evaluates the available literature on PHM. First, chapter 2 reviews the different types of maintenance techniques. For predictive maintenance, although the diagnostic part of such an approach is well established, the prognostic component requires further development [72, 110]. Moreover, given the current shift to more sustainable practices, a shortcoming is observed with respect to PHM studies on electric Aerial Vehicles (eAVs). Specifically, the rest of the report addresses PHM practices related to batteries, given their critical power supply role [19]. As batteries for large aircraft are not yet widely available [19, 125], this study focuses on electric Unmanned Aerial Vehicles (eUAVs). For eUAVs, Lithium batteries are most commonly used [28, 107, 118]. Hence, an in-depth analysis of Lithium battery health management methodologies is performed in chapter 3. Subsequently, chapter 4 discusses general performance metrics to evaluate predictive algorithms. Finally, maintenance costs as well as cost-benefit parameters are elaborated on in chapter 5. Here, the inclusion of sustainability parameters is also addressed.

Building on the findings in Part IIA a thesis project is defined. Part IIB first presents the problem statement in chapter 6. Here, the main objects, scope and research sub-questions are summarised. To perform the research, a brief overview of the project methodology and timeline is given in chapter 7.

This page intentionally left blank.

PART IIA

This page intentionally left blank.

2

Aircraft Maintenance

In this chapter, all topics directly related to aerospace maintenance is discussed. First, a general overview of the evolution of aircraft maintenance and important factors that need to be taken into account are elaborated on in section 2.1. Secondly, section 2.2 provides a detailed breakdown of the different types of maintenance strategies that exist. Here, the maintenance types corrective, preventive and predictive maintenance are specifically focused on. Finally, a conclusion of research gaps is drawn in section 2.3, highlighting the challenges and literature gaps encountered throughout aerospace maintenance practices.

2.1. General Background

The development and improvement of aircraft maintenance schedules is an ongoing process. Over the past couple of decades, the number of publications on this topic has grown significantly. For aircraft maintenance, Van den Bergh et al. [124] perceived a steady increase from approximately 10 to 30 papers issued yearly over a duration of 20 years compared to 1995. Moreover, for machine prognostics in general, Lei et al. [74] showed that the number of publications grew steadily from around 10 to over 200 pieces each year over 20 years compared to 1996, resulting in a total of 1426 papers. On the one hand, this is driven by the airworthiness of aviation. Aircraft safety is an important factor within the aviation industry, as a lack of integrity could lead to fatal accidents [41]. On the other hand, maintenance costs make up for approximately 10 - 20 % of an air carrier's total budget [6, 73, 87, 102]. Hence, optimising maintenance costs is beneficial.

In this section, the aspects of safety in aviation are first discussed in subsection 2.1.1. Then, relevant organisations and principles related to safety and maintenance are presented in subsection 2.1.2.

2.1.1. Safety in Aviation

As mentioned above, safety is an important factor within aviation. For safety-related to aeronautics, a distinction is made between the following types: man, environment and machine safety [41]. The first category, **man**, includes all people that are actively involved with flight operations such as the pilots, maintenance labourers and air traffic controllers. For **environment**, all external aspects that are related to air traffic are considered. For example, this category refers to meteorological conditions, communications and traffic zones. Finally, **machine** includes safety regarding aircraft construction. This last type is also known as **airworthiness**, which is especially relevant for maintenance. According to the Italian RAI-ENAC Technical Regulations, airworthiness is defined to be "the possession of the necessary requirements for flying in safe conditions, within allowable limits".

The airworthiness of aircraft depends on several factors including the materials, design and manufacturing process and operations and maintenance activities performed. Although aviation is often referred to as the safest mode of transport [35, 84], Duarte et al. [35] stress the importance of the continuation of research concerning flight safety. In their paper, several studies are addressed that support the thesis that failure analyses are of high importance concerning aircraft system safety.

Not only on the technical front but also from a social point of view, further development is important. Lack of communication between parties, such as the manufacturer and operator or an aircraft, can have a negative

influence on maintenance processes [50] or even result in fatal accidents [35]. If component failures are not communicated to the manufacturer, the design cannot be adjusted to correct these errors. Conversely, the maintainability of aircraft highly depends on the technical support given by the manufacturer to the operators and maintenance engineers. The percentage of accidents assigned to be the result of human errors vary vastly per study, ranging from 21% [128] to 96% [59]. However, it must be noted that human errors can be classified into several different stages, starting at an organisational level that ultimately influences all maintenance actions performed at lower stages [120]. Nevertheless, it can be concluded that the assurance of airworthiness is a joint effort between all parties that are related to aviation practices.

2.1.2. Organisations and Regulations

Aviation developed rapidly around World War I and World War II. In order to monitor technological developments, the International Civil Aviation Organisation (ICAO) was established in 1947 [41]. The ICAO's aim was to set up techniques and principles for international aviation to ensure safe and efficient service. Ultimately, the ICAO delivered 18 annexes that each cover air navigation-related subjects. Here, recommended maintenance practices are predominantly discussed in Annex 6 'Operation of Aircraft', Annex 8 'Airworthiness of Aircraft' and Annex 16 'Environmental Protection'.

Subsequently, authorities were established to ensure that airworthiness standards were actually met [41]. The authorities have the task to prescribe requirements and procedures, to inform all air navigation parties and control that the prescriptions are followed up. These regulations apply for aircraft design, manufacturing and material organisations, as well as air navigation operators. Finally, an airworthiness authority has the task of certificating materials and organisations. To stimulate cooperation between authorities, Bilateral Aviation Safety Agreements (BASA) were initialised. When concerning aircraft airworthiness, an additional document Implementation Procedures for Airworthiness (IPA) is developed to ensure all standards are met. Here, specific protocols are defined including activities related to aircraft design, production, repair and other technical operations.

Different authorities exist, partially newly established and partially evolved from former organisations [41]. Most authorities operate on a national level, but steps are being made to harmonise these air navigation regulations. For example, in 2016, the Certification Management Team (CMT) was officially founded [84] [32] consisting of four of the largest aviation authorities being the Federal Aviation Administration (FAA) in the United States, Transport Canada Civil Aviation (TCCA), Agência Nacional de Aviação Civil (ANAC) in Brasil and finally the European Aviation Safety Agency (EASA), that took control over the no longer existing Joint Aviation Authority (JAA) in 2007 [101]. The aim of the is to further develop and implement aviation regulations and policies globally in an efficient and effective manner.

2.2. Maintenance Types

Maintenance was introduced to improve the reliability of a system [65]. A common maintenance topic that is often elaborated on by researchers, are aircraft engines. Van den Bergh et al. [124] state that engines are safety-critical systems and therefore considered to be very important for maintenance. Engines are also very complex and expensive [102], making them an important maintenance factor to consider. Engine maintenance is performed at a specialised facility, usually depending on the type of engine. Hence, it is important to take engine maintenance into account for scheduling.

Not just engines but entire aircraft are subject to maintenance to improve the performance and reliability of a system. Different maintenance strategies exist which are further elaborated on in this section. In Figure 2.1, a breakdown tree of existing maintenance types is given. The main two categories that the types of maintenance can be distinguished, are Corrective Maintenance (CM) and Preventive Maintenance (PM) [53]. The difference between these categories is that CM is reactive, whereas PM is proactive.

The maintenance abbreviations in Figure 2.1 are:

- DOM Design-Out Maintenance
- FBM Failure-Based Maintenance
- **RTF** Run to Failure
- BM Breakdown maintenance
- TBM Time-Based Maintenance

- UBM Usage-Based Maintenance
- CBM Condition Based Maintenance
- CBM+ CBM, including a reliability analysis
- **PdM** Predictive Maintenance
- DBM Detection-Based Maintenance
- PHM Prognostics & Health Management

The definitions of these maintenance policies are discussed throughout this section.



Figure 2.1: Breakdown Tree of Maintenance Types [53]

First, subsection 2.2.1 elaborates on the principles of CM. Then, several approaches within PM are discussed in subsection 2.2.2, including scheduled and unscheduled maintenance types. Finally, subsection 2.2.3 explores several integrated maintenance approaches.

2.2.1. Corrective Maintenance

The most basic type of maintenance is Corrective Maintenance (CM), also known as Failure Based Maintenance (FBM), Run-To-Failure (RTF) or Breakdown Maintenance (BM), as shown in Figure 2.1. This is a maintenance type that is carried out after a fault is detected and therefore considered to be reactive [6, 24, 53, 124]. Failures occur unexpectedly, implying that CM is an unscheduled maintenance type [6, 53].

There are several reasons why CM is considered undesirable:

• Reliability

Depending on the severity of the failure condition and how critical the component is, implementing the CM strategy may lead to fatal accidents [41, 100].

• Availability

During maintenance, a component or entire vehicle is not available for use. Unscheduled maintenance may affect an operation schedule and cause undesired downtime resulting in an increase in operator costs [15, 99].

Secondary damages

Letting a component run-to-failure, secondary damages or failures may be inflicted to other components, resulting in higher total maintenance costs or even critical situations [24, 67].

Nevertheless, for some non-critical components, CM may be the most cost-effective maintenance strategy [53]. Depending on the system, CM may therefore be determined to be the optimal maintenance type.

2.2.2. Preventive Maintenance

In contrast to the corrective type, Preventive Maintenance (PM) has the aim to restore the system level of perfection by performing maintenance before failure has occurred [68]. By applying this maintenance strategy for critical components, the system's reliability is improved [99].

Design-Out Maintenance (DOM), as depicted in Figure 2.1, is not a synonym for PM, but a stand-alone category [36]. In this strategy, the design of a component is altered to account for shortcomings discovered during the component's usage.

Scheduled maintenance is often used as a synonym for PM [124]. However, there are also several other types of unscheduled PM that exist. These scheduled and unscheduled types of PM are presented in the following subsections.

2.2.2.1. Scheduled Maintenance

Two types of scheduled maintenance can be distinguished. Time-Based Maintenance (TBM) is a PM type of maintenance that is linked to fixed time intervals [65]. The second type is Usage-Based Maintenance (UBM), which is based on relevant usage parameters such as flight hours.

Obligatory scheduled checks have been defined by Federal Aviation Administration (FAA) to monitor the airworthiness of an aircraft, referred to as A-, B-, C- and D-checks [124]. These are checks that the operator is obliged to carry out after a specific number of flight hours or calendar days, varying in time interval duration and maintenance extensiveness.

TBM schemes are based on past failure time data to ensure safety requirements are met [72]. However, Acktert [1] concludes that using fixed TBM schemes are not beneficial for cases where the component failure is not dominant. Moreover, Lee et al. [72] argue that TBM is not effective if the current health state of the component is not taken into account. This motion is backed by Andreacchio et al. [6] who point out that TBM is not effective if for example CM has been performed within a scheduled time interval.

2.2.2.2. Condition Based Maintenance

Remedying the shortcomings for TBM mentioned above, Condition Based Maintenance (CBM) recommends maintenance based on the observed health status of a system or component [48, 65, 116, 121]. As CBM is a PM type, components are repaired or replaced before the event of total failure. The item's health is continuously monitored with the use of sensors. This activity of collecting data through sensors is called **condition monitoring** [65]. Repairing the system or component is done on an unscheduled basis.

In Figure 2.1, CBM+ and Detection-Based Maintenance (DBM) are also depicted as maintenance types. These strategies are very similar to CBM. CBM+ is more elaborate than CBM with a stronger focus on prognostics and reliability [14]. In DBM, condition monitoring is done through operator observations [126], instead of system monitors.

In the ISO-13374 documentation, Condition Monitory and Diagnostics of Machines [94], the process of CBM is stated to consist of the following six parts:

- 1. data acquisition,
- 2. data handling,
- 3. state determination,
- 4. health assessment,
- 5. prognostics, and
- 6. recommendation.

Common condition monitoring techniques include acoustic emission analysis, vibration monitoring and thermography mapping [50, 65, 121]. Three different monitoring data types are typically recognised: **value** (e.g. single measurement of temperature), **waveform** (e.g. sound wave) and **multidimensional** (e.g. 2D image) [65]. The sensor choice takes the criteria such as the installed weight and damage detection capabilities into account.

Data acquisition can either be done continuously or periodically, the latter commonly chosen as this set-up is less expensive and less susceptible to noise. A disadvantage of periodic condition monitoring is however the risk of missing a failure. Thus, research regarding the monitoring interval is very important [33, 121].

Condition monitoring accuracy is increased by combining information obtained through multiple sensors

monitoring the same system [65]. To avoid errors, data is always first cleaned before data analysis is done. To combine data input from multiple sensors, data-level, feature-level or decision-level fusion methods are used. Depending on the acquired data, reduction techniques and or regression or time series analyses are applied.

A health monitoring technique that is closely related to condition monitoring, is called Structural Health Monitoring (SHM) [121]. This method originates from inspecting structures and uses non-destructive testing techniques to determine the structure's health. SHM mainly focuses on increasing the probability of detecting a failure and can use active or passive sensors [33]. Here, active sensors represent those that detect damages by sending a signal and reviewing the results, while passive sensors detect signals that are generated by failures without external excitation.

Summarising steps 1 to 6 mentioned above, CBM consists of a diagnostic and an advisory phase [94, 116]. Diagnostic methods revolve around the mapping of the condition monitoring information to the machine fault, which can be done manually or automatically (e.g. statistics or artificial intelligence driven) [65]. Manual methods require high expertise. However, while an automatic approach is more simple, the lack of data slows down the development of data-driven models.

The diagnostic part is well established. However, the second part of CBM concerning the health analysis and advisory step are less well-developed and thus have limited applied examples [121]. During the prognostics phase, the Remaining Useful Life (RUL) is determined by combining the health assessment insights with knowledge obtained from past degradation trends. Furthermore, the probability of failure before the next TBM check is computed. Maintenance is planned when the prognostics indicate that a predefined critical threshold is passed [72, 74, 116]. Prognostic methods are based on a physical model, data-driven analysis or a hybrid combination of both approaches [37].

In Figure 2.2, a graphical depiction is given of the RUL prediction process [74]. At t_{FPT} (first prediction time, predefined through statistical analysis) a prediction is made of the RUL, represented by the green line. The probability density function (PDF) determines the confidence interval, expressing the range of time in which the End-of-Life (EOL) is predicted to occur. In this figure, the real-life component health is depicted by the red continuous line. The actual t_{EOL} (time of EOL) is finally reached when the health passes the set Failure Threshold (FT).



Figure 2.2: Determination of the Remaining Useful Life in Predictive Maintenance [74]

RUL predictions are done multiple times throughout the component's lifetime. Based on the evaluation of the data, with the aim to repair the component before failure occurs, a maintenance recommendation is given [65]. A drawback of CBM is that the maintenance moments cannot be plannend [48, 116].

2.2.2.3. Predictive Maintenance

Following up on CBM, Predictive Maintenance (PdM), which is also a type of unscheduled maintenance, takes the prognostics phase one step further [48, 116]. PdM is also referred to as Prognostics and Health Management (PHM) [119]. The aim of prognostics is to improve maintenance through for example the enhancement of safety and reliability, the reduction of costs and the increase of availability and elongation of a component's lifetime [65, 111, 121]. Moreover, authors such as Franciosi et al. [43] and Hoang et al. [58] point out that using prognostics also has potential to reduce the environmental impact of maintenance.

Similar to CBM, PdM monitors the health of a component or system through condition monitoring [74]. Moreover, event data is collected to include information regarding the components usage and maintenance activities. Like in CBM, prognostics are made by evaluating the monitored data and predicting the RUL of the component. Again, RUL predictions can be data-driven or based on a modelled approach, or a combination of these methodologies [37, 72, 74].

What distinguishes PdM from CBM, however, is the planning component that PdM can incorporated [48, 116]. Busse et al. [22] refer to two types of PdM prognostics. The maintenance can either be executed immediately (known as 'trigger strategies') or it can be planned ahead (referred to as 'planning strategies'). Here, a cost analysis is done to evaluate whether the trigger or planning strategy is more beneficial. Hence, it becomes apparent that in PdM tries to synchronize operation and maintenance activities such that the system's downtime is near-zero [72]. Furthermore, with PdM, a TBM scheduled maintenance plan of a given component may be altered as the time since the last maintenance activity has changed, which is not considered in CBM [50]. By taking the overall scheme of the system into account, PdM aims to achieve cost-effective maintenance schedules, whereas CBM cannot be planned [48, 116].

2.2.3. Integrated Approach

Instead of implementing merely one of the above-described types (CM or PM) as an aviation maintenance strategy, aircraft undergo a combination of these maintenance types [124]. A corrective approach is necessary when a defected fault has occurred, while preventive and predictive strategies are implemented to increase availability and reduce costs.

Reviewing the effectiveness of maintenance schemes is an important research topic [48, 121]. Tinga et al. [121] researched the similarities and differences of three maintenance disciplines: CBM, SHM and PHM. In their paper, they stress the benefit of combining the three approaches in order to enhance the overall maintenance effectiveness. Instead of using only data-driven approaches to provide maintenance prognostics, an integration with a physical model-based approach yields a better understanding of the system's health.

Another example that focuses on combining multiple types of maintenance strategies, is given by Andreacchio et al. [6]. In their research they state that PM schedules do not take unscheduled CM activities into account, resulting in premature aircraft asset replacement. As a solution for this problem, they propose the implementation of a Cyber-Physical Systems (CPS) approach to enhance aircraft asset utilisation and reduce replacement costs.

2.3. Research gaps

Based on the reviewed literature, two main gaps are identified. The first shortcoming is regarding prognostics, which is discussed in subsection 2.3.1. The second gap is concerning the type of aerial vehicles that are commonly researched, which is elaborated on in subsection 2.3.2.

2.3.1. Prognostics

As stated in subsection 2.2.2, the diagnostics part within PdM is already well established, while prognostics still require further research. Saxena et al. [110] and Lee et al. [72] point out that prognostic concepts are often inconsistent and ambiguous and lack standard definitions and that more research is required regarding the prediction of a system's performance and degradation. Furthermore, for data-driven prognostics, they stress that the shortage of data impedes the verification and validation phase required to evaluate prognostic theories.

2.3.2. Electric Aerial Vehicles

Recently there has been an increase in the usage of electric propulsion systems as these are more sustainable compared to conventionally fuel propelled aircraft [28, 118]. In addition to the sustainable advantages, electric power has control and operational advantages such as responsive thrust and noise reduction [27, 125]. Thus it is not surprising to see that there is a growth in research regarding battery development for aviation,

focusing on rechargeable batteries.

For maintenance, the most apparent difference between conventionally fuelled and electrically propelled aviation practices revolves around batteries [19]. Despite the increase in the use of electric propulsion systems, limited literature is found on battery PHM practices applied within the aviation industry.

In the interest of sustainability, battery PHM is chosen for more in-depth analysis in the following chapter. Batteries are used in More Electric Aircraft (MEA), electrical Aerial Vehicles (eAVs) and electrical Unmanned Aerial Vehicles (eUAVs) [19, 28, 118]. MEA are aircraft in which all non-propulsive systems are run on electric power, while eAVs and eUAVs are fully electric. Although all three types are electrically powered, it is chosen to focus on eUAVs in this study due to the availability of battery data. Small batteries are already widely used to power eUAVs, while battery compositions for MEA and eAVs are not yet widely available [19, 125]. Hence, throughout the rest of this literature study, battery practices for eUAVs are reviewed.

This page intentionally left blank.

3

Battery Health Management

Based on the research gap findings within the scope of aircraft maintenance, presented in subsection 2.3.1 and 2.3.2, deeper analysis of health management for batteries used in eUAVs is recommended. Literature on this subject is explored in this chapter.

First, in section 3.1, the relevance of reviewing battery health management for eUAVs is highlighted. Secondly, several types of batteries are briefly addressed in section 3.2. Thirdly, section 3.3 explains the most important battery characteristics related to health monitoring. Then, several battery health monitoring example approaches are reviewed in section 3.4. In section 3.5, several battery data and simulation tool sources are briefly listed. Finally, the main research gaps from this chapter are summarised in section 3.6.

3.1. Relevance

As previously addressed in subsection 2.3.2, electric power sources are increasingly used in aerial vehicles. While electrically powered non-propulsive systems are common in practice, the low energy density of batteries is still the biggest obstacle for large electrically propelled aircraft [125]. Currently, for fully electrical flight, batteries are merely able to achieve an energy density that is 60 times smaller than the energy density related to kerosene [19].

For smaller eUAVs, batteries are more frequently used as a power source for both non-propulsive and propulsive systems. eUAVs are favourable for several reasons [8, 52, 60, 61, 85, 113] including the fact that they have lower operating costs, lower environmental impact and are able to execute missions in extreme or dangerous environments compared to manned aircraft. Furthermore, eUAVs play important roles in a diverse range of activities including military missions, search and rescue operations, security patrols, agriculture tasks and mapping exercises.

Jing and Haifeng [29] and Rezvanizaniani et al. [104] reviewed the feasibility of implementing prognostic technologies of PHM to improve the battery health management capabilities and reliability for eUAVs and electric vehicles, respectively. For batteries, prognostic practices for example include predicting the moment when the battery reaches End-of-Discharge (EOD) and End-of-Life (EOL). However, future challenges they recognised are listed below, implying that this topic requires further research.

- · shortage of understanding in technologies,
- · difficulty in realising real-time data processing,
- challenge in filtering data,
- · complexity in representing and managing uncertainty related to the safety throughout battery life, and
- lack of data for validation and verification.

3.2. Battery Type

Several different types of batteries exist. Batteries are categorised into the following classes [90]:

- 1. **Primary** Discarded after used, these batteries can only be discharged once.
- 2. Secondary Rechargeable batteries.
- 3. **Reserve** Batteries that are activated when necessary and that can be stored over long periods of time.

The most common battery type in aviation is Lithium as these batteries are lightweight, have a high energy density, have a low discharge rate (implying their charge holds longer) and are characterised by a relatively long total lifetime [28, 90, 95, 107, 117, 118]. Lithium batteries are secondary type cells, and thus rechargeable.

Multiple different types of Lithium batteries exist. The first sort is Lithium-ion (Li-ion), which has several compositions related to the active materials present for the cathode and anode within the battery [66]. The active materials within the Li-ion battery result in small battery characteristics discrepancies [47]. The second type is Lithium-Polymer (Li-Po), which is a solid-state battery, implying that it uses solid polymer as an electrolyte [23].

An overview of the most common Li-ion battery types (name, abbreviation and scientific formula) is given in Table 3.1, along with a hexagonal spider graph of their characteristics in Figure 3.1 [23]. In this figure, the following characteristics are depicted:

1.	Specific energy	-	Capacity related to run time.
2.	Specific power	-	Ability to output a high current load.
3.	Safety	-	Stability of chemical processes within the battery.
4.	Performance	-	Usage performance at cold and hot temperatures.
5.	Life span	-	Lifetime and cycle life.
6.	Cost	-	Material, manufacturing and quality control expenses.

From the hexagonal diagrams, it becomes apparent that LMO batteries have a moderate overall performance. The types LCO, NMC and NCA show outstanding specific energy performance while LFP and LTO batteries have the best life span characteristics. Battery performance is similar for most types, though LTO batteries excel in this area. For specific power, the highest throughput is provided by LFP types. The safest performance is provided by LFP and LTO types. The battery costs are approximately equal for LCO, LMO, NMC and LFP batteries. From the charts, it can clearly be concluded that LTO types are the most expensive [23]. Depending on the battery application, different performance characteristics are desired. It is therefore important to evaluate the power source requirements before choosing a battery type.

Li-Po batteries are very similar to Li-ion batteries, though there are several small discrepancies. Compared to Li-ion types, Li-Po batteries have a slightly higher energy density and can be manufactured more thinly. Li-Po casing is able to be flexible, which further reduces the overall weight of the battery composition. The disadvantage of Li-Po batteries is that they are more costly [23].

In a study on eUAV batteries, Eleftheroglou et al. [37] state that they chose to research Lithium-Polymer batteries as they are easily monitored. Subsequently, the authors point out that Lithium batteries, in general, are relevant study cases as they commonly do not function well once voltage levels drop past a critical threshold, which is further elaborated on in section 3.3. Hence, it is useful to study indicators that predict the future performance of battery power systems.

3.3. Battery Characteristics

In order to understand the types of battery health management that can be applied, it is important to review the general background information regarding battery characteristics. First, battery performance and reliability is reviewed in subsection 3.3.1. Here, battery charge curves throughout a charge and discharge cycles are evaluated. Then, in subsection 3.3.2, battery degradation over multiple cycles is discussed.

3.3.1. Performance and Reliability

Battery performance and reliability are characterised by its capacity and internal resistance [78]. Capacity relates to how long the battery can provide energy, while the internal impedance limits the maximum power

Lithium Cobalt OxideLCOLiCoO2Lithium Manganese OxideLMOLiMn2O4Lithium Nickel Manganese Cobalt OxideNMCLiNiMnCoO2Lithium Iron PhosphateLFPLiFePO4Lithium Nickel Cobalt Aluminum OxideNCALiNiCoAlO2Lithium TitanateLTOLi2TiO3	Name	Abbreviation	Formula
Lithium Manganese OxideLMOLiMn2O4Lithium Nickel Manganese Cobalt OxideNMCLiNiMnCoO2Lithium Iron PhosphateLFPLiFePO4Lithium Nickel Cobalt Aluminum OxideNCALiNiCoAlO2Lithium TitanateLTOLi2TiO3	Lithium Cobalt Oxide	LCO	LiCoO ₂
Lithium Nickel Manganese Cobalt OxideNMCLiNiMnCoO2Lithium Iron PhosphateLFPLiFePO4Lithium Nickel Cobalt Aluminum OxideNCALiNiCoAlO2Lithium TitanateLTOLi2TiO3	Lithium Manganese Oxide	LMO	$LiMn_2O_4$
Lithium Iron PhosphateLFPLiFePO4Lithium Nickel Cobalt Aluminum OxideNCALiNiCoAlO2Lithium TitanateLTOLi2TiO3	Lithium Nickel Manganese Cobalt Oxide	NMC	LiNiMnCoO ₂
Lithium Nickel Cobalt Aluminum OxideNCALiNiCoAlO2Lithium TitanateLTOLi2TiO3	Lithium Iron Phosphate	LFP	LiFePO ₄
Lithium Titanate LTO Li ₂ TiO ₃	Lithium Nickel Cobalt Aluminum Oxide	NCA	LiNiCoAlO ₂
	Lithium Titanate	LTO	Li ₂ TiO ₃

Table 3.1: Overview of common Lithium-ion battery types listing the name, abbreviation and formula [23]



Figure 3.1: Hexagonal spider graphics of performance characteristics for Lithium-ion battery types LCO, LMO, NMC, LFP, NCA and NMC, showing specific energy, specific power, safety, performance, life span and cost [23]

level the battery is able to provide. According to Williard et al. [129], from these two parameters, the battery's capacity is the main characteristic used for performance and reliability evaluation.

Capacity is the electrical charge (Q) measured in the battery, expressed in Coulombs [C] or Ampere-Hours [Ah] [129]. To indicate a battery's performance and reliability, the maximum discharge capacity (Q_{max}) is typically referred to, as this shows the total amount of charge that a fully charged battery is able to deliver to its entirely discharged state. Hence, capacity is important as it determines how long the eUAV can fly. Not all batteries are identical, hence batteries' true Q_{max} values are distributed around the manufacturer's average capacity estimation [129]. When working with large data sets, such distributions are often represented by means of a normal distribution which is characterised by a specific mean and variance [31]. Baumhöfer et al. [13] also showed that this is the case for the initial capacity of a set of identical Lithium batteries.

Internal resistance (*R*), also referred to as internal impedance, is expressed in Ohm [Ω]. This parameter influences the maximum achievable throughput Current (*I*), measured in [*A*], within the system. A higher resistance results in restricted current, a drop in voltage and temperature rises within the battery [23]. For hybrid electric vehicles, resistance is an important performance indicator as an increase in battery impedance leads to lower acceleration capabilities or higher fuel usage [86]. For eUAVs, the maximum achievable power level is especially important for the take-off phase, as this requires the most power [125].

In the context the of current rate, the term C-rate is regularly used [23]. If a battery is discharged at 1C, it implies that when fully charged, it can provide a specific level of constant current for 1 hour. The level of current it can constantly discharge is governed by the initial Beginning-of-Life (BOL) battery capacity.

A battery's Q_{max} changes over time within a discharge cycle, affecting the power system's performance and reliability. This is more elaborately discussed in subsequent sections. Contrarily, the *R* within a discharge cycle is almost constant, leading to no major effects [23].

3.3.1.1. State of Charge

The State of Charge (SOC) refers to the remaining available charge of a system. For eUAVs, the mission duration is critical information to determine if the SOC is sufficient to complete the mission [129]. The SOC of a battery is 100% when it is fully charged, and defined to be 0% when it is fully discharged [20].

A battery's SOC is related to the voltage and current during a charge or discharge process as well as the battery's electrochemical properties. The SOC is defined by Equation 3.1, where Q(t) is the capacity at any given moment of time and Q_{max} is the nominal capacity at the beginning of the current cycle [26].

$$SOC(t) = \frac{Q(t)}{Q_{max}}\%$$
(3.1)

Within a cycle, the decay of SOC over time is related to the voltage. For an open circuit, this relationship is called Open Circuit Voltage (OCV) hysteresis [10]. The variation of OCV with respect to SOC is highly nonlinear [98]. Lithium batteries exhibit hysteretic behaviour both during charge and discharge cycles. However, the curves of these two processes differ slightly, as can be seen in Figure 3.2 [11]. For LFP batteries, the shape of this curve in Figure 3.2 is very typical. Note that the arrows in the graph represent the hysteresis orientation depending on whether the battery is undergoing a charge (SOC from 0 to 100%) or discharge (SOC from 100 to 0%) cycle. Baronti et al. [11] point out that LFP batteries are widely used due to their steady OCV throughput for SOC levels between 20 and 80%, which is very different from lead-acid battery behaviour. This OCV hysteresis curve differs slightly depending on the active materials, but all Lithium batteries exhibit similar flat OCV trajectories. For NMC battery types, for example, the OCV curve is slightly more slanted between 20% and 100% [134].



Figure 3.2: Typical Open Circuit Voltage hysteretic behaviour of a Lithium Iron Phosphate battery subject to a charge and discharge process at 20Ah [11]

The rate of SOC decrease within a cycle depends on the eUAV's power consumption throughout a mission. The flight load profile is therefore very important to accurately determine the required SOC needed to execute the mission. Saha et al. [108] characterised the typical flight load profile for eUAVs during flight which is depicted in Figure 3.3. Here, as previously stated, it can clearly be seen that take-off and landing required the most power.

Other parameters that influence power consumption are temperature, weight, density altitude, wind direction and flying velocity [12, 113].

The SOC can be determined through either an offline or online approach [20]. In the offline method, historical data is analysed to provide an accurate SOC prediction. With an online approach, real-time metrics are directly translated to a SOC estimation. Commonly these approaches are combined, such that the model is first trained using past data after which the online method can be applied [37].



Figure 3.3: Typical eUAV flight load during flight [108]

3.3.1.2. End-of-Discharge

The battery's End-of-Discharge (EOD) is at the end of the discharge cycle when SOC = 0% [17]. The SOC can be used to determine the EOD, which is similar to the estimation process of RUL presented in subsection 2.2.2. During operation, eUAV battery monitoring of SOC is done to ensure safe flight by observing and predicting if the battery has sufficient charge to complete the mission. Instead of letting the battery run to 0% charge, a capacity threshold is set **before** battery failure at 0% charge [85]. For Lithium batteries, this critical threshold is set to be when the battery has 30% SOC [60, 61, 112]. As authors Hogge et al. [60] point out, operating am eUAV with less than 30% SOC is defined to be high risk. This is due to the sudden voltage drop that occurs soon after passing the 30% SOC level, as can be seen in Figure 3.2, called the 'knee point' [112]. Upon reaching the critical threshold of 30% SOC, the batteries still have sufficient power to perform a minimum of two landing attempts safely [60].

3.3.2. State of Health

A battery's capacity and internal resistance change over time, affecting the battery's health. The battery's State of Health (SOH) is used as a metric to show the power source's number of remaining cycles before EOL [129]. The SOH is calculated using Equation 3.2, with Q_{max} being the nominal capacity at the start of a cycle and Q_0 the initial capacity at the battery's BOL [12].

SOH (t) =
$$\frac{Q_{max}(t)}{Q_0}$$
% (3.2)

Batteries degrade over time, this process is called ageing. As a consequence, two main characteristics change [12, 20]. Firstly, batteries suffer from **capacity loss**. This is due to the reduction in the available amount Lithium within the battery, resulting in a shorter discharge time for the same output current. Secondly, batteries experience an **increase in internal impedance** which causes the battery output voltage to decrease. For Lithium batteries, these health indicators degrade independently from each other [66].

Although some authors such as Viswanathan and Knapp [125] state that it is unknown whether capacity fade or resistance increase will be the limiting factor for eUAV applications, some researchers in the eUAV field only use SOH = 80% as the EOL threshold [96, 97, 129].

Two types of ageing categories exist:

Calendar ageing

This type of ageing includes all degradation that is not dependent on charge/discharge battery cycles [66]. Calendar ageing results in battery capacity loss that behaves according to a power law over time [78]. Moreover, this degradation process is weighted by the storage level of SOC and environment temperature [12, 78].

• Cycle ageing

This ageing kind is related to the charge/discharge cycles of a battery and many other different parameters, making it difficult to estimate or predict. Variables include cycle number, cycle duration, temperature, current rate, voltage range, Depth-of-Discharge (DOD) and average cycling SOC. Here, DOD is the range of SOC (Δ SOC) throughout the charge/discharge cycle of the battery [12, 78].

Saxena et al. [112] reviewed data-driven PHM approaches for batteries. After cycling the batteries (charge/discharge), the ageing phenomena described above can clearly be deduced. Using NASA's prognostic battery dataset [2], the researchers' findings are depicted in Figure 3.4. For one single battery, the same discharge process is repeated during which the battery starts at a discharge voltage of 4.2 *V* and is cut off at 2.7 *V*. Within one cycle, a similar discharge curve to the Lithium discharge trajectory in Figure 3.2 is observed. Then, as the cycles progress, the discharge curves in Figure 3.4 slowly shift. In the graph, it can be seen that cycles number 2 and 588 begin at the same discharge voltage at time = 0 *s*. Then, as time progresses, cycle 588 shows a steeper decrease and ultimately reaches the discharge cut-off voltage sooner than the battery did in cycle 2. Also, the battery's knee point has shifted to a slightly lower voltage level.



Figure 3.4: Discharge cycles at different battery life stages for constant load profiles from one single Lithium battery [112]

Battery ageing is unavoidable. Nonetheless, there are several parameters that accelerate this degradation process. Below, the effect of temperature, current rate, SOC and DOD on battery degradation are presented [47, 66, 83, 123, 132]. Then, it is briefly described how these parameters relate to eUAV applications.

1. Temperature

In practice, Lithium batteries can be operated between -20 and +60°C, while the temperature region from +15 to +35°C is considered to be optimal. For temperatures below 0°C, lower temperatures cause expedited battery degradation. On the contrary, for temperatures above 0°C, higher temperature levels lead to accelerated capacity loss and impedance gain. Temperatures affect battery ageing both during operation and rest. During active use of the battery, however, Xu et al. [131] point out the importance and complexity of decoupling temperature rise with other parameters such as rate of discharge. For storage, temperature ageing effects can more easily be decoupled. In a study by Keil et al. [66] the impact of higher temperatures during a battery's resting period is highlighted. For different Lithium chemical compositions (NCA, NMC and LFP), the study showed that higher storage temperatures lead to accelerated capacity degradation and an increase in internal resistance.

2. Current rate during charge and discharge process

High values of load current lead to accelerated battery ageing. For Lithium batteries, Ning et al. [93] found that the battery degradation is significantly influenced by the C-rate of the discharge cycles. When the battery was cycled at 1C discharge rate, the capacity loss was 9.5% after 300 cycles. In contrast, for 2C and 3C discharge rates the capacity degradation was 13.2 and 16.9%, respectively.

3. Depth-of-Discharge (DOD)

Larger DODs during charge and discharge cycles lead to accelerated battery capacity loss. Xu et al. [132] researched the effect of various different SOC % ranges during Lithium battery cycling. In Figure 3.5 it can clearly be seen that the smallest DOD interval (65-75%) results in the least battery capacity fade, while the battery cycled at the largest DOD (25-100%) shows the most severe degradation.

4. Average State of Charge (SOC) during operation and resting level

Batteries operated around high SOCs and/or stored at high SOC levels during rest period, show accelerated battery degradation [47, 66]. For Lithium batteries the capacity reduction is accelerated and depending on the active material types, the internal resistance growth is further stimulated. For the same DOD of 20% SOC, Gao et al. [47] concluded that battery capacity decrease is significantly more around higher SOCs (80-100%) compared to lower SOCs (0-20%), as depicted in Figure 3.6. Note the 'clusters' of degradation behaviour trends, as the battery capacity fades similarly for the DODs ranges 20-40%, 40-60% and 60-80%, while intervals 0-20% and 80-100% deviate significantly. In this figure, 'equivalent full cycles' are cycles that where the capacity throughput is normalised with the nominal battery capacity, such that different DOD can be compared.

For eUAV applications, regulating these above-mentioned parameters is not always trivial. For temperatures, studies show that operating eUAVs in extreme temperatures strongly affects the battery performance. For example, Li et al. [76] concluded that both high and low temperatures have a significant impact on eUAV battery degradation. Unfortunately, the location at which the mission must be executed cannot always be chosen. For charging strategies, however, regulating the temperature is more realisable. Though conclusions by Liao et al. [79] suggest that the charging temperature has a less significant effect on capacity degradation compared to the discharging temperature.

For current rates during the charging cycle of a battery, dedicated charging schedules could improve battery life. Klein et al. [69] explain that the most common charging method for Lithium batteries is using a Constant Current and Constant Voltage (CC/CV) approach. Manufacturers state the limits for CC/CV charging methods, as exceeding the prescribed values during a charging cycle leads to accelerated battery degradation. Exploring other charging schedules for electric vehicles such as Multistage Constant Current (MCC), researchers Liu et al. [80] explored the application of MCC for energy and cost optimisation purposes by finding current values where battery degradation is least severe.

Reviewing the current rates during the discharge process for eUAVs is less trivial due to the fact that the load current usage range of freedom is limited as this is determined by the mission profile [108]. For example, for take-off and landing a certain amount of power and thus current rate is required. During cruise, the velocity of the eUAV could be chosen such that the current rate is most optimal.

The DOD and average SOC relate to the amount of capacity available for flight. In order to reduce the DOD a battery undergoes, the distance the eUAV flies in between charging moments must be decreased. To achieve a lower average SOC, the battery should be charged to a SOC level below 100%. The challenge in this charging strategy is to be able to accurately predict the amount of charge required to complete a mission. This prediction is similar to EOD predictions, further described in subsection 3.4.3.

Though decreasing the DOD and avg SOC may elongate battery lifetime, it simultaneously limits the distance that the eUAV can cover. Nonetheless, several studies could provide solutions to this problem. Firstly, Bocewicz et al. [18] research the option of setting up a network of mobile battery swapping stations to optimise a routing problem with eUAVs travel paths. Another potential solution is proposed by Lu et al. [81], who review the use of powerlines to charge eUAVs during a mission through a wireless power transfer method. Similarly, Galkin et al. [45] research a combination of these innovative methods to ensure continuous eUAV operations. Their solution incorporates the option of swapping the eUAVs or their batteries, transferring power wirelessly or improving the battery energy density. Thirdly, optimal eUAV hub location problems are commonly addressed, such as in a research by Aurambout et al. [7]. In this study the authors review the economic viability of placing hubs in different locations, but such optimisation models could also incorporate other criteria such as sustainability. Finally, there are several studies such as a research by Goodchild and Toy [51] and Moshref-Javadi et al. [91] that evaluate the option of using UAVs in a last-mile delivery model with trucks, which would also allow for shorter flying distances.



Figure 3.5: Lithium-ion battery capacity retention behaviour related to different Depth-of-Discharge ranges during battery Dynamic Stress Testing (DST) [132]



Figure 3.6: Lithium-ion battery capacity loss due to different Depth-of-Discharge ranges, showing (a) maximum available capacity degradation, (b) nominal capacity degradation [47]

3.3.2.1. End-of-Life

The battery EOL is defined to be when SOH = 0% [12]. Operating the battery until 0% is however considered to be highly unsafe. Therefore, the useful life of a battery is defined to be until SOH = 70% or 80%, or when the internal resistance has doubled its initial value compared to at the beginning of life [39, 78, 129]. Operating the eUAV beyond this SOH point is dangerous, as the battery capacity may suddenly drop. The battery is therefore often considered as 'failed' when it reaches the critical SOH level between 80% [34, 78, 129].

3.4. Battery Health Management

Several studies have been done reviewing battery health management. Monitoring a battery's health is important to improve reliability and performance. eUAV battery health management aspects include determining the SOC and SOH, charging the rechargeable battery to ensure the system has sufficient charge to execute the next mission and ultimately replacing the battery when SOH < 80%.

First, the Battery Management System (BMS) that monitors the battery's health onboard most eUAVs is discussed in subsection 3.4.1. Then, different battery health management strategies are discussed in the following subsections 3.4.2 and 3.4.3. Finally, a general overview of battery health monitoring challenges is given in subsection 3.4.4.

3.4.1. Battery Management System

Most Lithium applications have a Battery Management System (BMS) to ensure that the battery remains within safe limits [23]. For example, the BMS will not let a battery charge beyond SOC = 100%. Depending on the battery usage application, the range of safe SOC levels may vary. Given the increased rate of battery degradation for higher SOC levels, it can be beneficial to set a lower maximum SOC level. For example, in a study with photovoltaic systems, the limits 20 and 80% are set as the minimum and maximum SOC levels, respectively [92].

The BMS monitors various parameters such as temperature, battery voltage and current and more [23, 75, 135]. The level of sophistication of a BMS varies. The basic features are providing the battery protection by ensuring operating limits are not exceeded, as well as determining the SOC. More sophisticated types of BMS are also able to show the SOH and make EOD or EOL predictions.

For eUAVs, size, weight and cost constraints are important drivers. Hence, researchers such as Sierra et al. [113] stress the need for efficient BMS computational resources while maintaining high accuracy. Moreover, despite recent BMS advancements, determining the battery SOH remains a challenge [23, 75]. This is due to the fact that the SOH is not a readily measurable output. Furthermore, extensive electrochemical models used to accurately determine SOH are often too computationally intensive to incorporate into a BMS. Finally, the most accurate SOC and SOH readings can be done when a battery is cycled from a fully charged to a fully discharged state. In reality, however, this is not always possible due to mission profile constraints and the SOC = 30% safety boundary for eUAV operations.

3.4.2. Scheduled Battery Health Management

The simplest form of battery health management is by using a 'scheduled' approach [23, 27, 113]. Here, within one discharge cycle, the eUAV's safe range of flying time is predefined before flight. Furthermore, the battery is regarded as unsafe to be used after a predetermined number of charge/discharge cycles. These estimations for flying time and useful lifetime cycles is based on past data.

When applying this battery health management approach, several undesirable traits arise [23, 27, 113]. Firstly, by using a fixed flying time, missions times are typically highly conservative, forcing the eUAV to stick to shorter missions. On the other hand, the other extreme would be that fatal accidents occur by disregarding the battery health that has potentially significantly degraded over time, implying that the available flying time has decreased with respect to the initial predetermined maximum available flight duration. Finally, for the SOH, similar problems come forward. The set number of battery cycles is either too safe or extremely dangerous. Currently, to avoid hazardous situations, engineers recognise that batteries are commonly replaced too soon [23].

3.4.3. Predictive Battery Health Management

Recognising the shortcomings of applying simple battery health management approaches to batteries, researchers are exploring predictive strategies. Researchers focus on various aspects within the topic of battery PHM, looking into topics such as EOD prediction and ageing mechanisms.

Estimation and prediction of SOC, EOD, SOH and EOL for eUAVs is important because these vehicles usually have cost constraints as well as weight and size restrictions [20, 113]. Prognostics play an important role in monitoring the battery's health and capacity [129]. Ultimately, accurate EOD estimation is the only protection against unexpected battery failure during flight [37]. When comparing battery monitoring to CM, the 'value' data type is typically considered, as parameters such as voltage and charge are commonly used to determine the battery charge and/or health status.

Methods to estimate the SOC include voltage models, coulomb counting and data-driven approaches [26, 129]. For SOH, similar techniques can be used including, for example, electrochemical, analytical, statistical models or data-driven mechanisms [12, 78]. For both SOC and SOH, determination approaches can also have a hybrid form combining for example model-based and data-driven techniques. These hybrid models often incorporate the benefits from both the physics and data-drive models, giving them optimal performance [26].

Mansouri et al. [85] state that RUL prediction for eUAVs is often done through data-driven methods as

physics-based approaches require a very high level of expertise. Batteries are non-linear electrochemical systems and behave dynamically over time, making them complex systems to comprehend.

Machine learning, regression, filtering and stochastic processes are often used as data-driven methods [85]. Two types of data-driven methods are recognised by Eleftheroglou et al. [37]. Most commonly, one overarching predictive model is built. The second method consists of building several simpler models and then combining these outputs.

Although battery PHM have come a long way, many researchers still recognise this as a subject for improvement. Accurately determining and predicting the SOC and SOH of a battery is extremely complex due to the influence of many different parameters such as operating environment, cycling profile and the non-linearity of electrochemical processes [26, 98]. For eUAVs, data-driven techniques are commonly chosen as electrochemical battery models are complex and require expert knowledge [29, 78]. Also, aviation electronics with eUAVs form a compounded integrated system, making it difficult to insert CM technologies. Finally, datadriven approaches are beneficial as they have the ability to include many different parameters such as voltage, temperature and current levels. They have a high computational efficiency and are flexible, implying that they are able to adjust to dynamic settings.

Currently, researchers recognise a **problem concerning SOC and SOH estimations**. If the SOC and SOH levels are not known with absolute certainty, accurate predictions for EOD and EOL for eUAVs cannot be made [27, 113, 135]. As a result, flight plans are decidedly conservative, implying that only those missions are executed that fall well within the estimated EOD range. Improving SOC predictions would be beneficial because this enables longer flight plans to be flown. With more accurate SOH predictions, the battery could be used for more cycles instead of being replaced too soon before reaching SOH = 80% which is regularly the case in reality [23].

For battery health management, several prognostic elements exist. Throughout charge and discharge cycles, the predictive system is required to [27, 78]:

- 1. accurately determine the actual charge (SOC) in the battery,
- 2. determine the required battery charge needed to safely complete a mission,
- 3. predict the battery's EOD,
- 4. forecast the battery ageing over time, and
- 5. estimate the battery's EOL.

Case Studies

As a remaining flying time example, Hogge et al. [61] verify their method that predicts the amount of charge a battery has left to perform a mission. Flying with SOC <30% was considered highly unsafe. Thus, the researchers conducted ground-based simulated testing to train the model. The flight time prognostics are firstly dependent on accurate online SOC estimations. Then, based on a predefined flight plan with set power loads, a future power demand prediction for the motor is done. To represent a real-life mission, parasitic battery loads are also accounted for in online estimations. Finally, the future flight plan battery discharge profile is predicted.

On the same topic, Saha et al. [107] present a PHM framework for Lithium eUAV batteries based on Bayesian learning techniques with the aim to predict the EOD within a cycle. They address several different flight operation regimes including take-off, cruise, turns and landing to accurately represent a mission. Throughout the mission, the PDF of the EOD is continuously updated. Based on these calculations it can successfully be determined if the eUAV is able to complete the flight safely.

Another example is given by Sierra et al. [113] who present a battery management system for rotatory-wing eUAVs that provides an accurate estimation of the SOC and EOD time of Lithium batteries. In this study, a prognostic framework based on models assisted by Bayesian methods is used. By evaluating the SOC and EOD parameters, decisions regarding the flight plan can effectively be made.

For capacity degradation and EOL estimation, Birkl et al. [17] look into the electrochemical properties of battery degradation and validate that Lithium battery capacity loss is linked to properties such as loss of lithium inventory and loss of active materials. The authors propose a model that is able to predict the trajectory of
changes of these properties within an acceptable margin. Incorporating such forecasting techniques enables users to keep track of the battery's SOH and, by doing so, ensure safe operation. In theory, their diagnostic tool is applicable for all types of Lithium batteries. Future work includes applying this real-time to commercial Lithium batteries.

When exploring research done on predictive battery health strategies that aim to maximise battery lifetime, little literature is found concerning eUAVs. For electric vehicles in the automotive industry, Valentina et al. [123] recognise the influence that charging has on Lithium battery degradation and stresses the importance of defining a methodology for users to promote battery lifetime elongation. On the same topic of electric vehicles, Abdullah et al. [3] acknowledge the impact that charging methods have on battery degradation and discover that introducing rest periods during battery charging reduces the degradation process.

3.4.4. Challenges

Throughout battery health management and prognostic studies, researchers define various challenges that decelerate the development and progress of these subjects. These oppositions are listed below.

• Noisy data

For all predictive battery health management methods, the accuracy and reliability of the prognostics are dependent on the input data. Raw data is generally noisy and requires manipulations before it can be used to identify trends and generate predictions [85, 129]. Filtering techniques are used to clean the data in order to use it for forecasting. It is important to continuously evaluate and further develop the applied filtering methods to improve the performance of the predictive model.

• Nominal capacity

To set up a useful predictive battery health management system, it is important to accurately determine the battery's SOC based on readily available performance metrics. Deriving the SOC from real-time input remains one of the toughest challenges in battery health management due to the intertwinement of many parameters and complexity of electrochemical compositions [26, 98, 129]. Prognostics for EOD and EOL cannot be done if the output of this fundamental step of determining SOC is not reliable.

• Flying plans

The rate of SOC decrease is highly dependent on the eUAV's flight plan. The power consumption differs depending on the flying manoeuvre, making EOD predictions a complex task [108, 113]. For example for eUAVs, the absolute level of power consumption is higher during landing than during cruise, as shown in Figure 3.3. Predefining the eUAV's flight plan is therefore very important, such that specific flight actions can be taken into consideration when estimating the EOD.

• Flying conditions

Anticipating the diverse range of varying flying conditions that an eUAV may encounter, remains a challenge for predicting the battery's EOD [12, 85, 108, 113]. As stated in section 3.3, the SOC decrease rate varies depending on the power demand which is influenced by a diverse set of parameters including the external temperature, wind and density altitude.

• Ageing

Battery ageing remains a problem, as this brings along uncertainty in battery PHM [12, 61, 132]. The rate of battery degradation is dependent on a large set of parameters, making it very complex to predict. Currently, studies mainly focus on battery capacity loss and internal resistance increase as ageing phenomena. In practice, ageing is related to many more events. Also, in order to benefit from degradation predictive models, future work is required to develop a real-time ageing estimator.

Cost-Benefit Analysis

Limited literature is found regarding Cost-Benefit Analyses (CBAs) of battery health management methods. Several overviews are given of advantages and disadvantages of different types of batteries [95], as well as comparative studies on various prognostic approaches [78]. However, no material is found that focuses on comparing two battery health management techniques for the same kind of battery. Quantifying battery health management costs is complex but essential in order to make justified charging strategy decisions.

3.5. Battery Data and Simulation Tools

To conduct a research in the field of batteries, it is important to verify that there is sufficient data and/or simulation tools available. A brief overview of potential sources is given in this section.

In a recent paper, don Reis et al. [34] listed all Lithium data sources. The most commonly used battery cycling data sets are:

1. NASA Prognostics Center of Excellence department

This NASA battery data set [2] is the first publicly available battery data set, showing cycling data for 34 Lithium 18650 cells and 28 LCO 18650 cells with a nominal capacity of 2.0 and 2.2 Ah, respectively. The battery cells undergo cycling for a range of different temperatures, charge and discharge regimes. The experiments are stopped until a SOH level is reached between 50-80%.

2. Center for Advanced Life Cycle Engineering (CALCE)

The CALCE data sets [42] feature LCO, LFP, and NMC chemistries of Lithium batteries with varying dimensions and different capacities (2.0, 2.23, 1.1, 1.35, 1.5). The data sets investigate different DODs, C-rates and ranges for partial charge/discharge until SOH = 80% is reached.

3. **Sandia National Laboratories (SNL)** Sandia National Laboratories (SNL) offer multiple 18650 cells with LFP, NCA and NMC compositions. The battery cells are cycled under different temperatures and DODs (0-100%, 20-80%, 40-60%). All batteries are charged according to the manufacturer guidelines at 0.5C, and discharged at different currents (0.5C, 1C, 2C, 3C) until EOL is reached at SOH = 80%.

Although there are a few other data sources available, these are not relevant for eUAV battery health management applications. For example, some data sets focus on calendar ageing, fast charging protocols or have very specific usage applications. The authors [34] recognise the lack of Lithium battery data and call for more open-source data sets to stimulate further battery research.

Instead of using data sets, it is also possible to use battery simulation tools. A benefit of using a simulation programme instead of a data set, is that it can be tailored to represent an eUAV application more accurately. Many simulation programmes, however, require expensive licenses and/or are difficult to integrate with programming languages such as Python and Matlab. To mitigate these shortcomings, a group of researchers launched the Python package 'PyBaMM' in 2019 [115]. PyBaMM is a tool to simulate battery models fast and flexibly. It can be used to solve models using readily available Lithium battery cell specifications or has the option to input new battery parameter sets. Batteries can be cycled through simple calling functions (for example, "Discharge at 1C until 3.0V"), or by means of a detailed excel input cycle data set. The package has an open-source collaboration platform on GitHub [103].

3.6. Research Gaps

In this chapter, fundamental understandings of Lithium batteries used in electrical aviation are listed. Deriving accurate battery characteristics such as SOC and SOH are crucial in order to enable safe flight. Concurrently, if the real-time SOC and SOH are accurately determined, batteries could be used for longer flights and more charge/discharge cycles, respectively.

Summarising the main points of interest for further analysis, it is clear there is a need for more accurate predictions of SOC, required charge, EOD, battery ageing and EOL. Addressing all these topics, however, would be out of scope for one thesis research. Hence, in the interest of sustainability, it is chosen to focus on a battery health management strategy that aims to elongate the battery lifetime by minimising the DOD and average SOC at which the battery is cycled, discussed in subsection 3.3.2. Applying this to eUAV applications has not been previously done in literature. Thus, this research could provide interesting insights for the aviation industry. Moreover, evaluating the other prognostic elements such are predicting SOC and SOH does not contribute to combatting battery degradation.

A more detailed description of the proposed thesis research topic regarding battery DOD and average SOC minimisation is presented in Part IIB of this report. Before providing a detailed overview of the thesis experimental setup, two additional topics related to prognostic models are discussed in the following chapters. In order to execute the research, battery health monitoring prognostics are required related to estimating the amount of charge required to complete a mission. Methods to evaluate the required charge prediction

method are discussed in chapter 4. Then, in 5, a method to compare the benefits of different battery health management strategies is presented.

This page intentionally left blank.

4

Performance Evaluation

Evaluation of the predictive technique's performance is essential to determine which methodology is most beneficial for a given component or system. In the thesis research proposed in Part IIB of this report, one/multiple of these metrics can be used to evaluate the performance of the model that predicts the amount of battery charge required to complete an eUAV mission, as briefly described in section 3.6.

Despite the fact that performance metrics vary depending on the type of system or applicable requirements, a survey done by Saxena et al. [109] showed that most metrics are related to the accuracy and precision of a method. In industries such as finance where reference prediction models are readily available, metrics such as Mean Square Error (MSE), Mean Absolute Deviation (MAD), Median Absolute Deviation (MAAD) and Mean Absolute Percentage Error (MAPE) are often used. These metrics can also be used to measure the accuracy of a model in the aerospace industry.

Within PdM and PHM, Saxena et al. [109] point out that there is a lack of standard definitions. In the authors' opinion, the methodologies are often ambiguous and inconsistent. Moreover, the shortcomings of model verification and validation are stressed. Establishing a set of performance metrics to evaluate the PdM and PHM approaches would enable straightforward assessment and comparison between models. These statements are also backed by Chang [26] who claims that setting up an accurate evaluation method and/or measurement of performance is required to determine which models best predict SOC trajectories and other battery characteristics.

Nine performance metrics are defined by Saxena et al. [110] to evaluate data-driven prognostic algorithms for maintenance in domains such as aerospace, automotive, electronics, nuclear and medicine. Performance metrics are required in order to evaluate the model to ultimately avoid the occurrence of False Positives (FP) and/or False Negatives (FN).

First, the difference between accuracy and precision is addressed in section 4.1. Then, a brief description is given about FP and FN in section 4.2. Subsequently, the evaluation metrics are presented in the following section 4.3. Finally, several challenges are addressed in section 4.4, as well as a brief conclusion of how performance evaluation metrics can be used in battery health management research.

The description of the terms and notations that are used throughout the sections in this chapter are listed in Table 4.1.

4.1. Accuracy and Precision

While the terms 'accuracy' and 'precision' are often used interchangeably, there is a fundamental difference between the two definitions. Accuracy refers to the average error distance between the predicted and actual RUL. High accuracy is defined when these errors are small. Precision, on the other hand, refers to the spread between the predicted RULs. Most desired are cases that have both a high accuracy and precision level [22, 56].

Name	Abbreviation
X	Number of units under testing
i_P	First time index <i>i</i> within set ℓ that a prediction is made
i_{EOP}	Last time index <i>i</i> within set ℓ that a prediction is made (EOP is End-of-Prediction)
i_{EOL}	Actual time index <i>i</i> that the component reaches its End-of-Life
l	Total time set of moments <i>i</i> at which predictions are made, $\ell = (i P \le i \le EOP) $
$\Delta^X(i)$	Error between the estimated and the true RUL at time point <i>i</i> for a given component for time set <i>l</i>
$r^X(i)$	Estimated RUL for the X th unit under testing at time point <i>i</i> using the prognostic algorithm
$r_*^X(i)$	True RUL for the X^{th} unit under testing at time point <i>i</i> provided that the data is available

Table 4.1: Overview terms and notations for prognostic data-driven algorithm evaluation metrics [110]

4.2. False Positives and False Negatives

A prediction that incorrectly gives an alert that failure is near, is called a False Positive (FP). This is considered an unacceptable early estimation if a predefined range r_{FP} is violated, as shown in Equation 4.1 [109]. On the other hand, False Negatives (FNs) are the result of late predictions, as defined in Equation 4.2 [109]. Here, the FN occurs when the critical range r_{FN} is exceeded.

$$FP = \begin{cases} 1 & \text{if } \Delta^X(i) > r_{FP} \\ 0 & \text{otherwise} \end{cases}$$
(4.1)

$$FN = \begin{cases} 1 & \text{if } -\Delta^X(i) > r_{FN} \\ 0 & \text{otherwise} \end{cases}$$
(4.2)

When evaluating an algorithm, FNs are typically penalised more severely as late predictions may result in fatal accidents.

4.3. Performance Metrics

Specifically, the evaluation standards focus on models estimating the RUL of a component. Ideally, evaluation metrics are unitless as this enables easy comparison of performance from different models.

The metrics are built upon findings in a previous survey done by Saxena et al. [109] regarding all existing evaluation metrics for prognostic use cases. After reviewing the metrics, nine methods are selected and proposed for common predictive algorithm evaluation. These evaluation standards are also used for the assessment of prognostic models made by other authors such as Hogge et al. [60], Busse et al. [22] and Saha et al. [108]. These metrics are discussed in subsection 4.3.1 to 4.3.9.

4.3.1. Average Bias

The average bias metric is one of the most important methods to determine the accuracy of a prediction. The average bias is calculated using Equation 4.3 [110], averaging the errors made per prediction at all times after the prediction is made for a total set of l time indices that an estimation of the RUL is made.

$$B_{\ell} = \frac{1}{\ell} \sum_{i=1}^{\ell} \Delta^{X}(i)$$
(4.3)

Using this metric, positive and negative errors cancel each other out. Also, the influence of outliers is not mitigated.

4.3.2. Sample Standard Deviation

To evaluate the precision of a prediction, the Sample Standard Deviation (SSD) is often used. This metric quantifies variability with respect to the sample mean. The SSD is determined with Equation 4.4 [110]. Note that SSD only applies to normal distributions. In this formula, *m* is the mean of the sample set of errors.

$$SSD = \sqrt{\frac{\sum_{i=1}^{\ell} (\Delta(i) - m)^2}{\ell - 1}}$$
(4.4)

4.3.3. Mean Squared Error

The Mean Squared Error (MSE) addresses both the accuracy and precision of a model. Unlike most evaluation metrics, MSE is not unitless. Thus, comparing two systems' MSEs must be done carefully. The MSE determines the average of the squared prediction errors and therefore takes both positive and negative errors into account. The metric is calculated using Equation 4.5 [110].

$$MSE = \frac{1}{\ell} \sum_{i=1}^{\ell} \Delta(i)^2$$
(4.5)

As a metric, MSE is sensitive to non-normal data and the presence of outliers. Moreover, MSE is unreliable for small samples of data.

The Root Mean Squared Error (RMSE) is a derivative of the MSE, which measures the accuracy of a function. The RMSE is calculated by taking the root function of MSE and is commonly used by researchers.

4.3.4. Mean Absolute Percentage Error

For prognostics, it is important to take the time at which the prediction is made, into account. Prediction errors made closer to a component's EOL are therefore often weighted heavier than those made at the beginning of a component's lifetime. The Mean Absolute Percentage Error (MAPE) weighs the prognostic errors with the RULs and then averages these absolute percentage errors. The formula for MAPE is given in Equation 4.6 [110].

$$MAPE = \frac{1}{\ell} \sum_{i=1}^{\ell} \left| \frac{100\Delta(i)}{r_{*}(i)} \right|$$
(4.6)

Using MAPE is only relevant for ratio-scaled data that have a meaningful zero. Furthermore, severe penalties are given when forecasts exceed the actual EOL compared to those that are less than the real EOL.

4.3.5. Prognostic Horizon

The Prognostic Horizon (PH) reflects on the difference between i_p and i_{EOP} , specifying a tolerable error bound (α). By using PH, predictions are reliable as it focuses on estimates that fall within specified limits close to the actual EOL. When comparing multiple data-driven models, the algorithm with a larger PH is desired. The PH is defined by Equation 4.7 [110].

$$PH = i_{EOP} - i_P, \quad \text{with } i = \min\left\{j \mid (j \in \ell) \land \left(r_*(1-\alpha) \le r^l(j) \le r_*(1+\alpha)\right)\right\}$$

$$(4.7)$$

For example, if α = 5%, then the PH will indicate when then the algorithm starts predicting the EOL within 5% confidence bound of the actual EOL.

4.3.6. $\alpha - \lambda$ **Performance**

Similar to PH, the $\alpha - \lambda$ performance metric can be used to determine if a prediction adheres to specified performance levels. For this metric, the horizon λ is defined as a percentage of the total RUL, measured from time point i_P . Moreover, the accuracy of the estimation is evaluated with the use of α which forms a cone around the actual RUL trajectory r_*^X , as can be seen in Figure 4.1. In this graph, $\alpha = 0.2$ implying that the metric reviews whether the prediction adheres to the 20% accuracy range. Equation 4.8 shows how predicted RUL, r^X , is evaluated. In this formula, $t_{\lambda} = i_P + \lambda(i_{EOP} - i_P)$.

$$[1 - \alpha] \cdot r_*(t) \le r^X(t_\lambda) \le [1 + \alpha] \cdot r_*(t)$$
(4.8)

4.3.7. Relative Accuracy

The Relative Accuracy (RA) metric is similar to $\alpha - \lambda$ performance. However, with RA the accuracy level is measured instead of determining if data points fall within an accuracy range. Algorithms with a high RA are desirable, as this implies that predictions are accurate. Using Equation 4.9 [110], the RA of a model is determined. Again, in this formula, $t_{\lambda} = i_P + \lambda(i_{EOP} - i_P)$.

$$RA_{\lambda} = 1 - \frac{\left|r_{*}\left(t_{\lambda}\right) - r^{X}\left(t_{\lambda}\right)\right|}{r_{*}\left(t_{\lambda}\right)}$$

$$\tag{4.9}$$



Figure 4.1: Schematic figure of the $\alpha - \lambda$ prognostic algorithm evaluation metric with $\alpha = 0.2$ [110]

4.3.8. Cumulative Relative Accuracy

If the RA of a model is evaluated multiple times, the results can be aggregated by using the Cumulative Relative Accuracy (CRA) metric. This parameter represents the normalised weighted sum of the RA measured at specific points of time. The CRA is calculated by using Equation 4.10 [110], in which w is the assigned weight. Usually, higher weights are assigned to predictions done closer to the EOL.

$$CRA_{\lambda} = \frac{1}{\ell} \sum_{i=1}^{\ell} w(r^{X}) RA_{\lambda}$$
(4.10)

4.3.9. Convergence

This metric is used to quantify if and how well the accuracy and/or precision of a model improves over time. In Figure 4.2, three example cases are depicted that have different convergence rates. The metric M(i) represents the performance of a system. For each case, the centroid is defined as (x_c, y_c) . Convergence is quantified using Equation 4.11 [110]. When the Euclidean distance between the graph's origin and the prognostic case curve's centroid is low, a high convergence is defined.

$$C_{M} = \sqrt{(x_{c} - t_{P})^{2} + y_{c}^{2}}$$

$$x_{c} = \frac{\frac{1}{2} \sum_{i=i_{P}}^{i_{EOP}} (t_{i+1}^{2} - t_{i}^{2}) M(i)}{\sum_{i=i_{P}}^{i_{EOP}} (t_{i+1} - t_{i}) M(i)}, \text{ and}$$

$$y_{c} = \frac{\frac{1}{2} \sum_{i=i_{P}}^{i_{EOP}} (t_{i+1} - t_{i}) M(i)^{2}}{\sum_{i=i_{P}}^{i_{EOP}} (t_{i+1} - t_{i}) M(i)}.$$
(4.11)

4.4. Conclusion

One of the main challenges for evaluating predictive models is the vast number of metrics that are used throughout literature. Aside from the nine metrics presented in section 4.3, there is an even more elaborate summary of existing evaluation methods presented in another study by Saxena et al. [109]. Deciding which metrics are most suitable, is a complex task. Each prognostic model is unique, making it difficult to choose one metric to evaluate and compare different systems.

When evaluating the nine metrics for prognostic algorithms discussed above, Saxena et al. [110] state that



Figure 4.2: Schematic figure of the convergence prognostic algorithm evaluation metric showing three cases that converge at different rates [110]

future work is required to investigate the possibility to define a single performance metric that covers all the predictive model's aspects. Currently, using a combined set of multiple evaluation metrics is recommended to properly evaluate and compare algorithms.

Unfortunately, similar to shortcomings mentioned in chapter 2 and 3, Saxena et al. [111] stress that lack of data delays the ability to further develop evaluation metrics suitable for all types of maintenance prognostics on a large scale.

To evaluate prognostic battery health management strategies, one or multiple of these performance metrics can be applied. Providing that the errors are normally distributed, the metrics SSD and MSE can be used to determine the precision and accuracy of the error, respectively. If this is not the case, other metrics must be applied.

This page intentionally left blank.

5

Cost-Benefit Analysis

A Cost-Benefit Analysis (CBA) is an effective method to evaluate and compare systems on metrics other than accuracy and precision. For maintenance, a CBA provides support for deciding which maintenance strategy is most beneficial for a specific system [54].

In order to determine the benefits of the battery charging strategy in which battery DOD and average SOC are minimised, described in section 3.6, a CBA can be executed. Through a CBA, the performance can be compared to the output of for example a simple battery charging strategy in which the battery is always charged to full charge.

Often, a CBA is immediately associated with financial costs. Although this is often the case, other metrics are also used such as the effectiveness of maintenance [54], the scientific performance of a model [109] or the social and/or environmental impact that a strategy has [49]. For the aerospace industry, the metrics for CBAs are still immature [110]. These CBA parameters are first discussed in section 5.1. Then, the second part of this chapter, section 5.2, discusses essential steps that need to be taken in order to successfully conduct a CBA. Note that both these sections can be used to conduct a general CBA for a wide variety of aviation related maintenance practices. To draw the relevant CBA aspects that can be used to analyse eUAV battery health management strategies, a concluding summary of this chapter is given in section 5.3.

5.1. Parameters

In this section, several CBA parameters are presented. First, monetary costs functions are elaborated on in subsection 5.1.1. Secondly, maintenance effectiveness is reviewed in subsection 5.1.2. Then, in subsection 5.1.3 the scientific performance of a maintenance strategy is discussed. Finally, subsection 5.1.4 and 5.1.5 focus on social and environmental CBA parameters.

5.1.1. Financial Costs

For prognostic maintenance strategies, several different types of economic factors can be used to conduct a CBA. The most evident main object is to minimise maintenance costs. However, other related objectives such as minimal delays, minimum repair time duration and optimal utilisation of resources can also be included [124]. It must be noted that economic factors are often difficult to demarcate as many maintenance operations are intertwined with other activities. Van den Bergh et al. [124] refer to a long list of parameters such as availability, flight scheduling, crew assignment and legal matters that have their own associated costs that may be affected by a specific maintenance activity.

First, high-level descriptions of maintenance costs for different strategies are given. Then, the most common methods to evaluate financial costs are discussed in the following subsections.

5.1.1.1. Maintenance Costs

Maintenance costs are complex to determine due to the many related factors. Kent and Murphy [67] define three methods to build simplified models that can be used to represent the financial costs: an analogy, parametric, and engineering estimation method.

• Analogy

Estimation of cost based on actual and historical data. The data may concern information from a similar existing or alternative recommended technique.

• Parametric

Approximation using 'cost estimating relationships' that are either statistical or mathematical formulas relating to one or multiple of the system characteristics.

• Engineering

Technique in which a detailed cost breakdown is made. The costs for the individual components are then combined to form an engineering estimate.

Compared to aircraft, MEA and eAVs have a reduced cost [125, 127]. This cost reduction is related to a decrease in operation, maintenance and fuel costs. Accurate saving estimates have not yet been made due to the uncertainty of battery technologies.

Unmanned Aerial Vehicles (UAVs), either battery or fuel-powered, also have a significantly lower overall cost [8]. Although operating costs are generally substantially lower (approximately 40% of an aircraft's operational cost), not all UAVs costs are less. For example, UAVs require surveillance and radio communication systems that may be even more extensive than a manned aircraft. Additionally, a control station may be required that can be either ground, air or sea-based. Secondly, although UAVs are smaller and weigh significantly less, their structure can consist of a similar number of components that come with comparable manufacturing costs. UAV maintenance costs are estimated to equal 20% of a manned aircraft cost.

Corrective Maintenance

The costs of a failure consists of the component repair costs and the downtime costs [15]. The actual repair costs are relatively easy to compute as this usually consists of quantifiable expenses such as material, equipment, workplace and labour costs. Downtime, however, is more complicated as this is related to the system's business value and any potential effects it has on other parameters such as delay and customer satisfaction. Most models assume there is no fixed cost for corrective maintenance this method is already being applied in practice and does not require any additional investments.

Scheduled Maintenance

Costs for PM consist of two parts [15]. Firstly, maintenance costs apply for maintenance carried out on a periodic basis. Secondly, failure costs (equal to costs for corrective maintenance) are induced in the occurrence of an unexpected failure. The probability of an unexpected failure arising in between scheduled maintenance checks and the costs that are paired with such an event, can be simulated using for example a Monte Carlo approach. Similar to corrective maintenance, authors assumed that no fixed costs are associated with PM.

Predictive Maintenance

For PdM, investment and operating costs are required [15, 40, 130]. As an investment, costs are made during the development, verification, and validation phase of the maintenance model. These costs include labour fees, but also all expenses related to software and administrative costs. Moreover, all additional hardware costs must be considered, including new sensor technologies, electronics, and testing facilities implied expenses. During operation, the costs consist of performing the prognostic analyses, maintaining the system, and physically repairing the system.

Costs for FP and FN must also be considered [15, 40]. For an FP, when a near failure is wrongly alerted, expenses consist of system downtime and any additional inspection cost factors. If a failure alert is missed due to an FN, failure costs equal to corrective maintenance are included.

5.1.1.2. Evaluation Methods

To review the financial costs related to a maintenance strategy, various metrics exist. Below, the parameters return on investment, payback period and net present value are detailed.

Return on Investment

For PdM and other prognostic maintenance strategies, an investment is required before being able to physically apply and use the maintenance method. The Return on Investment (ROI) is most commonly used to evaluate the financial benefit of an investment and is calculated using Equation 5.1 [40]. Here, the central fraction is the classical finance ROI definition, while the right ratio is the ROI applied for PHM assessment equation.

$$ROI = \frac{Return - Investment}{Investment} = \frac{Avoided\ Cost}{Investment} - 1$$
(5.1)

If ROI is larger than 0, there is a cost-benefit [40, 130]. Though in some cases, a ROI that is less than 0 may still be beneficial, if for example availability of the system is a lot higher and if this factor has not been included in the cost calculations.

In a study for NASA, Kent and Murphy [67] claimed that implementing prognostics for aircraft maintenance of structures results in a ROI of 0.58 within 3 years time, assuming that there is a 35% decrease in maintenance requirements.

Typically, businesses calculate different cost scenarios for conducting maintenance and then apply the setting for which the highest ROI is achieved [40]. For example, in a study done by Banks and Merenich [9], sensitivity analyses showed that the highest ROI was achieved for prognostics incorporating the longest time horizon.

Payback Period

Another method used to evaluate finances, is the Payback Period. This metric assesses the risk associated with the maintenance strategy as it indicates the time period it takes until the return is equal to the investment costs (ROI = 0) [130].

To determine the Payback Period expressed in months for a PHM strategy, Equation 5.2 [130] can be used. Here, *T* is the application time of the PHM approach.

$$Payback \ Period = \frac{Investment}{Monthly \ Benefit} = \frac{12T}{ROI + 1}$$
(5.2)

Net Present Value

The Net Present Value (NPV) is another parameter regularly used to evaluate economic factors as it indicates the value that the investment of implementing prognostics adds to the investor. The NPV is calculated using Equation 5.3 [102]. In this formula, *t* refers to the time during the total time period *T* that has passed. Over time, a positive Δ NPV (NPV > 0) indicates that there is a financial benefit [102].

$$NPV = Initial Costs + \sum_{t=1}^{T} \frac{Cash Flow at t}{(1 + Discount Rate)^{t}}$$
(5.3)

5.1.2. Effectiveness

Instead of calculating economic costs, some studies focus on the difference in the effectiveness of maintenance strategies. As stated previously, a negative ROI can still imply that the maintenance strategy is more desirable if, for example, the system effectiveness is higher.

According to Pecht and Rafanelli [99], effectiveness is determined by reviewing the availability, dependability and capability of a system. Availability refers to the probability that the system is able to operate at the beginning of the usage period. Dependability is related to the probability that the system does not fail during operation. Finally, capability measures the product performance. The effectiveness is calculated through Equation 5.4 [99].

$$Effectiveness = Availability \cdot Dependability \cdot Capability$$
(5.4)

Another commonly used term for 'dependability', is 'reliability'. For maintenance, to measure the reliability of a system, metrics like Mean Time Between Failure and its ratio to metric Mean Time Between Unit Replacements (MTBF/MTBUR) are applied [110].



Figure 5.1: Analysis of value for different maintenance strategy levels [53]

For Lithium batteries, Gandoman et al. [46] research the characterisation of failure models for Lithium batteries for electric vehicles and identify that understanding these modes is important for reliability assessment. According to their study, the loss of active anode materials contributes most to the reliability of the battery. Williard et al. [129] go about a similar approach and highlight that the chemical composition of Lithium batteries is crucial when designing the power source for reliability. In addition to assessing reliability during the design stage, the operation stage is reviewed. Here, the battery capacity is most commonly used to evaluate reliability and performance.

Other parameters may also be used to indicate the system effectiveness. Haddad et al. [54] refer to system effectiveness as maintenance value that is determined by parameters such as downtime, unpredictability and occurrence of catastrophic failures.

Inspired by their work, Guillén et al. [53] created a graph showing the maintenance value as a function of time, which is depicted in Figure 5.1. The chart also reflects the system health along the same timeline. In the graph, the maintenance types scheduled maintenance (TBM), CBM, CBM/PHM combined and corrective maintenance are marked. Here, unscheduled maintenance is defined to be the same as corrective maintenance.

The graph clearly shows that the highest maintenance value is achieved by using the CBM/PHM strategy. Furthermore the difference between scheduled and unscheduled maintenance is negligibly small as although fatal accidents are more likely for the unscheduled approach, scheduled maintenance faces the problem of high downtimes and early replacements (large unused RUL).

5.1.3. Scientific Performance

Another method to evaluate and compare maintenance strategies is by reviewing their accuracy and precision [109]. Methodologies to do this have been discussed in chapter 4. Note that these metrics are designed to evaluate prognostic algorithms are and therefore not useful to compare two different types of maintenance strategies. For example, these metrics could be used to determine which PdM algorithm performs best, but not to determine whether a corrective maintenance or PdM strategy should be applied. Busse et al. [22] state that cost savings depend on the accuracy and precision of a prognostic model. For models with higher accuracy and precision, a larger cost reduction is observed. This can be assigned to the fact that low prognostic accuracy and precision result in maintenance that is either carried to early or too late.

5.1.4. Social Impact

Another metric that can be used to select the most beneficial maintenance strategy, is social impact [49]. This is not a common parameter to include, though it is still very relevant to take into consideration. Social measures that could be evaluated are for example personnel safety, health, and wage, as well as general employment issues, government regulations and stakeholder participation.

Setting up a method to compare these social impact factors is not straightforward. Ghazi et al. [49] propose an approach with fuzzy rules by translating modes from *very low* to *very high* to a numerical scale that ranges from 0 to 1. The authors then determine which leading factors must be taken into consideration for the CBA, after which a weighted trade-off is conducted.

5.1.5. Environmental Impact

An upcoming factor to do take into account when conducting a CBA, addresses the sustainability aspect of the problem. Authors such as Franciosi et al. [44] and Ghazi et al. [49] state that current maintenance decision methods are solely focused on conventional parameters related to economical or technical features. The authors focus on general industrial and manufacturing industries, not just specifically the aerospace domain. With the current shift towards technologies that have a smaller environmental impact, more research on this topic related to maintenance is required.

The term 'green maintenance' is used in literature [4, 43, 44] to represent maintenance strategies that aim to minimise the negative impact on the environment. According to Iung [64], the integration of sustainability in maintenance is advocated by the upcoming trend in implementing PHM methodologies. By looking at the system as a whole instead of only focusing on one small component, prognostics are shifting from only reviewing the RUL to other matters such as energy consumption and/or efficiency. Here, it is essential that maintenance is not seen as an aftermarket service but rather an imminent and continuous process to enhance the overall system performance.

Environmental factors that can be considered when choosing a maintenance strategy are [43, 49, 64]:

- Energy consumption
- Pollutive emissions (air, water and soil)
- Noise emissions

- Material resources
- Material longevity and waste upon disposal
- Environment management

Similar to subsection 5.1.4, comparing different types of sustainability parameters within a CBA is complex, as different factors can be weighted differently depending on the stakeholders' preference. It is important to determine a CBA method beforehand to avoid confusion.

5.2. Experimental Setup

In order to determine which maintenance strategy is most beneficial, an analysis is performed. Before conducting the CBA, several steps need to be taken.

The experimental setup parts related to a CBA are discussed in the subsections below. Firstly, lifecycle cost data types are elaborated on in subsection 5.2.1. Subsequently, different system modelling analysis approaches are presented in subsection 5.2.2. Then, CBA parameter choice and trade-off methodologies are discussed in subsection 5.2.3. Finally, subsection 5.2.4 touches upon common assumptions applied for maintenance strategy CBAs.

5.2.1. Data

In order to determine the lifecycle cost of a system, Kent and Murphy [67] recognise three types of data categories: technical, programmatic and cost data. These types are briefly explained below.

• Technical

This data type includes the specifications for engineering, operational attributes and performance

characteristics. Here, the reliability and maintainability of the system are mainly focused on.

Programmatic

Programmatic data consist of the facts and/or assumptions concerning the system. These data points are related to several activities including the utilisation of the system as well as the logistics and support concepts.

• Cost

Items such as labour, equipment and material costs are categorised under cost data. This data type includes the facts and/or assumptions regarding the monetary value of the resources and/or consumption requirements.

5.2.2. Analysis

In order to conduct a CBA, separate systems following a specific maintenance strategy are modelled. The following subsections describe typical analysis methods to model systems.

5.2.2.1. Scenario Analysis

The simplest way to model a system is by using a scenario analysis [124]. Scenario analyses can be used for exploratory or decision orientated research based on qualitative and/or quantitative information [122]. Scenario cases can also be used to perform sensitivity analyses, to investigate how a model responds to different scenario inputs [21]. Often, these models are simple and not computational heavy. The downside is that scenario analyses are often less suitable to evaluate stochastic elements and perform detailed analysis on relationships between events.

5.2.2.2. Discrete Event Simulation

As the title of this simulation suggests, Discrete Event Simulation (DES) models operations as a discrete set of time points. DES can be used for several types of models including determination of the maintenance cost, availability, reliability and scheduling [5]. Usually, DES represent deterministic models. A DES model requires several inputs including an initial state, a time variable to order the events in chronological order, the types of events that may occur and the changes required to yield the different statuses [16].

For example, Hölzel et al. [62] used DES to conduct a CBA on lifecycle costs for aircraft using CBM and PHM maintenance strategies versus conventional TBM approaches.

5.2.2.3. Monte Carlo Simulation

Monte Carlo (MC) simulations are similar to DES, but instead of representing deterministic systems, MC simulations are mostly used to model uncertainty. In an MC simulation, several different events can be statistically modelled, each with their own Gaussian distribution [54]. Random samples are generated to analyse all possible scenarios [50]. Hence, MC is often applied to analyse the range of all possible outcomes for a given maintenance strategy.

In a study done by Wu et al. [130], MC simulation is used to model a system in order to perform a CBA while covering all uncertainties that may occur.

For maintenance modelling, MC simulation techniques could be used to model the uncertainties such as unexpected delays, varying maintenance repair rates and the availability of equipment or workforce personnel [124].

5.2.3. Cost-Benefit Model

In order to conduct a CBA, the relevant cost parameters along with a trade-off method need to be selected.

From the CBA parameters presented in section 5.1, a selection can be made based on the interest of the stakeholders. In most cases, a financial factor is included to minimise monetary costs and (if applicable) determine if the investment is worthy.

With the current shift towards more sustainable practices, including a review of the environmental impact linked to specific maintenance practice is recommended [44, 49]. For example, for maintenance of eUAV batteries, multiple sustainability parameters such as energy consumption and waste could be included.

As briefly touched upon in subsection 5.1.4 and 5.1.5, comparing different types of CBA parameters is not straightforward. Within for example parameter types 'social and environmental impact', it is challenging to compare different cost factors as different weights can be applied to reflect the importance of the parameter. The CBA process is further complicated when additional cost parameters such as financial cost and maintenance effectiveness are included.

Typically, a trade-off method is used to reach an ultimate number to determine an optimal strategy. Tradeoffs are essential in order to justify the choice of a preferred strategy in an objective manner. Although the outcome of the trade-off is important, Daniels et al. [30] state that the documentation of the trade-off is the most critical part, including the relevant CBA parameters, scoring functions and input values. In their study, the authors provide several standard trade-ff methods. Figures of merit are used to represent specific quantifiable items that are of interest to determine if and how well a strategy satisfies the stakeholder's requirements. Weights are often included to classify the importance of the parameters that are being evaluated.

Finally, a sensitivity analysis can be performed to analyse the contribution of each CBA parameter [30, 57]. Pareto's distribution shows that in most cases, 80% of the strategy's performance can be traced down to 20% of the parameters. A Pareto analysis can also be used to review trade-offs for multi-objective optimisation problems. Optimal Pareto solutions provide the best solution for all cost functions related to a CBA.

5.2.4. Assumptions

Several researchers state relevant assumptions when evaluating the cost-benefit results of different maintenance techniques. A selection of these assumptions is briefly discussed below.

In a CBA study comparing sensor based with scheduled maintenance for aircraft done by Dong and Kim [33], an assumption is made that sensors do not need to be replaced throughout the aircraft's lifetime. Furthermore, the authors assume that the sensors are error-free. Lastly, the use of real-time CM data is critically reviewed. Although using such a data set is thorough, it brings along many complications in reality. Using real-time data is impractical from a wiring perspective. Prognostic calculations would be done from a ground-based station, thus wireless sensors would be required. However, and the use of wireless sensors comes at a cost as the sensors' weight is higher. Secondly, the use of real-time data is undesirable as processing all the sensor data is time-consuming.

Another crucial assumption that is often implemented in a CBA, is the assumption that a maintenance prognosis is always correct. Haddad et al. [54] incorporate such an assumption in choosing the best PHM approach by elongating the operation time past the RUL prediction. Here, the benefit of an extended RUL is compared to increased downtimes if the component is repaired at a sooner moment in time. Recognising that PHM prognostics are not perfect, the authors assume an error rate equal to 0.05 implying that there is a 5% probability that the PHM provides a wrong prediction of RUL.

For maintenance, the assumption that a component is 'as good as new' after having been repaired is also sometimes applied. Gilabert et al. [50] include such an assumption when a replacement or refurbishment maintenance activity is performed as soon as a component is nearing its EOL. But researchers can also choose to assume this after regular maintenance has been carried out.

For Lithium batteries, assuming that the battery charge is 'as good as new' after charging, would imply that the SOH of the battery is assumed to remain constant. As the SOH degrades significantly over time, assuming the battery that has run a large number of charge/discharge cycles returns to a restored level of perfection equal to the level at BOL could lead to fatal accidents and is therefore not recommended.

Finally, conducting a CBA for maintenance strategies addressing a single aircraft or UAV will result in different insights compared to models accounting for an entire fleet. For a conventional aircraft fleet model, extra costs for, for example, flight scheduling, maintenance routing and crew assignment are included when the total network of multiple systems is reviewed as opposed to an individual aircraft. Assumptions regarding the fleet size must therefore be stated beforehand. Furthermore, sensitivity analyses can be performed to evaluate different scenarios [124].

5.3. Conclusion

In this section, the CBA parameters related to battery health management strategies are first discussed, after a high-level overview of a relevant experimental setup is given.

Parameters

For a CBA on maintenance strategies for eUAV batteries during which different charging approaches are compared, the parameters 'financial costs', 'effectiveness' and 'environmental impact' are most relevant [45, 77, 80, 114]. When assessing battery health management strategies, it is important to account for the efficiency of transferring power from the power plant to the grid, to the battery and finally, to the eUAV propulsion system. Goodchild and Toy [51] find that this total effiency factor is 0.79. Similarly, Stolaroff et al. [114] state that this efficiency is 0.77.

For **financial costs**, the total costs consist of the BOL battery costs, the operation and maintenance costs and the EOL costs. These three costs are briefly discussed below.

1. Beginning-of-Life costs

For batteries, the BOL costs are usually determined either by analysing off-the-shelf prices or by calculating the costs based on average price per *kWh* data [77].

2. Operation and maintenance costs

For simple charging strategies, operation and maintenance costs of battery systems mainly consist of electricity costs used to charge the battery [77]. These can also be determined by using open-source price per kWh data, which changes over time and is country-specific in Europe [38]. A detailed break-down between peak and off-peak electricity prices can be used to distinguish charging activities that occur during the day or at night [80]. When including prognostics, costs described in subsection 5.1.1 apply.

3. End-of-Life

Recently, there has been an increase in papers addressing battery recycling and second-life practices [63, 71, 88]. Though the total amount of batteries that are being recycled or reused is increasing, this is not yet standard practice. Furthermore, though waste is reduced, recycling techniques often bring along the use or emissions of other unsustainable materials [63, 106]. Lithium EOL battery costs are complex to determine due to the inconsistency in data and dependency on many factors such as battery specifications, usage application and location, and can hence be disregarded.

Depending on the interest of the stakeholders it can be chosen to simply compare the total sum of the financial costs, or to apply evaluation methods such as the ROI. For batteries specifically, however, it may be difficult to accurately determine the investment costs for prognostics as this is not yet a standard practice.

The **effectiveness** of a battery health management strategy can be measured through the formula presented in Equation 5.4. For simplicity, the dependability and capability of the battery can be neglected, as for eUAV applications the battery SOC will always be kept above 30%, and the SOH above 80%, as described in chapter 3. The 'availability' of the battery, however, may vary depending on the applied charging strategy. Galkin et al. [45] use the downtime per UAV to determine how efficient each model is.

For **environmental impact**, the parameters 'energy consumption', 'material longevity and waste' are most relevant. In a study assessing the environmental impact of a UAV package delivery model, Koiwanit [70] found that of the total Global Warming Potential (GWP), 80% consisted of carbon emissions. Thus, for the battery life cycle factors 'energy consumption' and 'material longevity and waste', carbon emissions will closely be evaluated.

Firstly, the energy consumed during the charging cycles of the battery can continuously be monitored and differs between charging methods. Emissions per kWh in the Netherlands, for example, are a mixture of emissions derived from sustainable 'green' and 'grey' electricity [25], on average resulting in 0.49 kg carbon emissions per kWh consumed electricity.

Secondly, material longevity refers to the battery lifetime that is affected by the applied charging strategy. Emissions arising from this parameter are related to BOL and EOL activities. For BOL, emissions are usually expressed in kg or carbon per kWh battery. On average, BOL production GWP emissions account for 99%

of the total battery life cycle emissions [70]. Data for Lithium battery production emissions vary between 35 and 250 kg per kWh [55, 82, 106]. For EOL, as mentioned above, recycling and reuse practices are not yet standard. Hence, recycling and reuse applications can be disregarded from this study.

Experimental Setup

For the experimental setup of the CBA, the data can be retrieved from example studies or open-source data centres. Some exemplary sample data has been presented above per parameter 'financial costs', 'effective-ness' and 'environmental impact'. For the analysis, an MC simulation can be run to test how the battery health management strategies respond to uncertainty. Throughout the CBA, it is essential to state all assumptions and acknowledge their impact on the CBA results. Due to the fact that no standard CBA framework for battery health management approaches exists, it can be chosen to retrieve a single final CBA score through the use of a trade-off method, a sensitivity analysis and/or review of the model's Pareto efficiency.

This page intentionally left blank.

PART IIB

This page intentionally left blank.

6

Problem Statement

Based on the literature gaps discussed in Part IIA of this Literature Study, a Cost-Benefit Analysis (CBA) comparing several battery health management strategies for electric Unmanned Aerial Vehicles (eUAVs) Lithium batteries is proposed. First, a summary of the research gaps found in Part IIA of this report is presented in section 6.1. Secondly, the project outline of the proposed thesis research is elaborated on in section 6.2. Finally, section 6.3 lists the research questions of the cost-benefit analysis.

6.1. Research Gaps

Throughout this Literature Study, several shortcomings in available maintenance and battery health management research studies have been addressed. These research gaps are shortly summarised below.

1. Battery Health Management

Reviewing maintenance studies, it can be concluded that there is a significant difference in the number of research papers regarding conventionally fuelled aircraft compared to electrical Aerial Vehicles (eAVs) health monitoring strategies. With the current shift towards more sustainable alternatives, there is an increase in eAV applications. Thus, it is beneficial to further investigate and improve eAV health monitoring approaches.

Given the main distinction from fuelled aircraft and the critical role of a eAV's power source, it is chosen to focus on health management methodologies for batteries. As batteries for large eAVs are not yet widely available, smaller electric Unmanned Aerial Vehicles (eUAV) battery applications will be considered.

2. Prognostics

Within the aerospace industry, Prognostic & Health Management (PHM) methodologies are increasingly being applied. Also in the field of batteries, prognostic practices are progressively being developed. For both general maintenance and battery health management strategies, however, the prognostic part requires further development. Authors call for an improvement in consistency within prognostics, more data acquisition, and an enhancement in performance for estimations and prediction methods and stress the importance of verifying and validating theories.

3. Sustainability

Although studies have proven that the average State of Charge (SOC) and Depth of Discharge (DOD) levels influence battery ageing processes, little research has been done applying this knowledge to eUAV battery models. Especially in the light of developing more sustainable battery health management methods, developing and evaluating charging schedules that aim to elongate battery lifetime, is of interest.

4. Cost-Benefit Analysis

In order to determine which battery health management strategy is optimal for a given system, a Cost-Benefit Analysis (CBA) can be conducted. Currently, there is no set framework to review and compare different types of battery health management strategies to determine which method is optimal when considering CBA parameters such as financial costs, effectiveness and sustainability.

6.2. Research Outline

The main research aim is to **review battery health management methods for eUAV batteries**, particularly focusing on a PHM approach. In the thesis research, a **simple battery charging approach** will be compared to a **more advanced prognostic strategy**. The performance of the methods will ultimately be evaluated by means of an CBA.

To evaluate both charging approaches in a fair manner, identical lists of missions target locations will be generated and executed in the same order by both the simple and the prognostic battery health management model. A more detailed description of these two strategies is given below.

In the first **simple SOC 100%** strategy, the eUAV battery always starts from 100% SOC level. For this model, battery degradation is expected to be most severe due to the high average SOC level. In contrast, the second **prognostic varying SOC** battery health management approach aims to decrease the rate of degradation. This is achieved using a predictive approach to determine the minimum SOC level that the battery needs to complete the next mission. By charging the battery to a lower SOC, the battery is protected from severe battery degradation. The required charge predictions are based on previous flying data using for example a regression or machine learning technique.

For both approaches, the predefined maximum flight time is chosen such that the battery always operates within SOC safety limits by maintaining a SOC level between 30 and 100%. Furthermore, capacity degradation is chosen as the indication parameter for the battery's State of Health (SOH). A battery is always replaced when its SOH drops below 80%. This is monitored by the Battery Management System (BMS).

The two proposed battery health management methods are briefly summarised below. These steps are repeated until the set list of missions has been complete.

1. Simple SOC 100% charging model

- Predefined maximum flight time which is never exceeded.
- The battery is charged to 100% SOC before the next mission.
- The battery is replaced when SOH < 80%.

2. Prognostic varying SOC charging model

- Predefined maximum flight time which is never exceeded.
- The required battery charge is predicted for the next mission based on the known target location.
- Before the next mission, the current battery SOC is evaluated.
- Then...
 - i. if the battery has sufficient charge to complete the next mission, the battery is directly used for the next flight, or,
 - ii. if the battery has insufficient charge to complete the next mission, the battery is charged until the predicted required SOC level before the next mission.
- The battery is replaced when SOH < 80%.

The scope of the research will focus on one eUAV system that must execute all the missions. By doing so, the impact of charging strategies can directly be linked to the degradation of the battery. It can be assumed that the number of available batteries for replacement is unlimited. To review the effect of different DOD on battery ageing, the two battery charging strategies will be tested for several mission ranges (short, medium, long). This is further elaborated on in chapter 7.

Additionally, case studies where the battery health management approach for example charges the battery before each flight to a lower maximum SOC level equal to 80% can be performed to review the sensitivity of the simple and prognostic charging strategies. Ultimately, the battery health management methods will be evaluated on several performance characteristics, including financial costs, effectiveness and sustainability.

6.3. Research Questions

The main research question is:

What is the sustainability and cost-benefit of simple versus prognostic battery health management strategies for Lithium batteries for electric unmanned aerial vehicles?

The following sub-questions are defined to conduct the research:

- 1. What is the sustainability and cost-benefit of using the simple SOC 100% battery health management charging strategy?
 - i. After how many cycles is the Lithium battery replaced?
- 2. What is the sustainability and cost-benefit of using the prognostic varying SOC battery health management charging strategy?
 - i. What is the performance predictive required SOC level model?
 - ii. After how many cycles is the Lithium battery replaced?
- 3. What are the optimal settings to minimise battery ageing for Lithium batteries?
 - i. What is the optimal average State of Charge level during operation?
 - ii. What is the optimal Depth of Discharge level during operation?

This page intentionally left blank.

7

Methodology and Project Timeline

In this chapter, a high-level outline of the project methodology is presented in section 7.1. Then, in section 7.2 these methodology phases are broken down into smaller tasks which are shown in a Gantt Chart.

7.1. Methodology

To build the battery health management models, the first part of the research consists of the approach preparations to develop the model, discussed in subsection 7.1.1. Then, the actual establishment of the model is addressed in subsection 7.1.2. Lastly, the final phase of the project is defined in subsection 7.1.3 during which the results are analysed, validated and conclusions are drawn.

7.1.1. Model Preparation

In order to develop the battery health management models for the Lithium eUAV batteries, available literature, data sets, simulation tools and open-source codes are reviewed. For the prognostic model, a data-driven approach is chosen due to the complexity of using an electrochemical physics of failure based model. Furthermore, data-driven techniques are desirable due to their computational efficiency and flexibility.

To simulate a battery's SOC and SOH, literature and data are analysed. A Lithium battery data set or simulation tool is chosen to simulate the battery decrease in charge during flight and degradation over time.

To build the simple and prognostic method, typical battery health management strategies for Lithium batteries are analysed. Literature and data supporting maximum flying time and useful lifetime values are summarised to determine predefined battery standards. For the PHM approach, open-source predictive algorithms are reviewed to gain a deep understanding of applicable methods.

7.1.2. Model Development

In this part, the two battery health management strategies, presented in section 6.2, are simulated. The model is developed in Python programming language.

The simple strategy is developed with the use of the literature findings regarding maximum flying time and lifetime cycles are translated into a simulation model.

For the PHM methodology, the predictive algorithms to predict the amount of charge required to fly a specific distance. In this way, the average SOC is reduced. The prognostic model can finally be evaluated using precision and accuracy performance metric such as Mean Squared Error (MSE) and Standard Deviation (SSD).

A set of three different mission areas are defined, in order to compare different battery DOD ranges. The maximum range within the largest mission area is set to be equal to the distance that can be travelled with Δ SOC = 70%, such that a battery is always able to deliver sufficient charge to go from the hub to the target and back to the hub for charging. This 70% SOC target value is defined by the minimum SOC threshold of 30% required to perform two additional landing attempts if required. Flight profiles are stochastically generated randomly anywhere within either the short, medium or long-range to represent a diverse set of missions.

Moreover, a set of 'mixed' missions of targets located in all three ranges can be generated. By differentiating these mission distances, the battery's DOD is intrinsically varied. Hence, the DOD's effect on battery ageing can be evaluated.

In order to compare the strategies, the sustainability and cost-benefit analysis model is prepared. Then, the two battery health management approaches are finally applied to the large set of mission profiles to evaluate their performance in the final phase of the project.

7.1.3. Results Analysis, Validation and Conclusion

In this last phase, the results of applying either the simple or PHM battery charge strategy to a statistically simulated set of mission profiles are analysed. To determine if the outcome of the models can be considered correct, the data is validated. Finally, conclusions for the sustainability and cost-benefit analysis can be drawn.

7.2. Project Timeline

Translating the research methodology into a project timeline, a total of six phases are defined. These parts are briefly elaborated on below in subsection 7.2.1 to 7.2.3. A Gantt chart depicting the timeline schedule of these six phases is shown in Figure 7.1.

7.2.1. Phase 1

This part of the project consists of the **model preparation**, discussed in subsection 7.1.1. In this first phase, the following project parts are completed within the first three weeks:

- Exploring battery data and simulation tool sources.
- Deciding on a battery data source or simulation programme.
- Simulating the battery.

7.2.2. Phase 2 and 3

Phases 2 and 3 of the project represents the **model development**, presented in subsection 7.1.2. This section runs for approximately 10 weeks, during which the following parts are carried out:

- Simulating the battery SOC and SOH.
- Defining of prognostic algorithms for the required amount of battery charge.
- Setting up the simple and prognostic battery health management strategy simulation models.
- Generating stochastic mission profiles.
- · Constructing sustainability and cost-benefit analysis trade-off parameters and methodology.

7.2.3. Phase 4, 5 and 6

The final project phases combined have the longest duration of roughly 16 weeks. This part includes the **results analysis, validation and conclusion**, addressed in subsection 7.1.3, as well as the documentation of the midterm and final thesis report. Furthermore, the midterm presentation, green light and final thesis presentation take place in this last section. To summarise, the following topics are covered:

- · Simulating the battery health management strategies and analysing and validating the results
- Drawing conclusions from the sustainability and cost-benefit analysis.
- Performing sensitivity analyses.
- Drafting the midterm and ultimately the final thesis report.
- Presenting methodology and findings during a midterm and final thesis presentation.

CANTT.	Y		21 Kickef					Midterm presentat	ion	2022	Ľ	hesis defense	
mhul		May		June	July August	September	October	November	December	January	February	March 2	April
Kirk-off	12/05/2021	17/05/2021	🔶 Kick										
1 Dettern deter sections	+CUC/JUL	scucioular			to a start sources								
 B Evolute data contract 	17/05/20121	1002/50/90		Evolore data									
· O Choose data source	1000/50/20	28/05/2021		Choose data									
 Implement open-source data into Pvth 	29/05/2021	06/06/2021		Impler	ment open-source data into Python moo	a la							
2. Battery simulation	07/06/2021	20/06/2021		I	2. Battery simulation								
- Simulate SOC	07/06/2021	13/06/2021		si Si	imulate SOC								
- Simulate SOH	07/06/2021	13/06/2021		is I	imulate SOH								
- Simulate resistance	07/06/2021	13/06/2021			imulate resistance								
Document findings and review for afte	14/06/2021	20/06/2021			e - Document findings and review for aft الم	er trrepk							
3. Battery strategies setup	14/06/2021	27/06/2021		L	3. Battery strategies setup								
··· © Simple battery strategy	14/06/2021	27/06/2021											
	14/06/2021	27/06/2021			Document simple battery strateg	y findings							
BREAK: International Research Project Delft	28/06/2021	29/08/2021		6		BREAK: Interr	ational Research I	Project Delft					
Research Methodologies	30/08/2021	12/09/2021				Rese	arch Methodologie	5					
2. Battery simulation	1202/80/0E	12/09/2021				2. Ba	ittery sțmulation						
- Review previous work and document	30/08/2021	05/09/2021				Review p	revious work and do	ocumentation					
 Setup prognostic required SOC model 	30/08/2021	12/09/2021				Setu	p prognostic requir	ed SOC model					
3. Battery strategies setup	13/09/2021	17/10/2021				L	е С	. Battery strategies se	etup				
··· PHM battery strategy	13/09/2021	03/10/2021					PHM batte	ny strategy					
Missions generation	04/10/2021	10/10/2021					Missio	ns generation					
© Sustainability and cost benefit analysis	11/10/2021	17/10/2021					s	ustainability and cos	tt benefit analysis pa	arameters trade-off			
4. Simulation	27/09/2021	11/12/2021							4. Sim	ulation			
··· Battery model simulations	1202/60/22	11/12/2021							Batten	y model simulations			
·· Results and analysis	11/10/2021	11/12/2021					.		Results	s and analysis			
··· BREAK: Holiday	27/10/2021	07/11/2021						BREAK	Holiday				
 Sensitivity analysis 	08/11/2021	11/12/2021							Sensiti	ivity analysis			
5. Reporting and reviewing	27/09/2021	12/12/2021					1		5.Rep	porting and reviewing	8		
 Midterm documentation 	27/09/2021	25/10/2021						📙 Midterm docume	entation		8		
Handin final Literature Study report	26/10/2021	26/10/2021						🔶 Handin final Li	terature Study repor	Ŧ			
·· Midterm presentation	27/10/2021	27/10/2021						🔶 - Midterm prese	ntation				
Implement model feedback	08/11/2021	12/12/2021							Imple	ment model feedbac	ck		
6. Final stage	13/12/2021	20/02/2022							L			6. Final stage	
··· Final thesis reporting	13/12/2021	06/02/2022							Ē.		Final th	esis reporting	
 BREAK: Christmas Holiday 	25/12/2021	02/01/2022							_	BREAK: Chri	istmas Holiday		
··· O Handin draft final thesis report	10/01/2022	10/01/2022								🔷 Handir	n draft final thesis	report	
- Prepare for green light	10/01/2022	24/01/2022									Prepate for gree	n light	
··· Oreen light presentation	25/01/2022	25/01/2022								•	🕨 Gregn light pr	esentation	
 B Handin final thesis report 	07/02/2022	07/02/2022									🔶 Handi	n final thesis report	
Prepare for thesis defense	07/02/2022	20/02/2022										Prepare for thesis defen	Jse
··· © Thesis defense	21/02/2022	21/02/2022										▶ Thesis defense	•
in the second	form of a	trop of the set										<u>31</u>	21

Figure 7.1: Project timeline in the form of a Gantt Chart

This page intentionally left blank.



Appendices

1

Statistical Analysis

A statistical analysis of the results and validation output presented in the scientific paper in Part I of this thesis report is presented below. The following elements are reviewed:

- *Monte Carlo (MC)* the stabilisation of the mean target distance covered by each eUAV battery is checked.
- Distribution of Sustainability and Cost Assessment results the distributions of the \tilde{B} , \tilde{P}_{total} and Efficiency variables are reviewed for normality to determine if a t-test can be applied. This is done by visually inspecting the histrograms of the results, as well as by running Shapiro-Wilk tests and reviewing Quantile-Quantile (Q-Q) plots for variables that show uncertain Shapiro-Wilk outputs. The Shapiro-Wilk test is a powerful method to check the normality of the distributions [89].

From the histogram graphs, $\widetilde{CO}_{2, BOL}$ and \widetilde{Cost}_{BOL} are both related to \widetilde{B} tested with Shapiro-Wilk tests. Furthermore, $\widetilde{CO}_{2, charge}$ and $\widetilde{Cost}_{charge}$ link to \widetilde{P}_{total} . The *Efficiency* distributions are assessed separately with Shapiro-Wilk tests.

• *Statistical T-test* - an overview of the p-values for each unpaired one-sided T-test applied to test the hypotheses presented in the scientific paper in Part I.

The MC, distributions and T-tests for the mixed, short, medium and long range distance results are first analysed in section 1.1, 1.2, 1.3 and 1.4, respectively. Thereafter, the MC plots of the validation results are presented in section 1.6.

1.1. Mixed range

The MC graphs for the mixed range are first discussed. Then, the normality of the results is checked. Finally, the T-test outputs of the tested hypotheses are elaborated on.

1.1.1. Mixed range - Monte Carlo

The MC graphs of the 1000 eUAV batteries tested with the SOC 100%, SOC 80% and mission-based strategy for the mixed range are displayed in Figure 1.1. From the graphs, it is concluded that sufficient batteries are tested as the means converge after approximately 300 runs. It can be seen that the mission-based strategy mean takes the longest to converge due to the bigger variance in battery lifetime, resulting in a larger spread in total distance flown. Due to the convergence of means after 300 runs for all three battery health management strategies, it is chosen to run the MC simulations a total of 300 times for the remaining short, medium and long range to speed up computational running time.



Figure 1.1: Stabilisation of mean distance results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range

1.1.2. Mixed range - Distribution of Sustainability and Cost Assessment results

From the histogram graphs for battery lifetime and battery usage presented Figure 1.9 and 1.10, respectively, it can be seen that the sustainability and cost assessment results seem to follow normal distributions.

Reviewing the Shapiro-Wilk normality check tests in Table 1.1, it is concluded that all outputs except \tilde{B} for the SOC 100% and mission-based strategy are normally distributed. Zooming in on these parameters, Q-Q plots are generated to review how severe the outputs deviate from a normal distribution. The Q-Q plots in 1.2a and 1.2b show that the set of \tilde{B} for the SOC 100% and mission-based strategy are 'sufficiently normal' to apply T-tests.

1.1.3. Mixed range - Statistical T-tests

To test the hypotheses presented in the scientific paper in Part I, Table 1.2 provides a summary of the p-values retrieved from the unpaired one-sided T-tests tested for the mixed range. From the results, it is concluded that the mission-based strategy outperforms the SOC 100% strategy for \tilde{B} , \tilde{P}_{total} and Efficiency. Secondly, the mission-based strategy performs better than the SOC 80% strategy for \tilde{B} and \tilde{P}_{total} . However, the SOC 80% results for Efficiency are higher than the mission-based strategy.

	\widetilde{P}_{total} - mixed range							
Strategy	W	p-value	Normal	Strategy		W	p-value	Normal
SOC 100%	0.990	$1.66 \cdot 10^{-6}$	False	SOC 1009	%	0.998	0.291	True
SOC 80%	0.999	0.664	True	SOC 80%		0.999	0.835	True
Mission-based	0.992	$4.41 \cdot 10^{-5}$	False	Mission-	based	0.999	0.940	True
		Eff	ficiency - r	nixed range	9			
		Strategy	W	p-value	Norn	nal		
		SOC 100%	0.999	0.748	True			
		SOC 80%	0.999	0.844	True			
		Mission-base	ed 0.999	0.705	True			

Table 1.1: Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range



Figure 1.2: Quantile-Quantile (Q-Q) plots for outputs of 1000 batteries tested through Monte Carlo simulation for mixed range

	\widetilde{B} - mixed range					\tilde{P}_{total} - mixed range				
Strategy A	Strateg	gy B	p-v	alue	Strateg	yА	Strategy B	p-value		
SOC 100%	Missio	n-based	0.00	00	SOC 100)%	Mission-based	0.000		
SOC 80%	Mission	n-based	0.00	00	SOC 809	%	Mission-based	0.000		
		E_{j}	ffic	iency -	mixed rai	nge				
		Strategy	γA	Strategy B		p-v	alue			
		SOC 100	00% Missie		n-based	0.00	00			
		SOC 80%	80% Missio		n-based	1.00	00			

Table 1.2: Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range

1.2. Short range

In this section, the statistical analysis of the MC simulations, normality and T-test results are given for the short range.

1.2.1. Short range - Monte Carlo

The MC plots for the short range distance for the SOC 100%, SOC 80% and mission-based strategy are depicted in 1.3a, 1.3b and 1.3c, respectively. The graphs show that the mean target distance flown stabilises, implying that sufficient MC runs are simulated.



Figure 1.3: Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range

1.2.2. Short range - Distribution of Sustainability and Cost Assessment results

The distributions of the battery lifetime and usage results for the short range are presented in histograms in Figure 1.11 and 1.12. From Shapiro-Wilk test results listed Table 1.3 it is concluded that \tilde{B} outputs for the SOC 100% and SOC 80% strategies are not normally distributed. Moreover, the *Efficiency* results for the SOC 80% strategy does not follow a normal distribution. Reviewing the Q-Q plots for these parameters in Figure 1.4, however, it is concluded that the distributions are 'sufficiently normal' to apply T-tests.

1.2.3. Short range - Statistical T-tests

The results of the unpaired one-sided T-tests reviewed for the short range are presented in Table 1.4. From these p-values results, it is concluded that the mission-based strategy performs better than the SOC 100% and SOC 80% approach for \tilde{B} , \tilde{P}_{total} and Efficiency.
\widetilde{B} - short range				\widetilde{P}_{total} - short range				
Strategy	W	p-value	Normal	Strategy		W	p-value	Normal
SOC 100%	0.988	0.012	False	SOC 100%		0.997	0.796	True
SOC 80%	0.984	0.002	False	SOC 80%		0.994	0.317	True
Mission-based	0.992	0.103	True	Mission-b	ased	0.995	0.379	True
	-	Ι	Efficiency					
	_	Strategy	W	p-value	Nor	mal		
	-	SOC 100%	0.997	7 0.800	True	<u>e</u>		
		SOC 80%	0.98	6 0.005	Fals	e		
		Mission-ba	ased 0.993	3 0.189	True	e		

Table 1.3: Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range



Figure 1.4: Quantile-Quantile (Q-Q) plots for outputs of 300 batteries tested through Monte Carlo simulation for short range

	\widetilde{B} - short	range				\widetilde{P}_t	otal - short rang	е
Strategy A	Strateg	уB	p-v	value	Strateg	yА	Strategy B	p-value
SOC 100%	Missior	n-based	0.0	00	SOC 100	0%	Mission-based	0.000
SOC 80%	Missior	n-based	0.0	00	SOC 809	%	Mission-based	0.000
		E	Effi	ciency ·	short ran	ıge		
		Strateg	yА	Strateg	gy B	p-v	alue	
		SOC 100)%	Missio	n-based	0.0	00	
		SOC 809	%	Missio	n-based	0.0	00	

Table 1.4: Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range

1.3. Medium range

For the medium range, the MC simulation plots are first discussed. Then, the results are checked for normality. Lastly, the T-test outputs are presented.



Figure 1.5: Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium range

1.3.1. Medium range - Monte Carlo

The MC plots for the medium range for the SOC 100%, SOC 80% and mission-based strategy are displayed in 1.5a, 1.5b and 1.5c, respectively. From these figures, it can be concluded that sufficient MC simulations are

run, because the means for each plot converge to a stable value.

1.3.2. Medium range - Distribution of Sustainability and Cost Assessment results

The results for the sustainability and cost assessment results for the medium range are given in Figure 1.13 and 1.14. The Shapiro-Wilk results presented in Table 1.5 show that the distributions for \tilde{B} for all three strategies are not normally distributed. Additionally, the \tilde{P}_{total} results for the SOC 80% strategy do not follow a normal distribution. From the Q-Q plots presented in Figure 1.6, however, it is concluded that the results are distributed 'sufficiently normal' for T-tests to be applied.

\widetilde{B}	\widetilde{P}_{total} - medium range							
Strategy	W	p-value M	Normal	Strategy		W	p-value	Normal
SOC 100%	0.981	0.001 H	False	SOC 1009	%	0.993	0.136	True
SOC 80%	0.982	0.001 H	False	SOC 80%		0.988	0.013	False
Mission-based	0.981	$4.74 \cdot 10^{-4}$ H	False	Mission-	based	0.996	0.751	True
		Effic	<i>ciency</i> - m	edium ranş	ge			
		Strategy	W	p-value	Norn	nal		
		SOC 100%	0.996	0.602	True			
		SOC 80%	0.998	0.979	True			
		Mission-based	d 0.997	0.804	True			

Table 1.5: Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium range



Figure 1.6: Quantile-Quantile (Q-Q) plots for outputs of 300 batteries tested through Monte Carlo simulation for medium range

1.3.3. Medium range - Statistical T-tests

Table 1.6 provides an overview of the p-values derived from the unpaired one-sided T-tests performed for the medium range. Here, it is concluded that the mission-based strategy performs better than the SOC 100% strategy for \tilde{B} , \tilde{P}_{total} and Efficiency. When comparing the mission-based strategy to the SOC 80% strategy, however, the results show that the mission-based strategy merely outputs an improved \tilde{P}_{total} .

\widetilde{B} - medium range					\widetilde{P}_{total} - medium range			
Strategy A	Strateg	gy B	p-v	value	Strateg	yА	Strategy B	p-value
SOC 100%	Missio	n-based	0.0	00	SOC 10	0%	Mission-based	0.000
SOC 80%	Missio	n-based	1.0	00	SOC 80	%	Mission-based	0.000
		Ef	fic	iency -	medium ra	ange		
		Strategy	уA	Strate	gy B	p-v	alue	
		SOC 100)%	Missio	on-based	0.0	00	
		SOC 809	%	Missio	on-based	1.0	00	

Table 1.6: Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for medium range

1.4. Long range

Below, the MC results for the long range are discussed. Then, the outputs are checked for normality, after which an overview of the T-test results is given.

1.4.1. Long range - Monte Carlo

From the MC plots for SOC 100%, SOC 80% and mission-based strategy for long range shown in 1.7a, 1.7b and 1.7c, respectively, it is concluded that sufficient MC simulations are run.

1.4.2. Long range - Distribution of Sustainability and Cost Assessment results

The histograms of the battery lifetime and usage performance for the long range are presented in Figure 1.15 and 1.16. In Table 1.7 the Shapiro-Wilk test results for the long range are given. Here, it becomes apparent that the \tilde{B} output for the SOC 100% strategy is not normally distributed. Also, the SOC 100% and mission-based \tilde{P}_{total} results do not follow a normal distribution. Lastly, this is also the case for the SOC 80% *Efficiency* results. For these parameters, Q-Q plots are generated. From the Q-Q graphs depicted in Figure 1.8 it is concluded that the results are 'sufficiently normal' in their distributions for T-tests to be applied.

j	ange		\widetilde{P}_{total} - long range				
Strategy	W	p-value	Normal	Strategy	W	p-value	Normal
SOC 100%	0.981	0.001	False	SOC 100%	0.990	0.046	False
SOC 80%	0.991	0.074	True	SOC 80%	0.996	0.664	True
Mission-based	0.992	0.111	True	Mission-based	0.990	0.033	False

Efficiency - long range						
Strategy	W	p-value	Normal			
SOC 100%	0.994	0.228	True			
SOC 80%	0.979	$1.85 \cdot 10^{-4}$	False			
Mission-based	0.996	0.690	True			

Table 1.7: Shapiro-Wilk normal test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long range



Figure 1.7: Stabilisation of mean distance results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long range

1.4.3. Long range - Statistical T-tests

The unpaired one-sided T-test results for the hypotheses tested for the long range are presented in Table 1.8. For the long range, the mission-based strategy performs better than the SOC 100% and SOC 80% strategy across all assessment parameters \tilde{B} , \tilde{P}_{total} and Efficiency.

	\widetilde{B} - long	range				\tilde{P}_t	otal - long range	
Strategy A	Strateg	уB	p-v	alue	Strateg	yА	Strategy B	p-value
SOC 100%	Missior	1-based	0.0	00	SOC 10	0%	Mission-based	0.000
SOC 80%	Missior	n-based	0.0	00	SOC 80	%	Mission-based	0.000
		i	Effi	ciency	- long ran	ge		
		Cture to a		Crency	- 1011g 1a11	5C		
		Strateg	yА	Strate	ду Б	p-v	alue	
		SOC 10	0%	Missio	n-based	0.0	00	
		SOC 809	%	Missio	n-based	0.0	00	

Table 1.8: Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long range



Figure 1.8: Quantile-Quantile (Q-Q) plots for outputs of 300 batteries tested through Monte Carlo simulation for long range

1.5. Distances - Statistical T-tests

In the scientific paper, the hypothesis that the short range distance outperforms the medium and long range distance within each strategy is tested. Table 1.9 presents the p-values for each test. Here, it is concluded that the short range yields better results than the medium and long range for the SOC 100% and mission-based strategy for \tilde{B} , \tilde{P}_{total} and Efficiency. However, the SOC 80% range results show that the medium range Efficiency has a better performance than the short range distance. For \tilde{B} and Efficiency, the SOC 80% short range does perform better than the other distances.

\widetilde{B} - short,	medium and long rai	nge	\widetilde{P}_{total} - short, medium and long range			
Strategy	Distances	p-value	Strategy	Distances	p-value	
SOC 100%	Short vs. Medium	0.000	SOC 100%	Short vs. Medium	0.000	
SOC 100%	Short vs. Long	0.000	SOC 100%	Short vs. Long	0.000	
SOC 80%	Short vs. Medium	0.000	SOC 80%	Short vs. Medium	0.000	
SOC 80%	Short vs. Long	0.000	SOC 80%	Short vs. Long	0.000	
Mission-based	Short vs. Medium	0.000	Mission-based	Short vs. Medium	0.000	
Mission-based	Short vs. Long	0.000	Mission-based	Short vs. Long	0.000	

<i>Efficiency</i> - short, medium and long range							
Strategy	Distances	p-value					
SOC 100%	Short vs. Medium	0.000					
SOC 100%	Short vs. Long	0.000					
SOC 80%	Short vs. Medium	1.000					
SOC 80%	Short vs. Long	0.000					
Mission-based	Short vs. Medium	0.000					
Mission-based	Short vs. Long	0.000					

Table 1.9: Unpaired one-sided T-test results for \tilde{B} , \tilde{P}_{total} and Efficiency outputs of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short, medium and long range



Figure 1.9: Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range

Sustainability and Cost Assessment (Battery Usage) for SOC 100% strategy for mixed range



Sustainability and Cost Assessment (Battery Usage) for SOC 80% strategy for mixed range Costs_{charge} [€/s] CO2, charge [kg/s] Efficiency 60 60 40 40 20 20 0 0 0.0000050 0.0000052 0.0000054 0.00000255 0.0000048 0.00000225 0.00000230 0.00000235 0.00000240 0.00000245 0.00000250 0.00000220 0.260 0.265 0.270 0.280 0.275 0.285



Figure 1.10: Distribution of Sustainability and Cost Assessment (Battery Usage) results of 1000 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for mixed range

60

40

20

0



Sustainability_and Cost Assessment (Battery Lifetime) for SOC 100% strategy for short range CO_{2, BOL} [kg/s] Cost_{BOL} [€/s]

Sustainability and Cost Assessment (Battery Lifetime) for SOC 80% strategy for short range CO_{2, BOL} [kg/s] Cost_{BOL} [€/s]







Figure 1.11: Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range

Sustainability and Cost Assessment (Battery Usage) for SOC 100% strategy for short range



Figure 1.12: Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for short range

0.280

0.285

0.270

0.275

0.290



simulation for SOC 100%, SOC 80% and mission-based strategy for medium range



Figure 1.13: Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 300 batteries tested through Monte Carlo



Cost_{charge} [€/s]

Sustainability and Cost Assessment (Battery Usage) for mission-based strategy for medium range

Figure 1.14: Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested through Monte Carlo

Cost_{charge} [€/s]

0.00000240

0.00000242



15

10

5

0

20

15

10

0

0.00000216 0.00000218 0.00000220 0.00000222 0.00000224 0.00000226

0.00000230

0.00000232 0.00000234 0.00000236 0.00000238

CO2

0.00000505

 \widetilde{CO}_2

0.00000500

15

10

5

0

20

15

10

5 0

0.00000465

0.00000470

0.00000475

0.00000495

[kg/s]

0.00000510

_{charge} [kg/s]

0.00000480

0.00000485

simulation for SOC 100%, SOC 80% and mission-based strategy for medium range

0.00000515

0.00000520



Efficiency

0.295

Efficiency

0.252 0.252

0.250

0.300

0.253

0.305

0.254 0.255



20

15

10

5

0

15

10

0

0.248

0.290

0.249

0

0.000044

0.000046



Figure 1.15: Distribution of Sustainability and Cost Assessment (Battery Lifetime) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long range

0.000050

0.000052

0.000048

0

0.0032

0.0033 0.0034

0.0035 0.0036 0.0037

0.0038

0.0039



Figure 1.16: Distribution of Sustainability and Cost Assessment (Battery Usage) results of 300 batteries tested through Monte Carlo simulation for SOC 100%, SOC 80% and mission-based strategy for long range

1.6. Validation results

Now that the results for the mixed, short, medium and long range distances have been analysed, this section reviews the outputs of the validation simulations. To validate the model, several sensitivity analyses are run. In Figure 1.17 and 1.18, the MC plots show that the mean target distance flown by each battery converge to a stable value for each sensitivity parameter tested. Therefore, it is concluded that sufficient runs have been simulated.



Mean sum of target distances with simple strategy for Current [A] sensitivity parameter at maximum value 48500 fotal target distance [m] 48000 47500 47000 46500 50 100 150 200 250 300 ò Runs

Mean sum of target distances with simple strategy

for Time [s]] sensitivity parameter at maximum value

150

Runs

200

250

300

(a) Current [A] with minimum value

(b) Current [A] with maximum value

53400





100

(c) Time [s] with minimum value

(d) Time [s] with maximum value

50

ò



Figure 1.17: Sensitivity Analysis (Part 1) - Stabilisation of mean distance flown per battery through Monte Carlo runs per sensitivity parameter varied for minimum and maximum values



(a) SEI kinetic rate $[ms^{-1}]$ with minimum value



Mean sum of target distances with simple strategy

Mean sum of target distances with simple strategy

(b) SEI kinetic rate $[ms^{-1}]$ with maximum value



(d) Initial capacity [Ah] with maximum value



(f) Initial voltage [V] with maximum value

Figure 1.18: Sensitivity Analysis (Part 2) - Stabilisation of mean distance flown per battery through Monte Carlo runs per sensitivity parameter varied for minimum and maximum values

Mean sum of target distances with simple strategy for Initial capacity [Ah] sensitivity parameter at minimum value 49000 Total target distance [m] 48500 48000

150

Runs

250

зóо

200

(c) Initial capacity [Ah] with minimum value

100

50

47500

47000

This page intentionally left blank.

2 Research Limitations and Recommendations

A review of the limitations of this study originating from assumptions and other external factors is given below. The mission profiles are first discussed in section 2.1. Secondly, the battery model and health management strategies are reviewed in section 2.2 and 2.3. Lastly, limitations of the sustainability and cost assessment are briefly addressed in section 2.4

2.1. Mission Profiles

In order to establish the mission profiles, DJI Matrice 100 quadcopter flying data from the research by Rodrigues et al. [105] is used. Translating this into the battery model used for this study, several shorting comings are listed below.

The derived current throughput and time duration values from their study are implemented into the experiments in PyBaMM as constant values throughout a manoeuvre. However, the accuracy of the research's model would be increased if this is inputted as a so-called 'drive cycle' which specifies the current rate for smaller time periods (for example per second). In addition, the eUAV modelled flight only consists of simple manoeuvres. To more closely replicate real-life eUAV flights, additional movements including turns should be included, as well as more variation in flying plan settings such as speed, altitude and payload. Although these two limitations decrease the accuracy of the model compared to a real-life eUAV application, it is not expected that the results would be significantly influenced.

Finally, apart from the 2.6% standard deviation inputted in the current and time period values, this model does not account for environmental influences such as wind which would considerably influence the eUAV battery operating conditions. Although the SOC 80% and mission-based battery health management strategy showed an improved battery lifetime and usage performance compared to the SOC 100% approach, operators could argue that mission safety is reduced due a reduction of available battery capacity for flight. The model would be improved when more data is inputted into the mission-based strategy such that external factors such as wind are accounted for. An assessment parameter reflecting on flight safety could also be implemented to evaluate the performance of the mission-based model.

2.2. Battery Model

For the battery model, several uncertainties are identified. Foremostly, the non-deterministic errors that are inputted in the MC simulations originate from several different researches and may therefore be case specific. Depending on the battery and BMS used, the battery and non-deterministic model inputs in should be adjusted accordingly.

Additionally, this model incorporates battery ageing by using the SEI kinetic rate equal derived by Yang et al. [133]. For Lithium batteries, this is the dominating degradation phenomenon while SOH > 80%. For more accurate results, Lithium plating could also be included in the model.

To model an eUAV battery, multiple battery cells that form a battery pack could be analysed, instead of using single battery cells. It is important to review the performance of a battery pack as battery degradation is more severe compared to single cells. To model this is, however, very complex to model as interrelated ageing modes arise related parameters such as temperature and resistance. Nonetheless, due to the large demand for such models, PyBaMM is currently developing a separate module called 'Liionpack' which could potentially be used.

The last battery model limitation that became apparent during the MC simulations, is the high computational power required to run through the model. On average, one battery MC iteration took 45 minutes to complete. This is a highly constraining factor when testing the model in the SA and case studies, resulting in the MC simulations only being run 300 times. Especially for SA which is performed to validate the model, the computational intensity restrained the variation of tests executed. Executing other SA experiments such as an extensive variance decomposition to review how the output is affected by multiple stochastic inputs, for example, would enhance the certainty of the conclusions about the robustness of the model.

Despite the fact that the accuracy of the battery model is reduced by these battery model assumptions, the overall performance of the battery health management strategies is not expected to change.

2.3. Battery Health Management Strategies

The biggest limitation of PyBaMM is the absence of a SOC function which creates an obstacle of twofold. First of all, when simulating charging operations, PyBaMM takes voltage values as input instead of SOC levels. For the SOC 100% charging strategy, this is not an issue as the battery can always be charged to the initially defined maximum voltage. However, for the SOC 80% and mission-based battery health management strategy, the SOC to which the battery model needs to be charged varies, implying that these SOC values are required to be converted to voltage levels.

Secondly, the lack of a standard SOC function results in another limitation, namely in the setup of the SOC 80% and mission-based model. To estimate the required SOC to complete a flight, data from SOC 100% charging model is used. Here, the SOC required to complete a certain flight is again translated to voltage values. Because Lithium batteries are characterised by an S-shaped discharge voltage curve, this leads to small discrepancies. Effectively, this means that for example 40% SOC DOD does not yield the same voltage DOD when starting from 100% SOC compared to lower mission-based values such as 70%. Due to the S-shaped voltage curves, the mission-based model overestimates the expected voltage DOD.

In this model, SOC conversion to voltage levels is done through regression using SOH and target distance as an input. But as battery electrochemical relations are complex, an regression is supposedly not able to capture the intricacy of SOC couplings. Moreover, PyBaMM brings along the constraint that the SOH cannot be determined during operation and only at the start and end of each flight. Alternatively, other approaches such as Machine Learning could yield higher accuracy.

2.4. Cost-Benefit Analysis

The last set of limitations revolves around the assessment of the model outputs. These limitations are attributed to the lack of battery specific data. For the environmental impact, the values of the BOL CO_2 emissions strongly vary per source. Moreover, a more extensive analysis of other sustainability parameters such as material resources, waste and EOL impact would enhance the thoroughness of the analysis. In the financial cost bucket, the costs of EOL practices as well as investment of setting up the mission-based and SOC 80% battery health management strategy are neglected. By disregarding the SOC estimation investment costs, the financial results of the mission-based and SOC 80% strategy are slightly optimistic. Lastly, the sustainability and cost assessment lacks a single final 'score' which combines the outcome of the environmental, financial and efficiency assessments.

Bibliography

- [1] Shannon P Ackert. Evaluation & Insights of Commercial Aircraft Maintenance Programs. Basics of Aircraft Maintenance Programs for Financiers, pages 1–23, 2010.
- [2] National Aeronautics and Space Administration. Prognostics center of excellence datasets. https: //ti.arc.nasa.gov/tech/dash/groups/pcoe/prognostic-data-repository. Retrieved: 16-04-2021.
- [3] Ahmed Abdullah Al-karakchi, Gillian Lacey, and Ghanim Putrus. A method of electric vehicle charging to improve battery life. In 2015 50th International Universities Power Engineering Conference (UPEC), pages 1–3, 2015. doi: 10.1109/UPEC.2015.7339846.
- [4] V.N. Ajukumar and O.P. Gandhi. Evaluation of green maintenance initiatives in design and development of mechanical systems using an integrated approach. Journal of Cleaner Production, 51:34–46, 2013. ISSN 0959-6526. doi: 10.1016/j.jclepro.2013.01.010.
- [5] Abudllah Alabdulkarim, Peter Ball, and Ashutosh Tiwari. Applications of simulation in maintenance research. World Journal of Modelling and Simulation, 9(1):14–37, 2013. ISSN 1746-7233.
- [6] Marco Andreacchio, Abdelghani Bekrar, Rachid Benmansour, and Damien Trentesaux. Balancing preventive and corrective maintenance of aircraft assets: A cyber-physical systems approach. In IEEE 14th International Conference on Industrial Informatics (INDIN), volume 0, pages 500–503. Institute of Electrical and Electronics Engineers Inc., jul 2016. ISBN 9781509028702. doi: 10.1109/INDIN.2016.7819213.
- [7] Jean-Philippe Aurambout, Konstantinos Gkoumas, and Biagio Ciuffo. Last mile delivery by drones: an estimation of viable market potential and access to citizens across european cities. European Transport Research Review, 11, 12 2019. doi: 10.1186/s12544-019-0368-2.
- [8] Reg Austin. Unmanned Aircraft Systems. John Wiley and Sons Ltd., 1st edition, 2010. ISBN 9780470058190.
- [9] Jeffrey Banks and John Merenich. Cost benefit analysis for asset health management technology. In 2007 Annual Reliability and Maintainability Symposium, pages 95–100, 2007. doi: 10.1109/RAMS.2007. 328097.
- [10] Anup Barai, Dhammika Widanage, James Marco, Andrew Mcgordon, and Paul Jennings. A study of the open circuit voltage characterization technique and hysteresis assessment of lithium-ion cells. Journal of Power Sources, 295:99–107, jul 2015. ISSN 03787753. doi: 10.1016/j.jpowsour.2015.06.140.
- [11] Federico Baronti, Nicola Femia, Roberto Saletti, Ciro Visone, and Walter Zamboni. Hysteresis modeling in li-ion batteries. IEEE Transactions on Magnetics, 50(11):1–4, 2014. doi: 10.1109/TMAG.2014. 2323426.
- [12] Anthony Barré, Benjamin Deguilhem, Sébastien Grolleau, Mathias Gérard, Frédéric Suard, and Delphine Riu. A review on lithium-ion battery ageing mechanisms and estimations for automotive applications. Journal of Power Sources, 241:680–689, nov 2013. ISSN 03787753. doi: 10.1016/j.jpowsour. 2013.05.040.
- [13] Thorsten Baumhöfer, Manuel Brühl, Susanne Rothgang, and Dirk Uwe Sauer. Production caused variation in capacity aging trend and correlation to initial cell performance. Journal of Power Sources, 247: 332–338, 2014. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2013.08.108.
- [14] Jack Bell. Condition Based Maintenance Plus Department of Defense Guidebook. Technical report, 2008.

- [15] Ashyrmuhammet Berdinyazov, Fatih Camci, Mehmet Sevkli, and Saim Baskan. Economic analysis of maintenance policies for a system. In 2009 IEEE International Symposium on Diagnostics for Electric Machines, Power Electronics and Drives, SDEMPED 2009, 2009. ISBN 9781424434411. doi: 10.1109/ DEMPED.2009.5292766.
- [16] Maurizio Bevilacqua, Filippo Ciarapica, Giovanni Mazzuto, and Leonardo Postacchini. Air traffic management of an airport using discrete event simulation method. In 2012 IEEE International Conference on Industrial Engineering and Engineering Management, pages 1034–1038. IEEE, 2012.
- [17] Christoph Birkl, Matthew Roberts, Euan McTurk, Peter Bruce, and David Howey. Degradation diagnostics for lithium ion cells. Journal of Power Sources, 341:373–386, feb 2017. ISSN 03787753. doi: 10.1016/j.jpowsour.2016.12.011.
- [18] Grzegorz Bocewicz, Peter Nielsen, Zbigniew Banaszak, and Amila Thibbotuwawa. A declarative modelling framework for routing of multiple uavs in a system with mobile battery swapping stations. In Anna Burduk, Edward Chlebus, Tomasz Nowakowski, and Agnieszka Tubis, editors, Intelligent Systems in Production Engineering and Maintenance, pages 429–441, Cham, 2019. Springer International Publishing. ISBN 9783319974903.
- [19] Robert Cameron Bolam, Yuriy Vagapov, and Alecksey Anuchin. A review of electrical motor topologies for aircraft propulsion. In 2020 55th International Universities Power Engineering Conference (UPEC), pages 1–6, 2020. doi: 10.1109/UPEC49904.2020.9209783.
- [20] Brian Bole, Chetan S Kulkarni, and Matthew Daigle. Adaptation of an Electrochemistry-based Li-Ion Battery Model to Account for Deterioration Observed Under Randomized Use. In Annual Conference of the Prognostics and Health Management Society 2014, pages 502–510, 2014.
- [21] Emanuele Borgonovo and Elmar Plischke. Sensitivity analysis: a review of recent advances. European Journal of Operational Research, 248(3):869–887, 2016.
- [22] Alexander Busse, Joachim Metternich, and Eberhard Abele. Evaluating the Benefits of Predictive Maintenance in Production: A Holistic Approach for Cost-Benefit-Analysis. In Advances in Production Research, pages 690–704. Springer International Publishing, nov 2019. doi: 10.1007/978-3-030-03451-1_67.
- [23] Isidor Buchmann Cadex Electronics Inc. Battery university. https://batteryuniversity.com/. Retrieved: 19-04-2021.
- [24] European Committee for Standardization (CEN). Maintenance Maintenance terminology. British Standards Institution, 2010. ISBN 9780580641848.
- [25] Centraal Bureau van Statistieken (CBS). 11 procent energieverbruik in 2020 afkomstig uit hernieuwbare bronnen. https://www.cbs.nl/nl-nl/nieuws/2021/22/ 11-procent-energieverbruik-in-2020-afkomstig-uit-hernieuwbare-bronnen. Retrieved: 26-11-2021.
- [26] Wen-Yeau Chang. The State of Charge Estimating Methods for Battery: A Review. International Scholarly Research Notices Applied Mathematics, 2013:1–7, jul 2013. doi: 10.1155/2013/953792.
- [27] Quach Cuong Chi, Brian Bole, Edward Hogge, Sixto Vazquez, Matthew Daigle, José Celaya, Adam Weber, and Kai Goebel. Battery Charge Depletion Prediction on an Electric Aircraft. In Annual Conference of the Prognostics and Health Management Society 2013, pages 1–11, 2013.
- [28] Leszek Cwojdziński and Mirosław Adamski. Power units and power supply systems in UAV. Aviation, 18(1):1–8, jan 2014. ISSN 16487788. doi: 10.3846/16487788.2014.865938.
- [29] Jing Dai and Haifeng Wang. System health management for Unmanned Aerial Vehicle: Conception, state-of-art, framework and challenge. In International Conference on Electronic Measurement and Instruments, volume 2, pages 859–863, 2013. ISBN 9781479907571. doi: 10.1109/ICEMI.2013.6743144.
- [30] Jesse Daniels, Paul Werner, and Terry Bahill. Quantitative methods for tradeoff analyses. Systems Engineering, 4(3):190–212, jan 2001. ISSN 10981241. doi: 10.1002/sys.1016.

- [31] Frederik Michel Dekking, Cornelis Kraaikamp, Hendrik Paul Lopuhaä, and Ludolf Erwin Meester. A Modern Introduction to Probability and Statistics: Understanding Why and How. Springer Texts in Statistics. Springer, 2005. ISBN 9781852338961.
- [32] Airworthiness Department. ANAC-EASA-FAA-TCCA Certification Management Team Collaboration Strategy. 2016.
- [33] Ting Dong and Nam Kim. Cost-Effectiveness of Structural Health Monitoring in Fuselage Maintenance of the Civil Aviation Industry †. Aerospace, 5(3):87, 2018. doi: 10.3390/aerospace5030087.
- [34] Gonçalo dos Reis, Calum Strange, Mohit Yadav, and Shawn Li. Lithium-ion battery data and where to find it. Energy and AI, 5:100081, 2021. ISSN 2666-5468. doi: 10.1016/j.egyai.2021.100081.
- [35] Diogo Duarte, Bruno Marado, João Nogueira, Bruno Serrano, Virgínia Infante, and Filipa Moleiro. An overview on how failure analysis contributes to flight safety in the Portuguese Air Force. Engineering Failure Analysis, 65:86–101, jul 2016. ISSN 13506307. doi: 10.1016/j.engfailanal.2016.03.003.
- [36] Idriss El-Thalji, Santiago Echeverri Duque, and Helge Nordal. Design for Intelligent Maintenance: A Potential Reference Standard Complies with Industry 4.0 Requirements. Technical report, 2020.
- [37] Nick Eleftheroglou, Sina Sharif Mansouri, Theodoros Loutas, Petros Karvelis, George Georgoulas, George Nikolakopoulos, and Dimitrios Zarouchas. Intelligent data-driven prognostic methodologies for the real-time remaining useful life until the end-of-discharge estimation of the Lithium-Polymer batteries of unmanned aerial vehicles with uncertainty quantification. Applied Energy, 254:113677, nov 2019. ISSN 03062619. doi: 10.1016/j.apenergy.2019.113677.
- [38] Eurostat. Electricity price statistics. https://ec.europa.eu/eurostat/statistics-explained/. Retrieved: 26-11-2021.
- [39] Alexander Farmann, Wladislaw Waag, Andrea Marongiu, and Dirk Uwe Sauer. Critical review of onboard capacity estimation techniques for lithium-ion batteries in electric and hybrid electric vehicles. Journal of Power Sources, 281:114–130, 2015. ISSN 03787753. doi: 10.1016/j.jpowsour.2015.01.129.
- [40] Kiri Feldman, Taoufik Jazouli, Peter Sandborn, and Senior Member. A Methodology for Determining the Return on Investment Associated with Prognostics and Health Management. IEEE Transactions on Reliability, 58(2):305–316, 2009.
- [41] Filippo De Florio. Airworthiness: An Introduction to Aircraft Certification. Elsevier LTd., 1st edition, 2006. ISBN 9780750669481.
- [42] Center for Advanced Life Cycle Engineering. Battery data. https://calce.umd.edu/battery-data. Retrieved: 25-11-2021.
- [43] Chiara Franciosi, Benoit Iung, Salvatore Miranda, and Stefano Riemma. Maintenance for Sustainability in the Industry 4.0 context: a Scoping Literature Review. IFAC-PapersOnLine, 51(11):903–908, jan 2018. ISSN 24058963. doi: 10.1016/j.ifacol.2018.08.459.
- [44] Chiara Franciosi, Alexandre Voisin, Salvatore Miranda, Stefano Riemma, and Benoit Iung. Measuring maintenance impacts on sustainability of manufacturing industries: from a systematic literature review to a framework proposal, jul 2020. ISSN 09596526.
- [45] Boris Galkin, Jacek Kibilda, and Luiz A. DaSilva. Uavs as mobile infrastructure: Addressing battery lifetime. IEEE Communications Magazine, 57(6):132–137, 2019. doi: 10.1109/MCOM.2019.1800545.
- [46] Foad Gandoman, Joris Jaguemont, Shovon Goutam, Rahul Gopalakrishnan, Yousef Firouz, Theodoros Kalogiannis, Noshin Omar, and Joeri Van Mierlo. Concept of reliability and safety assessment of lithium-ion batteries in electric vehicles: Basics, progress, and challenges. Applied Energy, 251:113343, 2019. ISSN 0306-2619. doi: 10.1016/j.apenergy.2019.113343.
- [47] Yang Gao, Jiuchun Jiang, Caiping Zhang, Weige Zhang, and Yan Jiang. Aging mechanisms under different state-of-charge ranges and the multi-indicators system of state-of-health for lithium-ion battery with Li(NiMnCo)O2 cathode. Journal of Power Sources, 400:641–651, oct 2018. ISSN 03787753. doi: 10.1016/j.jpowsour.2018.07.018.

- [48] Yawei Ge, Mingqing Xiao, Zhao Yang, Lei Zhang, Zewen Hu, and Delong Feng. An integrated logarithmic fuzzy preference programming based methodology for optimum maintenance strategies selection. Applied Soft Computing, 60:591–601, 2017. ISSN 1568-4946. doi: 10.1016/j.asoc.2017.07.021.
- [49] Farnaz Ghazi Nezami and Mehmet Yildirim. A sustainability approach for selecting maintenance strategy. International Journal of Sustainable Engineering ISSN:, 6(4):332–343, 2013. ISSN 1939-7046. doi: 10.1080/19397038.2013.765928.
- [50] Eduardo Gilabert, Santiago Fernandez, Aitor Arnaiz, and Egoitz Konde. Simulation of predictive maintenance strategies for cost-effectiveness analysis. Proceedings of the Institution of Mechanical Engineers, Part B: Journal of Engineering Manufacture, 231(13):2242–2250, 2017. doi: 10.1177/ 0954405415578594.
- [51] Anne Goodchild and Jordan Toy. Delivery by drone: An evaluation of unmanned aerial vehicle technology in reducing co2 emissions in the delivery service industry. Transportation Research Part D: Transport and Environment, 61:58–67, 2018. ISSN 1361-9209. doi: 10.1016/j.trd.2017.02.017. Innovative Approaches to Improve the Environmental Performance of Supply Chains and Freight Transportation Systems.
- [52] Clare Grey and Jean-Marie Tarascon. Sustainability and in situ monitoring in battery development, dec 2017. ISSN 14764660.
- [53] Antonio Jesús Guillén, Adolfo Crespo, Marco Macchi, and Francisco José Gómez. On the role of Prognostics and Health Management in advanced maintenance systems. Production Planning and Control, 27(12):991–1004, sep 2016. ISSN 13665871. doi: 10.1080/09537287.2016.1171920.
- [54] Gilbert Haddad, Peter Sandborn, and Michael Pecht. An options approach for decision support of systems with prognostic capabilities. IEEE Transactions on Reliability, 61(4):872–883, 2012. ISSN 00189529. doi: 10.1109/TR.2012.2220699.
- [55] Han Hao, Zhexuan Mu, Shuhua Jiang, Zongwei Liu, and Fuquan Zhao. Ghg emissions from the production of lithium-ion batteries for electric vehicles in china. Sustainability, 9(4), 2017. ISSN 2071-1050. doi: 10.3390/su9040504.
- [56] Avijit Hazra and Nithya Gogtay. Biostatistics series module 7: The statistics of diagnostic tests. Indian Journal of Dermatology, 62(1):18, 2017. doi: 10.4103/0019-5154.198047.
- [57] Martijn Hendriks, Marc Geilen, and Twan Basten. Pareto analysis with uncertainty. In 2011 IFIP 9th International Conference on Embedded and Ubiquitous Computing, pages 189–196, 2011. doi: 10. 1109/EUC.2011.54.
- [58] Anh Hoang, Phuc Do, and Benoît Iung. Investigation on the use of energy efficiency for conditionbased maintenance decision-making. IFAC-PapersOnLine, 49(28):73–78, jan 2016. ISSN 24058963. doi: 10.1016/j.ifacol.2016.11.013.
- [59] Alan Hobbs and Ann Williamson. Associations between errors and contributing factors in aircraft maintenance. Human Factors, 45(2):186–201, 2003. doi: 10.1518/hfes.45.2.186.27244. PMID: 14529193.
- [60] Edward Hogge, Chetan S Kulkarni, Sixto Vazquez, Kyle Smalling, Thomas Strom, Boyd Hill, and Cuong Quach. Flight Tests of a Remaining Flying Time Prediction System for Small Electric Aircraft in the Presence of Faults. In Conference of the Prognostics and Health Management Society, pages 1–12, 2017.
- [61] Edward Hogge, Brian Bole, Sixto Vazquez, Chetan Kulkarni, Thomas Strom, Boyd Hill, Kyle Smalling, and Cuong Quach. Verification of Prognostic Algorithms to Predict Remaining Flying Time for Electric Unmanned Vehicles. International Journal of Prognostics and Health Management, 021, 2018.
- [62] Nico Hölzel, Thomas Schilling, and Volker Gollnick. An aircraft lifecycle approach for the cost-benefit analysis of prognostics and condition-based maintenance-based on discrete-event simulation. Technical report, DLR-German Aerospace Center Hamburg Germany, 2014.

- [63] Bin Huang, Zhefei Pan, Xiangyu Su, and Liang An. Recycling of lithium-ion batteries: Recent advances and perspectives. Journal of Power Sources, 399:274–286, 2018. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2018.07.116.
- [64] Benoit Iung and Eric Levrat. Advanced maintenance services for promoting sustainability. Procedia CIRP, 22:15–22, 2014. ISSN 2212-8271. doi: 10.1016/j.procir.2014.07.018. Proceedings of the 3rd International Conference in Through-life Engineering Services.
- [65] Andrew Jardine, Daming Lin, and Dragan Banjevic. A review on machinery diagnostics and prognostics implementing condition-based maintenance. Mechanical Systems and Signal Processing, 20(7):1483– 1510, oct 2006. ISSN 08883270. doi: 10.1016/j.ymssp.2005.09.012.
- [66] Peter Keil, Simon Schuster, Jörn Wilhelm, Julian Travi, Andreas Hauser, Ralph C. Karl, and Andreas Jossen. Calendar Aging of Lithium-Ion Batteries. Journal of The Electrochemical Society, 163(9):A1872– A1880, jul 2016. ISSN 0013-4651. doi: 10.1149/2.0411609jes.
- [67] Renee Kent and Dennis Murphy. Health Monitoring System Technology Assessments Cost Benefits Analysis. Technical report, ARINC Aeronautical Radio, Incorporated, 2000.
- [68] Harry Kinnison and Tariq Siddiqui. Aviation Maintenance Management. The McGraw-Hill Companies, Inc, 2nd edition, 2004. ISBN 9780071805025.
- [69] Reinhard Klein, Nalin Chaturvedi, Jake Christensen, Jasim Ahmed, Rolf Findeisen, and Aleksandar Kojic. Optimal charging strategies in lithium-ion battery. In Proceedings of the 2011 American Control Conference, pages 382–387, 2011. doi: 10.1109/ACC.2011.5991497.
- [70] Jarotwan Koiwanit. Analysis of environmental impacts of drone delivery on an online shopping system. Advances in Climate Change Research, 9(3):201–207, 2018. ISSN 1674-9278. doi: 10.1016/j.accre.2018. 09.001.
- [71] Laura Lander, Tom Cleaver, Mohammad Ali Rajaeifar, Viet Nguyen-Tien, Robert J.R. Elliott, Oliver Heidrich, Emma Kendrick, Jacqueline Sophie Edge, and Gregory Offer. Financial viability of electric vehicle lithium-ion battery recycling. iScience, 24(7):102787, 2021. ISSN 2589-0042. doi: 10.1016/j.isci.2021. 102787.
- [72] Jay Lee, Jun Ni, Dragan Djurdjanovic, Hai Qiu, and Haitao Liao. Intelligent prognostics tools and e-maintenance. Computers in Industry, 57(6):476–489, aug 2006. ISSN 01663615. doi: 10.1016/j. compind.2006.02.014.
- [73] Minwoo Lee, Larry Li, and Wenbin Song. Analysis of direct operating cost of wide-body passenger aircraft: A parametric study based on Hong Kong. Chinese Journal of Aeronautics, 32(5):1222–1243, May 2019. ISSN 10009361. doi: 10.1016/j.cja.2019.03.011.
- [74] Yaguo Lei, Naipeng Li, Liang Guo, Ningbo Li, Tao Yan, and Jing Lin. Machinery health prognostics: A systematic review from data acquisition to RUL prediction. Mechanical Systems and Signal Processing, 104:799–834, may 2018. ISSN 10961216. doi: 10.1016/j.ymssp.2017.11.016.
- [75] Jie Li, Kasim Adewuyi, Nima Lotfi, Robert Landers, and Jonghyun Park. A single particle model with chemical/mechanical degradation physics for lithium ion battery state of health (soh) estimation. Applied Energy, 212:1178–1190, 2018. ISSN 0306-2619. doi: 10.1016/j.apenergy.2018.01.011.
- [76] Niansi Li, Xiaoyong Liu, Bendong Yu, Liang Li, Jianqiang Xu, and Qiong Tan. Study on the environmental adaptability of lithium-ion battery powered uav under extreme temperature conditions. Energy, 219:119481, 2021. ISSN 0360-5442. doi: 10.1016/j.energy.2020.119481.
- [77] Xin Li, Konstantinos J. Chalvatzis, and Phedeas Stephanides. Innovative energy islands: Life-cycle costbenefit analysis for battery energy storage. Sustainability, 10(10), 2018. ISSN 2071-1050. doi: 10.3390/ su10103371.
- [78] Yi Li, Kailong Liu, Aoife Foley, Alana Zülke, Maitane Berecibar, Elise Nanini-Maury, Joeri Van Mierlo, and Harry Hoster. Data-driven health estimation and lifetime prediction of lithium-ion batteries: A review, oct 2019. ISSN 18790690.

- [79] Lixia Liao, Pengjian Zuo, Yulin Ma, XinQun Chen, Yongxin An, Yunzhi Gao, and Geping Yin. Effects of temperature on charge/discharge behaviors of lifepo4 cathode for li-ion batteries. Electrochimica Acta, 60:269–273, 2012.
- [80] Kailong Liu, Xiaosong Hu, Zhile Yang, Yi Xie, and Shengzhong Feng. Lithium-ion battery charging management considering economic costs of electrical energy loss and battery degradation. Energy Conversion and Management, 195:167–179, 2019. ISSN 0196-8904. doi: 10.1016/j.enconman.2019.04. 065.
- [81] Maxim Lu, Alex James, and Mehdi Bagheri. Unmanned aerial vehicle (uav) charging from powerlines. In 2017 IEEE PES Asia-Pacific Power and Energy Engineering Conference (APPEEC), pages 1–6, 2017. doi: 10.1109/APPEEC.2017.8308912.
- [82] Nicholas Lutsey and Dale Hall. Effects of battery manufacturing on electric vehicle life-cycle greenhouse gas emissions. International Council on Clean Transportation, 02 2018.
- [83] Shuai Ma, Modi Jiang, Peng Tao, Chengyi Song, Jianbo Wu, Jun Wang, Tao Deng, and Wen Shang. Temperature effect and thermal impact in lithium-ion batteries: A review. Progress in Natural Science: Materials International, 28(6):653–666, 2018. ISSN 1002-0071. doi: 10.1016/j.pnsc.2018.11.002.
- [84] Andrzej Majka and Pawel Wacnik. Cooperation in aviation beyond Europe's Borders. In European Conference for Aeronautics and Space Sciences (EUCASS), 2019. doi: 10.13009/EUCASS2019-852.
- [85] Sina Sharif Mansouri, Petros Karvelis, George Georgoulas, and George Nikolakopoulos. Remaining Useful Battery Life Prediction for UAVs based on Machine Learning. IFAC-PapersOnLine, 50(1):4727– 4732, jul 2017. ISSN 24058963. doi: 10.1016/j.ifacol.2017.08.863.
- [86] Manoj Mathew, Stefan Janhunen, Mahir Rashid, Frank Long, and Michael Fowler. Comparative analysis of lithium-ion battery resistance estimation techniques for battery management systems. Energies, 11 (6), 2018. ISSN 1996-1073. doi: 10.3390/en11061490.
- [87] Michael McFadden and D. Scott Worrells. Global Outsourcing of Aircraft Maintenance. Journal of Aviation Technology and Engineering, 1(2):63–73, April 2012. ISSN 2159-6670. doi: 10.5703/1288284314659.
- [88] Hans Eric Melin. The lithium-ion battery end-of-life market–a baseline study. In World Economic Forum: Cologny, Switzerland, volume 2018, pages 1–11, 2018.
- [89] Nornadiah Mohd Razali and Bee Yap. Power comparisons of shapiro-wilk, kolmogorov-smirnov, lilliefors and anderson-darling tests. Journal of Statistical Modeling and Analytics, 2, 01 2011.
- [90] Melissa Morris and Sabri Tosunoglu. Comparison of Rechargeable Battery Technologies. In ASME Early Career Technical Conference (ECTC), 2012.
- [91] Mohammad Moshref-Javadi, Ahmad Hemmati, and Matthias Winkenbach. A truck and drones model for last-mile delivery: A mathematical model and heuristic approach. Applied Mathematical Modelling, 80:290–318, 2020. ISSN 0307-904X. doi: 10.1016/j.apm.2019.11.020.
- [92] Nirmal-Kumar C. Nair and Niraj Garimella. Battery energy storage systems: Assessment for small-scale renewable energy integration. Energy and Buildings, 42(11):2124–2130, 2010. ISSN 0378-7788. doi: 10.1016/j.enbuild.2010.07.002.
- [93] Gang Ning, Bala Haran, and Branko N. Popov. Capacity fade study of lithium-ion batteries cycled at high discharge rates. Journal of Power Sources, 117(1):160–169, 2003. ISSN 0378-7753. doi: 10.1016/ S0378-7753(03)00029-6.
- [94] International Organization of Standardization. Iso-13374 condition monitory and diagnostics of machines. Technical report, ISO, 2012.
- [95] Maria Palacín. Recent advances in rechargeable battery materials: a chemist's perspective. Chemical Society Reviews, 38(9):2565–2575, 2009. doi: 10.1039/b820555h.

- [96] Sangyoung Park, Licong Zhang, and Samarjit Chakraborty. Design space exploration of drone infrastructure for large-scale delivery services. In Proceedings of the 35th International Conference on Computer-Aided Design, ICCAD '16, New York, NY, USA, 2016. Association for Computing Machinery. ISBN 9781450344661. doi: 10.1145/2966986.2967022.
- [97] Sangyoung Park, Licong Zhang, and Samarjit Chakraborty. Battery assignment and scheduling for drone delivery businesses. In IEEE/ACM International Symposium on Low Power Electronics and Design (ISLPED), pages 1–6, 2017. doi: 10.1109/ISLPED.2017.8009165.
- [98] Lalit Patnaik and Sheldon Williamson. A Five-Parameter Analytical Curvefit Model for Open-Circuit Voltage Variation with State-of-Charge of a Rechargeable Battery. In IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), pages 1–4. Institute of Electrical and Electronics Engineers Inc., jul 2018. ISBN 9781538693155. doi: 10.1109/PEDES.2018.8707806.
- [99] Michael Pecht and Anthony Rafanelli. Product Reliability, Maintainability, and Supportability Handbook. 2. CRC Press, 2009. ISBN 9780849398797. doi: 10.1115/1.2792151.
- [100] Enrico Petritoli, Fabio Leccese, and Lorenzo Ciani. Reliability and Maintenance Analysis of Unmanned Aerial Vehicles †. Sensors (Basel), 18(9):3171, 2018. doi: 10.3390/s18093171.
- [101] Jon Pierre and Guy Peters. From a club to a bureaucracy: JAA, EASA, and European aviation regulation. Journal of European Public Policy, 16(3):337–355, April 2009. ISSN 13501763. doi: 10.1080/ 13501760802662706.
- [102] Ahmad Ali Pohya, Kai Wicke, and Johannes Hartmann. Comparison of direct operating cost and life cycle cost-benefit methods in aircraft technology assessment. In AIAA Aerospace Sciences Meeting, 2018. American Institute of Aeronautics and Astronautics Inc, AIAA, January 2018. ISBN 9781624105241. doi: 10.2514/6.2018-0282.
- [103] PyBaMM. Python battery mathematical modelling, github. https://github.com/pybamm-team. Retrieved: 25-11-2021.
- [104] Seyed Mohammad Rezvanizaniani, Zongchang Liu, Yan Chen, and Jay Lee. Review and recent advances in battery health monitoring and prognostics technologies for electric vehicle (ev) safety and mobility. Journal of Power Sources, 256:110–124, 2014. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2014.01.085.
- [105] Thiago Rodrigues, Jay Patrikar, Arnav Choudhry, Jacob Feldgoise, Vaibhav Arcot, Aradhana Gahlaut, Sophia Lau, Brady Moon, Bastian Wagner, H. Scott Matthews, Sebastian Scherer, and Constantine Samaras. In-flight positional and energy use data set of a dji matrice 100 quadcopter for small package delivery. Scientific Data, 8, 06 2021. doi: 10.1038/s41597-021-00930-x.
- [106] Mia Romare and Lisbeth Dahllöf. The life cycle energy consumption and greenhouse gas emissions from lithium-ion batteries. IVL Swedish Environmental Research Institute, C243, 2017.
- [107] Bhaskar Saha, Edwin Koshimoto, Cuong Quach, Edward Hogge, Thomas H. Strom, Boyd Hill, Sixto Vazquez, and Kai Goebel. Battery health management system for electric UAVs. In IEEE Aerospace Conference Proceedings, pages 1–9, 2011. ISBN 9781424473502. doi: 10.1109/AERO.2011.5747587.
- [108] Bhaskar Saha, Edwin Koshimoto, Cuong C. Quach, Sixto Vazquez, Edward Hogge, Thomas Strom, Boyd Hill, and Kai Goebel. Predicting battery life for electric UAVs. In AIAA Infotech at Aerospace Conference and Exhibit 2011, 2011. ISBN 9781600869440. doi: 10.2514/6.2011-1517.
- [109] Abhinav Saxena, José Celaya, Edward Balaban, Kai Goebel, Bhaskar Saha, Sankalita Saha, and Mark Schwabacher. Metrics for evaluating performance of prognostic techniques. In 2008 International Conference on Prognostics and Health Management, PHM 2008, 2008. ISBN 9781424419357. doi: 10.1109/PHM.2008.4711436.
- [110] Abhinav Saxena, José Celaya, Bhaskar Saha, Sankalita Saha, and Kai Goebel. Evaluating algorithm performance metrics tailored for prognostics. In IEEE Aerospace Conference, pages 1–13, 2009. ISBN 9781424426225. doi: 10.1109/AERO.2009.4839666.

- [111] Abhinav Saxena, Indranil Roychoudhury, Jose Celaya, Sankalita Saha, Bhaskar Saha, and Kai Goebel. Requirements Specification for Prognostics Performance - An Overview. In AIAA Infotech@Aerospace 2010, Reston, Virigina, apr 2010. American Institute of Aeronautics and Astronautics. ISBN 9781600869631. doi: 10.2514/6.2010-3398.
- [112] Abhinav Saxena, José Celaya, Indranil Roychoudhury, Bhaskar Saha, Sankalita Saha, and Kai Goebel. Designing Data-Driven Battery Prognostic Approaches for Variable Loading Profiles: Some Lessons Learned. First European Conference of the Prognostics and Health Management Society 2012, page 10, 2012.
- [113] Gina Sierra, Marcos Orchard, Kai Goebel, and Chetan Kulkarni. Battery health management for smallsize rotary-wing electric unmanned aerial vehicles: An efficient approach for constrained computing platforms. Reliability Engineering and System Safety, 182:166–178, feb 2019. ISSN 09518320. doi: 10.1016/j.ress.2018.04.030.
- [114] Joshuah K Stolaroff, Constantine Samaras, Emma R O'Neill, Alia Lubers, Alexandra S Mitchell, and Daniel Ceperley. Energy use and life cycle greenhouse gas emissions of drones for commercial package delivery. Nature communications, 9(1):1–13, 2018.
- [115] Valentin Sulzer, Scott G. Marquis, Robert Timms, Martin Robinson, and S. Jon Chapman. Python Battery Mathematical Modelling (PyBaMM). Journal of Open Research Software, 9(1):14, 2021. doi: 10.5334/jors.309.
- [116] Gian Antonio Susto, Alessandro Beghi, and Cristina De Luca. A predictive maintenance system for epitaxy processes based on filtering and prediction techniques. IEEE Transactions on Semiconductor Manufacturing, 25(4):638–649, 2012. doi: 10.1109/TSM.2012.2209131.
- [117] Koji Suzuki, Paulo Kemper Filho, and James Morrison. Automatic battery replacement system for UAVs: Analysis and design. Journal of Intelligent and Robotic Systems: Theory and Applications, 65(1-4):563– 586, jan 2012. ISSN 09210296. doi: 10.1007/s10846-011-9616-y.
- [118] Mohd Tariq, Ali Maswood, Chandana Jayampathi Gajanayake, and Amit Kumar. Gupta. Aircraft batteries: current trend towards more electric aircraft. IET electrical systems in transportation, 7:93–103, 2017.
- [119] Wieger Willem Tiddens, Anne Johannes Jan Braaksma, and Tiedo Tinga. Selecting suitable candidates for predictive maintenance. International Journal of Prognostics and Health Management, 9(1), 2018.
- [120] Li Yuan Ting and Dzwo Min Dai. The identification of human errors leading to accidents for improving aviation safety. In IEEE Conference on Intelligent Transportation Systems, Proceedings, ITSC, pages 38–43, 2011. ISBN 9781457721984. doi: 10.1109/ITSC.2011.6082851.
- [121] Tiedo Tinga and Richard Loendersloot. Aligning PHM, SHM and CBM by understanding the physical system failure behaviour. In A. Bregon and M.J. Daigle, editors, Proceedings of European Conference of the Prognostics and Health Management Society, pages 162–171. PHM society, 2014. ISBN 978-1-936263-16-5.
- [122] Yousra Tourki, Jeffrey Keisler, and Igor Linkov. Scenario analysis: a review of methods and applications for engineering and environmental systems. Environment Systems & Decisions, 33(1):3–20, 2013.
- [123] Rhea Valentina, Alexander Viehl, Oliver Bringmann, and Wolfgang Rosenstiel. Battery aging estimation for eco-driving strategy and electric vehicles sustainability. In 40th Annual Conference of the IEEE Industrial Electronics Society, pages 5622–5627. Institute of Electrical and Electronics Engineers Inc., feb 2014. ISBN 9781479940325. doi: 10.1109/IECON.2014.7049361.
- [124] Jorne Van den Bergh, Philippe De Bruecker, Jeroen Beliën, and Jonas Peeters. Aircraft maintenance operations: state of the art. Technical report, Katholieke Universiteit Leuven and Hogeschool-Universiteit Brussel, 2013.
- [125] Venkatasubramanian Viswanathan and Matthew Knapp. Potential for electric aircraft. Nature, 2(2): 88–89, feb 2019. ISSN 23989629. doi: 10.1038/s41893-019-0233-2.

- [126] Geert Waeyenbergh, Liliane Pintelon, and Ludo. Gelders. A stepping stone towards knowledge based maintenance. The South African Journal of Industrial Engineering, 12(1), 2001.
- [127] Pat Wheeler. Technology for the more and all electric aircraft of the future. In 2016 IEEE International Conference on Automatica (ICA-ACCA), pages 1–5, 2016. doi: 10.1109/ICA-ACCA.2016.7778519.
- [128] Kevin Williams. A summary of unmanned aircraft accident/incident data: Human factors implications. page 17, 12 2004.
- [129] Nick Williard, Wei He, Michael Osterman, and Michael Pecht. Predicting Remaining Capacity of Batteries for UAVs and Electric Vehicle Applications. In IMAPS Advanced Technology Workshop on High Reliability Microelectronics for Military Applications, 2011.
- [130] Bo Wu, Xisheng Jia, Lianghua Xia, Congying Lei, and Yabin Wang. Cost-benefit analysis of the application of prognostics and health management technology. In 2013 Proceedings of 2013 International Conference on Quality, Reliability, Risk, Maintenance, and Safety Engineering, pages 1923–1927, 2013. ISBN 9781479910144. doi: 10.1109/QR2MSE.2013.6625955.
- [131] Bin Xu, Weiping Diao, Guangrui Wen, Song-Yul Choe, Jonghoon Kim, and Michael Pecht. Decoupling the thermal and non-thermal effects of discharge c-rate on the capacity fade of lithium-ion batteries. Journal of Power Sources, 510:230390, 2021. ISSN 0378-7753. doi: 10.1016/j.jpowsour.2021.230390.
- [132] Bolun Xu, Alexandre Oudalov, Andreas Ulbig, Göran Andersson, and Daniel S. Kirschen. Modeling of lithium-ion battery degradation for cell life assessment. IEEE Transactions on Smart Grid, 9(2):1131– 1140, 2018. ISSN 19493053. doi: 10.1109/TSG.2016.2578950.
- [133] Xiao Guang Yang, Yongjun Leng, Guangsheng Zhang, Shanhai Ge, and Chao Yang Wang. Modeling of lithium plating induced aging of lithium-ion batteries: Transition from linear to nonlinear aging. Journal of Power Sources, 360:28–40, 2017. doi: 10.1016/j.jpowsour.2017.05.110.
- [134] Quan-Qing Yu, Rui Xiong, Le-Yi Wang, and Cheng Lin. A comparative study on open circuit voltage models for lithium-ion batteries. Chinese Journal of Mechanical Engineering, 31(4), 2018. doi: 10. 1186/s10033-018-0268-8.
- [135] Haitao Zhang, Ming Zhou, and Xudong Lan. State of charge estimation algorithm for unmanned aerial vehicle power-type lithium battery packs based on the extended kalman filter. Energies, 12(20), 2019. ISSN 1996-1073. doi: 10.3390/en12203960.